Western SGraduate & Postdoctoral Studies

Western University Scholarship@Western

Electronic Thesis and Dissertation Repository

5-7-2013 12:00 AM

Origins of stable isotopic variations in Late Pleistocene horse enamel and bone from Alberta

Nicolle S. Bellissimo The University of Western Ontario

Supervisor Dr. Fred J. Longstaffe *The University of Western Ontario*

Graduate Program in Geology A thesis submitted in partial fulfillment of the requirements for the degree in Master of Science © Nicolle S. Bellissimo 2013

Follow this and additional works at: https://ir.lib.uwo.ca/etd

Part of the Geochemistry Commons, and the Geology Commons

Recommended Citation

Bellissimo, Nicolle S., "Origins of stable isotopic variations in Late Pleistocene horse enamel and bone from Alberta" (2013). *Electronic Thesis and Dissertation Repository*. 1273. https://ir.lib.uwo.ca/etd/1273

This Dissertation/Thesis is brought to you for free and open access by Scholarship@Western. It has been accepted for inclusion in Electronic Thesis and Dissertation Repository by an authorized administrator of Scholarship@Western. For more information, please contact wlswadmin@uwo.ca.

ORIGINS OF STABLE ISOTOPIC VARIATIONS IN LATE PLEISTOCENE HORSE ENAMEL AND BONE FROM ALBERTA

(Thesis format: Monograph)

by

Nicolle Sara Bellissimo

Graduate Program in Geology

A thesis submitted in partial fulfillment of the requirements for the degree of Master of Science

The School of Graduate and Postdoctoral Studies The University of Western Ontario London, Ontario, Canada

© Nicolle Sara Bellissimo 2013

Abstract

Oxygen and carbon isotopic measurements of coevally formed bone and tooth enamel bioapatite from a modern equid show that these tissues record drinking water and diet isotopic signals in an identical fashion. Hence, data for both tissues can be combined to track movement, dietary changes, and seasonal variability over the animal's lifetime, and climatic variability over longer time periods. This tool was tested for horses using ten paired tooth and bone samples to reconstruct conditions in Alberta during the Late Pleistocene. While *post-mortem* isotopic alteration confounded interpretation of the results, two key findings emerged: (i) pre- and post-Last Glacial Maximum (LGM) time periods based on radiocarbon dates correspond with high and low δ^{15} N collagen values, respectively; and (ii) pre- and post-LGM horses have similar drinking water δ^{18} O and diet δ^{13} C values, suggesting that environmental conditions, including seasonality, were similar across the time periods represented by these samples.

Keywords: stable isotopes, phosphate, carbonate, Pleistocene, Alberta, palaeoenvironment, horses, enamel, bone, tissue spacing, incremental enamel sampling

Acknowledgments

I would like to thank my supervisor, Fred Longstaffe, for allowing me to choose a research project that I was interested in and for providing financial, professional, and personal support to help me complete this project. Thank you for dedicating your time to guide, help, and advise me during my Masters. Thank you Emily Webb for being my peer mentor and taking time out of your schedule to teach me the foundations needed to build my masters. Thank you for your help in Edmonton, and for going with me to get my modern horse sample (which I wouldn't have gotten if you weren't there, you know why).

This research was supported through funding provided by: the University of Western Ontario, Ontario Graduate Scholarship in Science and Technology (OGSST), Queen Elizabeth II Scholarship, Graduate Thesis Research Fund (GTRF), and by a Natural Sciences and Engineering Research Council of Canada (NSERC) Discovery grant and Canada Foundation for Innovation (CFI) and Ontario Research Fund (ORF) infrastructure grants to F.J. Longstaffe, and in part, through release time provided to F.J. Longstaffe by the Canada Research Chairs Program.

To Kim Law, Li Huang, Miss Grace Yau, Lisa Munro, Zoe Morris, Rachel Schwartz-Narbonne, Alicia Donis, and Andrea Prentice for helping me complete my lab work in the Laboratory for Stable Isotope Science. I would not have been able to finish without your knowledge and help. You, as well as the other LSIS members (i.e. the other graduate students) made my time in the lab fun and very enjoyable and I will cherish those memories forever.

To Chris Jass, Peter Milot, and the Royal Alberta Museum for providing access to Pleistocene horse samples. To Ed Eastaugh for providing access to and help in the Zooarchaeology lab to deflesh the modern horse.

I would like to thank Keith, my family, and Toby for their love, support, and encouragement throughout my Masters. Thank you to my friends in the department, for being supportive and providing me with sanity via the many social, sport, and grad club activities (i.e. beer drinking), you made the department feel like an extended family.

| Acknowledgments | iii |
|--|-------------|
| Table of Contents | iv |
| List of Tables | vi |
| List of Figures | vii |
| Chanter 1. Introduction | 1 |
| | ····· · |
| 1.1 General background and objectives | |
| 1.2 Background - Stable isotopes | |
| 1.2.1 Carbon and nitrogen | 4 |
| 1.2.2 Oxygen | 6 |
| | |
| 1.3 Background - Skeletal tissues | 9 |
| 1.3.1 Bones and teeth | 9 |
| 1.3.2 Horse dentition | 11 |
| 1.3.3 Sequence of tooth eruption and mineralization in horses | 11 |
| | |
| 1.4 Background - Horses | |
| 1.4.1 Pleistocene horses | 14 |
| 15 Declaround Disistence Alberta | 16 |
| 1.5 Background - Pleistocene Alberta | 10 |
| | |
| Chapter 2: Material and Methods | |
| | |
| 2.1 Samples | 19 |
| 2.2 Sampling techniques | 20 |
| 2.3 Diagenesis | |
| 2.4 Stable isotopes | |
| 2.4.1 Phosphate preparation | |
| 2.4.1.1 Phosphate analysis: BrF_5 method | |
| 2.4.1.2 Phosphate analysis: TC-EA method | |
| 2.4.2 Structural carbonate analysis | |
| 2.4.3 Collagen analysis | |
| 25 Deinling meter succession composition | 21 |
| 2.5 Drinking water oxygen isotope composition | |
| 2.0 Radiocarbon analysis | |
| 2.7 Deneshing considerations | |
| | |
| Chapter 3: Results | |
| - | |
| 3.1 Assessment of diagenesis of bioapatite and associated collagen | 35 |
| 3.2 Oxygen isotopic analysis of bioapatite phosphate | |
| 3.2.1 BrF ₅ method versus TC-EA | |
| | |
| 3.3 Radiocarbon dating | |
| 3.4 Oxygen and carbon isotopic analysis of bioapatite structural carbonate | |
| 3.4.1 Bulk structural carbonate | |
| 3.4.2 Incremental structural carbonate | 41 |
| 2.5. Collecon | A 77 |
| 3.6 Enamel hone tissue specing | / 4 / ۸۰ |
| 5.6 Enamer bone ussue spacing | |

Table of Contents

| 3.6.1 | Phosphate spacing | 48 |
|----------|---|----|
| 3.6.2 | Structural carbonate spacing | 48 |
| | | |
| Chapte | er 4: Discussion | 50 |
| 4.1 Pos | sible limitations on the life history record for the modern horse | 50 |
| 4.2 Oxy | vgen isotone tissue-snacing | 52 |
| 4.2.1 | Modern horse | |
| 4.2.2 | Phosphate-structural carbonate | 53 |
| 13 Bul | k and serially sampled structural carbonate | 60 |
| 4.5 Dui | Modern horse structural carbonate drinking water and diet isotonic signatures | |
| 4.3.2 | Differences between bulk and serially sampled structural carbonate | 61 |
| 4.3.3 | Pleistocene bulk structural carbonate drinking water and diet isotopic signatures | 61 |
| 4.3.4 | Growth rate estimates for serially sampled Pleistocene horse teeth | 62 |
| 4.4 Pala | aeoecological implications | 63 |
| 4.4.1 | Late Pleistocene climate | 63 |
| 4.4.2 | Bulk patterns in $\delta^{18}O_{sc}$ and $\delta^{13}C_{sc}$ values | 67 |
| 4.4.3 | Serial sampling patterns in $\delta^{18}O_{sc}$ and $\delta^{13}C_{sc}$ | 69 |
| 4.4.4 | Structural carbonate-collagen carbon isotope relationship | 72 |
| 4.4.5 | Collagen δ^{15} N values | 73 |
| | | |
| Chapte | er 5: Conclusions and Future Work | 76 |
| 5.1 Ma | jor Conclusions | 76 |
| 5.2 Fut | ure Work | 77 |
| Defe | | 70 |
| Keiere | nces | |
| Curric | ulum Vitae | |

List of Tables

| Table 2.1 | Sample information | 19 |
|-------------------------------|---|----|
| Table 2.2 | Phosphate and structural carbonate (pretreated and untreated) bone isotopic ositions of the defleshing tests. | 33 |
| Table 3.1 | Horse bone and tooth diagenetic tests | 36 |
| Table 3.2 compo | Horse bone and tooth $\delta^{18}O_p$ and associated drinking water oxygen isotope ositions ($\delta^{18}O_{dw}$). | 37 |
| Table 3.3 and co | Stable isotopic compositions for structural carbonate (untreated and pretreated) ollagen, and corresponding diet and drinking water δ -values | 40 |
| Table 3.4 diet ar | Stable isotopic compositions of serially sampled enamel and corresponding drinking water δ -values | 42 |
| Table 3.5 treated | Structural carbonate and phosphate bone and tooth isotopic compositions (pre- d, bulk samples) and corresponding bone-tooth isotopic tissue spacing | 49 |
| Table 4.1 phospl to you | The height of Pleistocene horse teeth (distal end, mm) and corresponding hate and structural carbonate $\Delta^{18}O_{bone-tooth}$ spacing, listed in order from oldest ngest horse | 59 |
| Table 4.2 | Growth time estimates (in months) for serially sampled Pleistocene horse teeth | 63 |

List of Figures

| Figure 1.1 An illustration of Retzius stria (identified by white arrows) in horse enamel. Note: this is a magnified view of a thin section under crossed polarized light | 10 |
|--|----|
| Figure 1.2 The Triadan system of dental nomenclature, showing adult dentition. The quadrants are numbered from 1 to 4 in a clockwise direction | 12 |
| Figure 1.3 Location of Alberta, Canada under glacial ice during the LGM | 17 |
| Figure 2.1 Modern horse tooth 308 (LP4). Dashed line marks the outline of the enamel layer, illustrating that it does not extend all the way to the root | 20 |
| Figure 2.2 Radiograph of left horse mandible. There are no deciduous teeth in the jaw below the permanent teeth. Scale bar=8.5 cm. | 21 |
| Figure 2.3 A) Location of 06 and 08 mandibular cheek teeth. Teeth were cut along buccal-lingual axis (dental orientations are shown) before isotopic analysis. B) Position of cut tooth on drill stage. | 22 |
| Figure 2.4 A) Example of a sample line within the inner enamel (IE) layer; RAM 3T, line 4 (from occlusal surface). Black lines mark the division between outer cementum, enamel, and dentin. B) An example of sampling line spacing within the IE layer. | 24 |
| Figure 2.5 Diagram of a horse mandible showing modern bone and teeth samples. CW, WW, and E were defleshing tests and represent cold water, warm water, and enzyme experiments | 34 |
| Figure 3.1 Bone and tooth $\delta^{18}O_p$ values for the Pleistocene horses. Results are averaged from both methods of phosphate analysis. | 38 |
| Figure 3.2 Pretreated bone and tooth $\delta^{18}O_{sc}$ values of the Pleistocene horses | 39 |
| Figure 3.3 $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values for serially sampled Pleistocene horse teeth. Bulk $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values are also plotted for each tooth. | 44 |

| Figure 3.4 Collagen stable carbon and nitrogen isotopic compositions for Pleistocene horses. Radiocarbon ages are listed adjacent to the dated samples | 47 |
|---|----|
| Figure 4.1 A) Map of amount-weighted monthly precipitation δ^{18} O values for North America. The black box identifies the general area in which the modern horse lived. B) Monthly precipitation δ^{18} O range for southwestern Ontario, where the modern horse lived. | 51 |
| Figure 4.2 $\delta^{18}O_p$ versus $\delta^{18}O_{sc}$ for individual samples of Pleistocene teeth A) and bone B). The line of best fit for modern equids is shown by the solid black line (Bryant et al., 1996a). | 56 |
| Figure 4.3 Calculated $\delta^{18}O_{dw}$ for Pleistocene and modern bone and teeth based on $\delta^{18}O_p$ values (Equation 2.2) | 58 |
| Figure 4.4 Bone collagen carbon and nitrogen isotopic compositions for Pleistocene horses. Radiocarbon ages are listed adjacent to the dated samples. LGM = Last Glacial Maximum. | 65 |
| Figure 4.5 Map and timing of the ice-sheet retreat near Alberta after the LGM, following Dyke et al. (2003) and as modified from Munyikwa et al. (2011) | 67 |

Chapter 1

Introduction

1.1 General background and objectives

The oxygen isotopic composition of vertebrate bone or enamel bioapatite can been used to reconstruct climatic conditions recorded during bioapatite formation (e.g. Longinelli, 1984; Sharp & Cerling, 1998; Fricke et al., 1998, etc.). Similarly, the carbon isotopic composition of bioapatite has been used to reconstruct dietary components consumed during an animal's life (e.g. Sullivan & Krueger, 1981; Lee-Thorp & van der Merwe, 1987; Wang et al., 1994, etc.). The bioapatite phosphate and structural carbonate oxygen and carbon isotope fractionation between enamel and bone in modern and Pleistocene horses is investigated in this thesis. An understanding of these fractionations is important for accurate reconstruction of climate and migratory histories, and dietary preferences. This is especially true in light of the proposition of Warinner and Tuross (2009), that bone and enamel in pigs may not be isotopically equivalent tissues, possibly the result of different biological factors occurring during bioapatite formation (see section 1.2.2). If this applies to different taxa, it will have many influences on palaeoecological and palaeoclimatological reconstructions, such as: what tissues (enamel or bone) are sampled for analysis, the relevance of tissue comparisons, and whether or not a fractionation factor has to be applied to bone or enamel, etc. Therefore, it is necessary to determine what isotopic signals are incorporated during enamel and bone formation for reliable environmental reconstructions. To investigate this question, the oxygen isotopic composition of phosphate and the oxygen and carbon isotopic compositions of structural carbonate have been analyzed from modern (Equus caballus) and Pleistocene horse bone and enamel samples.

In addition, the organic component of bone, mainly collagen, from modern and Pleistocene horses has been analyzed for its carbon and nitrogen isotopic compositions. These data have been utilized to examine environmental conditions during the Late Pleistocene (126–12 ka BP) in central Alberta (Head et al., 2008). This is an area for which current understanding of Pleistocene ecosystems is quite limited because there is minimal isotopic research, particularly on large herbivores, for that region. It is important to fill this gap of knowledge for a key area affected by continental ecosystem and climatic changes at the end of the Pleistocene. Choosing a large herbivore for this study also simplifies interpretation of the carbon and nitrogen isotopic results for bone collagen, as these compositions are a direct measure (± metabolic effects) of the flora consumed by these animals.

The Pleistocene Epoch (2.588 Ma-12 ka BP) was characterized by extensive ice sheet cover over most of the Northern Hemisphere (Head et al., 2008; Miller et al., 2010). For the past 700 ka BP, glacial cycles in the Northern Hemisphere have lasted ~100 ka and were separated by short warm interglacial periods (Miller et al., 2010). During glacial periods, ocean levels decreased enough to periodically expose the land between the Bering Strait, creating the landmass known as Beringia. Beringia had an area of ~34 million km², which included eastern Russia, northwestern North America and the 'Bering land bridge' (Harrington, 2005). Beringia was a significant part of the mammoth steppe, a very large continental biome that extended from western Europe to North America (Guthrie, 1968; Szpak et al., 2010). Beringia was ice-free during glacial intervals, thus permitting the exchange of flora and fauna among various areas of Beringia (i.e. between Alaska and Siberia) via the Bering Strait (Miller et al., 2010; Shapiro et al., 2004 and references therein). The array of flora (e.g. sage and grass-like plants, etc.) and megafauna (animals >44kg, e.g. woolly mammoth, bison, horse, etc.) that lived in Beringia and the mammoth steppe during the Late Pleistocene created an ecosystem for which there is no modern analogue (Zazula et al., 2003; Harrington, 2005; Koch & Barnosky, 2006).

Throughout the Late Pleistocene, especially during periods of reduced ice cover, the plants and megafauna that occupied Beringia were able to extend their ranges into much of North America, including Alberta and the continental United States (Shapiro et al., 2004 and references therein; Fox-Dobbs et al., 2008; Burns, 2010 and references therein). After the Last Glacial Maximum (LGM) 22–18 ¹⁴C ka BP (26.2–21.4 ka BP) the animals returned to these areas, including Alberta (e.g. Burns, 1996). There is some debate as to when the ice sheets retreated in Alberta after the LGM and whether or not an 'ice-free' passage existed (e.g. Burns, 1996; Munyikwa et al., 2011, etc.). For example,

Dyke and Prest (1987) and Dyke et al. (2003) (based on radiocarbon chronologies) portray Alberta as ice-free between 12–11 ¹⁴C ka BP (13.8–12.8 ka BP, converted from INTCAL09), which was thought to be too late to account for humans already being south of the ice sheets in North America (Munyikwa et al., 2011 and references therein). But recent estimates using optically stimulated luminescence dating suggest it was ice-free a thousand years earlier, which would be early enough for humans to have used Alberta as an 'ice-free' passage from Beringia to south of the ice sheets (Munyikwa et al., 2011 and references therein). Regardless, Alberta played an important role in megafauna migrations and was central to large-scale Late Pleistocene glacial events.[Dates reported in this thesis are given as uncalibrated radiocarbon years before AD 1950, for example 10 ¹⁴C ka BP, where ka is thousands of years. Conversion to calibrated radiocarbon years (ka BP) was performed using INTCAL09 (Reimer et al., 2009), and is shown in brackets beside the radiocarbon date. The only exceptions made are for research papers that report only calibrated radiocarbon years.]

Horses were chosen for this study because they were a common herbivore throughout Beringia, the mammoth steppe, and Alberta and are abundantly represented in the fossil record (Burns ,1996; Guthrie, 2001; Hoppe et al., 2004a). The isotopic analysis undertaken on these samples is important for several reasons. First, understanding more about the phosphate oxygen and carbonate oxygen isotope fractionation between horse bone and teeth is necessary because these patterns have not been clearly defined for horses and large-bodied herbivores. This information will add to the overall physiological knowledge of large-bodied mammals. Second, this research will improve our understanding of the terrestrial record of environmental change in central Alberta during the Pleistocene for a significant group of animals. This is especially important considering Alberta's possible role as an 'ice-free' passage, facilitating human migration from Beringia to south of the ice sheets. In addition, a better understanding of the environment in which extinct wild horses lived will help characterize the changing nature of their ecological niche within North America near the Pleistocene-Holocene transition when 70 % of megafaunal genera became extinct, including Pleistocene horses (Guthrie, 2003; Koch & Barnosky, 2006; Gill et al., 2009). Radiocarbon dates for 220 Alaska and Yukon Territory horses suggest they went extinct $\sim 12.5^{14}$ C ka BP ($\sim 14.9-14.8$ ka BP),

with a rapid decline in body size occurring beforehand (Guthrie, 2003 and references therein). The causes of the extinction are still debated but commonly attributed to either human impact or climate change, or both (Guthrie, 2003; Shapiro et al., 2004; Koch & Barnosky, 2006).

1.2 Background - Stable isotopes

Stable isotopes are atoms that contain the same number of protons and electrons, but a different number of neutrons, resulting in different atomic masses. This difference in mass (e.g. 12 C vs. 13 C) causes isotopes of the same element to have slightly different thermodynamic and kinetic properties (Urey, 1947; Koch, 1998; Hoefs, 2009). These differences can cause separation of isotopes during physical and chemical processes, making stable isotopes useful tracers. For example, light isotopes (those with fewer neutrons) of a given element take part in chemical reactions at a faster rate than heavier isotopes (Ambrose, 1993; Koch, 1998; Hoefs, 2009). Fractionations of stable isotopes are relatively small and are therefore reported using delta (δ) notation expressed in per mil (‰):

$$\delta(\%) = [(R_{sample}/R_{standard})-1], \qquad [Equation 1]$$

where R_{sample} and $R_{standard}$ are the high-mass to low-mass isotope ratios of the same element (e.g. ${}^{18}O/{}^{16}O$, ${}^{13}C/{}^{12}C$, ${}^{15}N/{}^{14}N$) in the sample and standard, respectively (Hoefs, 2009).

1.2.1 Carbon and nitrogen

The stable carbon and nitrogen isotopic compositions of food consumed by an animal are transferred to the animal's tissues during the incorporation and digestion of food (Koch, 1994; Stevens, 2009). As such, the isotopic composition of bones and teeth can act as a tracer of different dietary inputs (i.e. different plants in the case of herbivores) with specific isotopic compositions (Koch et al., 1994). The carbon isotope composition of animal tissues (e.g. bone and teeth) is mostly related to C_3 versus C_4 plant

intake (Kohn & Cerling, 2002). Ingestion of C₃ (trees, most shrubs and herbs, and grasses adapted to cool or wet climates) or C₄ (grasses adapted to warm or arid habitats) plants can be identified because their different photosynthetic pathways produce distinct isotopic compositions (average δ^{13} C: C₃ –26.5 ‰; C₄ –12.5 ‰, respectively) (Smith, 1972; O'Leary, 1988; Ambrose, 1993; Koch et al., 1994). The δ^{13} C values of plants can also be influenced by canopy cover, light intensity, moisture level, atmospheric CO₂, temperature changes, and salinity (Kohn & Cerling, 2002; Clementz et al., 2009; Heaton, 1999; Stevens et al., 2009).

The carbon isotopic composition of the diet is transferred to animal tissues with some associated fractionations. The carbon isotope composition of the structural carbonate that substitutes for PO_4^{3-} and OH^- sites in bioapatite (part of bone and teeth; see below) reflects bulk carbon from the whole diet (i.e. carbohydrates, lipids, and proteins) in most situations (Ambrose & Norr, 1993; Tieszen & Fagre, 1993; Kohn & Cerling, 2002; Clementz et al., 2009). Cerling and Harris (1999) determined that structural carbonate is enriched in ¹³C by +14.1 ± 0.5 ‰ compared to the diet of large ungulate mammals.

The carbon and nitrogen isotopic compositions of bone collagen mainly reflect dietary protein (Ambrose & Norr, 1993; Stevens & Hedges, 2004; Jim et al., 2004). Because collagen takes a long time to turnover (>10 years), the carbon and nitrogen isotope compositions of bone collagen will represent the average composition of dietary protein consumed over a significant period of time (Ambrose, 1993; Stevens & Hedges, 2004; Ayliffe et al., 2004; Hedges et al., 2007). For large herbivores, there is a carbon isotope fractionation between collagen and diet of about +5.5 ‰ (Vogel et al., 1978; Sullivan & Krueger, 1981; van der Merwe, 1989; Ambrose & Norr, 1993). As to whether there are systematic differences in collagen δ^{13} C values among different bones in an animal, Warinner and Tuross (2009) reported no difference for a humerus vs. a mandible bone (both formed by different ossification processes) from a pig. This suggests that collagen from different bones within such larger animals record very similar δ^{13} C values.

The δ^{15} N values of collagen are mainly determined by the nitrogen isotopic composition of the diet and there is little difference (typically $\leq 1\%$) between the δ^{15} N values of various tissues within a body (e.g. nail vs. hair vs. liver vs. muscle; DeNiro &

Epstein, 1981; Ambrose, 1993; Nardoto et al., 2006; Crowley et al., 2010). With an increase in δ^{15} N values of ~3 ‰ each step up the food chain under most circumstances, nitrogen isotopic compositions can be used to determine the trophic level status of an animal (DeNiro & Epstein, 1981; Schoeninger & DeNiro, 1984; Koch et al., 1994; Bocherens & Drucker, 2003; Sponheimer et al., 2003a). However, horses are herbivores and therefore any variation in the nitrogen isotopic compositions of their tissues would not normally reflect a change in trophic level (Stevens & Hedges, 2004).

The δ^{15} N and δ^{13} C values of soils, plants, and herbivores can be influenced by stress, climate, and physiological factors (Ambrose, 1993; Sponheimer et al., 2003a; Fuller et al., 2005). For example, canopy cover, light availability, moisture level, atmospheric CO₂, temperature, and salinity changes can affect the δ^{13} C values of plants because these factors influence the amount of isotopic fractionation that occurs during photosynthesis (see Heaton, 1999; Stevens et al., 2009; Kohn & Cerling, 2002; Clementz et al., 2009). Environmental factors also play a role. The canopy effect is one such factor; it causes ¹³C depletion in plants growing under a closed forest canopy compared to plants growing in more open environments. This effect is influenced by the incorporation of ¹³C-depleted CO₂ from respiration and/or soil decomposition (e.g. van Der Merwe & Medina, 1989; see review in Heaton, 1999). Likewise, plant δ^{15} N values tend to decrease with decreasing average annual temperature and with increasing average annual precipitation (Amundson et al., 2003).

As mentioned earlier, different tissues (e.g. bone collagen vs. tooth dentin) have different turnover rates for both carbon and nitrogen. Thus their isotopic compositions can be used to identify certain 'trophic level effects', for example, those associated with weaning. Tooth dentin laid down during nursing would have an offset (up to about +3 ‰ relative to the mother's nitrogen isotope composition) compared to bone/tooth dentin remodelled/deposited after weaning (which has nitrogen isotopic compositions closer to adult values) (Fogel et al., 1989; Richards et al., 2002; Metcalfe et al., 2011).

1.2.2 Oxygen

The oxygen isotope compositions of biogenic materials, such as bones and teeth and the bioapatite that they contain, record a range of environmental parameters and biological processes, and hence can be used to reveal palaeoclimate information (Kohn & Cerling, 2002). The oxygen isotope composition of bioapatite is determined by the temperature and composition of body water when it formed (Koch et al., 1994). Bioapatite is thought to precipitate in oxygen isotopic equilibrium with body water. Body water composition is determined by drinking water and water in the diet. The oxygen isotope composition of large herbivores, in particular, may be affected by the water in their food, especially if it has a high water content (Koch et al., 1994; Daux et al., 2008). The oxygen isotope composition of ingested water is affected by several interrelated factors including the δ^{18} O value of: (i) precipitation, which is related to average annual temperature, humidity, and air mass source, and (ii) plant water, which varies with environmental stressors such as temperature, and humidity, as well as soil-water composition, rooting depth, photosynthetic pathway, and plant morphology (Dansgaard, 1964; Koch et al., 1994).

Palaeoclimate information (e.g. during the Pleistocene) can be obtained from vertebrate fossils by analyzing the oxygen isotope composition of bioapatite phosphate and/or structural carbonate. These isotopic signatures are related to the isotopic composition of ingested water, which is related to local rainfall δ^{18} O values, which in turn are related to climate (Dansgaard, 1964; Longinelli, 1984; Luz et al., 1984; Bryant & Froelich, 1995; Zazzo et al., 2004a). Since mammalian body temperature is constant at \sim 37 °C, the oxygen isotope composition of bioapatite is not directly influenced by external temperature. Bioapatite mineralizes in equilibrium with body water and reflects only the composition of oxygen inputs. Because horses are large, water-dependent herbivores, their body-water δ^{18} O values are mainly controlled by drinking water (Longinelli, 1984; Luz et al., 1984; Bryant & Froelich, 1995); there is usually an offset of about +18 ‰ between the δ^{18} O of bioapatite PO₄ and body water in mammals (Kohn & Cerling, 2002). δ^{18} O values of precipitation in mid- to high-latitudes usually follow a sinusoidal pattern on a seasonal basis, with one cycle representing one year (Rozanski et al., 1993). Therefore, an animal such as a horse whose teeth grow for a period of many months to a few years, will display a sinusoidal curve in δ^{18} O of enamel bioapatite if it is drinking water from a somewhat constant source with an input dominated by

precipitation (e.g. Sharp & Cerling, 1998; Balasse et al., 2003a; Zazzo et al., 2005; Feranec et al., 2009; Metcalfe et al., 2011).

In bioapatite, oxygen contained in both its structural carbonate (CO₃) and phosphate (PO₄) components has exchanged isotopes with body water, resulting in linearly correlated oxygen isotope compositions (Bryant et al., 1996a; Iacumin et al., 1996). The offset between CO₃ and PO₄ is slightly variable: Lecuyer et al. (2010) estimated 1000 ln α (CO₃-PO₄) values, based on O'Neil et al. (1969) and Kim and O'Neil (1997)'s equations, of +7 to +8 ‰ for temperatures ranging from 10–37 °C. This matches the calculations of Bryant et al. (1996a), who combined the fractionation equations of Shemesh et al. (1988) and Zheng (1996) to achieve the same result. The measured oxygen isotope fractionation between hydroxyapatite-bound carbonate and phosphate has a value of +7.5 ‰ at 37 °C (Lecuyer et al., 2010; Kolodny et al., 1983). Lecuyer et al. (2010) compiled oxygen isotope data for structural carbonate and phosphate from modern mammals to obtain a value of +8.1 ‰ for the bioapatite oxygen isotope CO₃-PO₄ fractionation. They suggest that this slightly higher value might result from nonequilibrium oxygen isotope fractionation or changes in pH of the extracellular fluid.

It has been proposed that differences can exist in the oxygen (and carbon) isotope compositions of structural carbonate between different tissues of the same animal, e.g. enamel bioapatite vs. bone bioapatite. Warinner and Tuross (2009) examined pig bone and enamel bioapatite formed at the same time and found them to have different isotopic compositions. They reported oxygen and carbon isotope fractionations between enamel bioapatite and bone bioapatite of +1.7 ‰ and 2.3 ‰, respectively. They suggest the offset could be because of "the differential crystallinity indices of the two mineral types or poorly understood local fractionation factors in different body water and micronutrient pools" but that more work is needed to identify the cause. It is presently unknown whether or not a similar offset exists for the oxygen isotope composition of enamel vs. bone bioapatite PO₄, particularly for larger mammals.

1.3 Background - Skeletal tissues

1.3.1 Bones and teeth

Bones and teeth in animals are composed of an inorganic component (mineral: bioapatite) and an organic component (protein: collagen). Bones are made of ~70 wt% mineral and ~20–30 wt% collagen and a small portion of other proteins and water. This mineral/collagen ratio varies among animals, among bones in an animal, and with age (Ambrose, 1993; Lee-Thorpe, 2008; Pasteris et al., 2008). Unlike bones, mammalian teeth have three types of mineralized tissue: enamel, dentin, and cementum. Dentin and cementum have a similar amount of mineral and collagen as bone, while enamel has a much higher mineral component of 96 wt% (Driessens & Verbeeck, 1990; Skinner, 2005; Hillson, 2005). The mineral component in both bones and teeth consists of carbonated hydroxylapatite (\sim Ca₉[(PO₄)_{4.5}(CO₃)_{1.5}](OH)_{1.5}), which has a range of compositional variability arising from the ease of elemental substitution (Driessens & Verbeeck, 1990; Hoppe et al., 2004a). An example of this range is provided by Skinner (2005): (Ca, Na, Mg, K, Sr, Pb...)₁₀(PO₄, CO₃, SO₄...)₆(OH, F, Cl, CO₃)₂.

During bone formation, collagen is first laid down and plate-like bioapatite crystallites then nucleate on and grow within the collagen framework. Bone is remodelled throughout an animal's life, and depending on the size and health of the animal, it can take many years; the original collagen-mineral network laid down dissolves allowing osteoblasts to deposit new bone material (Pasteris et al., 2008; Clementz et al., 2009).

The cells that form the components of teeth are very different from those that form bones. The most common tooth-forming cells are odontoblasts, which produce dentin, and ameloblasts, which produce enamel. Dentin is formed in a similar way to bone, but it does not undergo remodelling (exceptions apply to secondary and tertiary dentin, see section 1.3.2) (Pasteris et al., 2008). Mammalian enamel grows from the occlusal surface (at the enamel-dentin junction, EDJ) to the root (moving outwards towards the outer enamel surface as it progresses downwards) (Hillson, 2005). Enamel formation takes place in a two-stage process. The matrix production stage is first: ameloblast cells secrete a lightly mineralized organic matrix; at this point it is only one third mineral. A successive series of incremental lines form during matrix production, providing a record of the direction and rate of tooth crown formation. These lines, which under an optical microscope appear as regularly spaced light and dark lines, referred to as "Retzius' striae", represent material that started mineralizing at the same time; however, they may have finished mineralizing at different times during enamel maturation (Fig. 1.1) (Hillson, 2005; Hoppe et al., 2004a). The second stage is maturation: the ameloblasts then remove the proteins and water as crystallites increase in size, producing heavily mineralized enamel (Hillson, 2005; Pasteris et al., 2008). Zazzo et al. (2005) showed two aspects of the mineralization front in bovid teeth: "a vertical component from the tip of the tooth crown to the neck, and a horizontal component from the enamel-dentine junction to the outer enamel". Enamel bioapatite crystallites are larger and have higher crystallinity than bone bioapatite crystals (LeGeros, 1991; Lee-Thorp, 2008).



Figure 1.1 An illustration of Retzius stria (identified by white arrows) in horse enamel. Note: this is a magnified view of a thin section under crossed polarized light. C=cementum; d=dentin (from Hoppe et al., 2004a).

1.3.2 Horse dentition

As is the case for all mammalian teeth, horse teeth contain three calcified dental tissues: enamel, cementum, and dentin. Horse teeth contain two types of enamel: equine type-1 and type-2 (Dixon, 2002 and references therein). Equine type-1 enamel consists of parallel layers of bioapatite prisms and flat, interprismatic, bioapatite plates, which makes this enamel hard and brittle. Cheek teeth, particularly the upper cheek teeth are composed of this type of enamel. Equine type-2 enamel is composed of groups of enamel prisms that are intricately interwoven in three dimensions, and is the main component of incisor teeth. The whole tooth is covered by cementum before tooth wear, which involves surrounding the deep infoldings in the lower cheek teeth (Dixon, 2002; Hillson, 2005). There are three types of horse dentin tissues: primary, secondary, and tertiary. Primary dentin is laid down during tooth eruption. Secondary dentin is laid down over the life of the tooth by the odontoblasts that line the outer pulp (the soft tissue in dental pulp cavities containing connective tissues, blood, lymphatic vessels, and nerves). Secondary dentin slowly replaces the pulp over the life of the tooth. Tertiary dentin is laid down where insults (i.e. infection or trauma) have occurred (Dixon, 2002 and references therein).

1.3.3 Sequence of tooth eruption and mineralization in horses

The Triadan system of dental nomenclature, used in veterinary medicine, is used here for describing equine teeth (Fig. 1.2). A horse's dentition is classified as diphyodont because they have two sets of teeth: temporary/deciduous teeth and permanent (Dixon, 2002). The deciduous cheek teeth (premolars 06s–08s) are replaced by three permanent teeth in the adult. The first premolar (05s) and three molars (09s–11s) have no deciduous teeth. In modern equids, there are 24 cheek teeth in the maxillary (upper jaw is composed of the maxillae and premaxillae) and mandibular (lower jaw) bones (Dixon, 2002; Hillson, 2005). They are rectangular in shape (except the 06s and 07s, which are triangular). The upper cheek teeth are wider than the lower cheek teeth, which have a taller erupted crown. The first cheek teeth (06s) are the shortest, with the remaining cheek teeth reaching 7–8 cm in height before wear. All permanent cheek teeth are in wear by the time the horse reaches 5 years of age. These teeth wear at a rate of ~2–3 mm/yr; therefore a 7.5 cm tall tooth will be fully worn at about 30 years of age (Dixon, 2002). In domestic horses, the 09s are the first permanent teeth to erupt at \sim 8–12 months of age. The eruption of the 10s follows at \sim 20–26 months. Permanent premolars, 06s and 07s, replace their deciduous precursors at 2.5 and 3 years of age, respectively. The last teeth to erupt are the 11s at 3.5 years and the 08s at 4 years (Hoppe et al., 2004a and references therein).



Figure 1.2 The Triadan system of dental nomenclature, showing adult dentition. The quadrants are numbered from 1 to 4 in a clockwise direction, beginning at the 'right' (as labelled in figure) maxillary quadrant (from Dixon, 2002).

Teeth grow while being completely encased in the jaw. Understanding the sequence and timing of tooth mineralization in horses is very important when considering which tooth and what part of that tooth to sample for isotopic investigation. Hoppe et al. (2004a) showed that the relative order of mandibular tooth mineralization in domestic horses is in close agreement with the sequence of tooth eruption; mandibular teeth can either form at the same time as maxillary teeth or several months before. The first adult teeth to mineralize are the 09s, which start at 2 weeks of age. The 10s are the second teeth to mineralize, followed by the 06s, 07s, and 08s. The last to mineralize are the 11s. It is important to consider that horses typically nurse for 9 months; therefore the isotopic compositions of tooth enamel that mineralized during this time (i.e. parts of 09s and 10s) could be offset relative to adult compositions (Hoppe et al., 2004a and references therein).

All teeth continue to mineralize after eruption. The duration of mineralization varies among the different teeth and is independent of the order in which they erupt and begin mineralization. The 06s have the shortest mineralization time of 18 months. The 09s and 07s finish mineralizing at 2 years of age, the 10s finish by 2.5 years of age. The 08s and 11s are the last to appear and finish mineralizing at 2.6 and 2.8 years, respectively (Hoppe et al., 2004a).

The following is a list of important considerations for interpreting isotopic compositions of horse enamel: (1) Enamel is lost as teeth wear; therefore the worn teeth of older individuals preserve isotopic signatures only from the last stages of tooth mineralization (Hoppe et al., 2004a). (2) Our understanding of tooth eruption and mineralization sequences are based on the modern horse; therefore only inferences can be made for extinct species. The similarities in tooth pattern and eruption observed amongst different species of extant equids implies they have similar patterns of mineralization and that this could also be true for extinct species (Bryant et al., 1996c; Hoppe et al., 2004a). The rate of tooth growth varies amongst modern equid species, again suggesting that such variation could have existed for extinct species (Hoppe et al., 2004a). (3) Mandibular tooth eruption/mineralization patterns are assumed to be similar to maxillary patterns.

1.4 Background - Horses

The modern domesticated horse (*Equus caballus*) is quite different from the first members of its family (Equidae), who lived 58 million years ago. In terms of diet, the evolution of horses involved a change from a predominantly C₃-based diet (e.g. trees and shrubs) to a diet consisting mainly of C₄ and C₃ grasses for several taxa during the Late Miocene to Early Pliocene (Simpson, 1951; MacFadden, 1988; Wang et al., 1994; MacFadden et al., 1999 and references therein). That said – Pleistocene and modern C₃ grasses in North America dominated/dominate cooler environments at higher elevations or above 45°N latitude. Ancestor horses were much smaller (25–50 kg) and had brachyodont (low-crowned) teeth (Bryant et al., 1996b). They lived in wooded biomes and fed on woody plants (browsers). With evolution, modern horses became much larger (75–400 kg), developed hypsodont (high-crowned) teeth and fed on grasses (grazers) (MacFadden, 1986; Bryant et al., 1996b).

1.4.1 Pleistocene horses

Horses are abundantly represented in the fossil record over the course of their 58 Ma history. Fossil horses, as well as modern horses, are widespread geographically (Bryant et al., 1994). As such, they serve as ideal proxies for palaeodiet and palaeoclimate isotopic reconstructions, because the different physiological fractionation factors associated with interspecies comparisons can be eliminated. Their abundance also allows for multi-continental examination of a climatic event (e.g. glacial-interglacial cycles). In addition, access to modern equids provides the opportunity to perform controlled studies on close analogues.

MacFadden et al. (1999) for example, used the carbon isotope compositions of horse tooth enamel to learn the Pleistocene distribution of C_3 vs. C_4 grasses over the range of latitudes spanning Alaska to Argentina. They reported a $\delta^{13}C$ gradient of 10 ‰, reflecting the changing proportion of C_3 to C_4 grasses between high and equatorial latitudes.

Stevens and Hedges (2004) analyzed the carbon and nitrogen isotope compositions of bone and tooth dentin collagen of horses that occupied northwest Europe between 40 ¹⁴C ka BP (~44.1 ka BP) and now. While glacial conditions existed 40 ¹⁴C ka BP (~44.1 ka BP) years ago in northwest Europe (British Isles, Belgium, and Germany), steppe and tundra environments were present in areas not occupied by glacial ice (Stevens & Hedges, 2004 and references therein). Stevens and Hedges (2004) observed a decrease in the δ^{13} C values of horse collagen towards the present time, which they attributed mainly to an increase in atmospheric CO₂ concentrations over the same period. This is supported by the observed relationship between decreasing plant δ^{13} C values and increasing CO₂ concentrations (e.g. Polley et al., 1993). Stevens and Hedges (2004) also observed a post-glacial lowering in horse δ^{15} N compared to pre-glacial values, followed by an increase around the onset of the Holocene. They suggested that the lower δ^{15} N values in the British Isles and Germany relative to Italy (and France to a lesser extent) could be the result of changing temperatures and the resulting interrelated changes in other factors (e.g. water availability, soil nitrogen cycling, etc.). They also suggested that there could be a delay of possibly a few thousand years in the response of plant (and therefore bone collagen) δ^{15} N values to retreating glacial conditions (e.g. post-glacial soil development, increased water availability because of increased permafrost melting rates, etc.).

Feranec et al. (2009) analyzed the carbon and oxygen isotope compositions of bioapatite from Late Pleistocene horse and bison tooth enamel from Rancho La Brea in southern California to learn how herbivores can partition and compete for resources within an ecosystem. They predicted, based on data from modern horse and bison, that the isotopic compositions for fossil horse and bison would be characteristic of grazers with some browse in their diets. Their oxygen isotope data revealed a sinusoidal seasonal growth pattern (see section 1.2.2) for enamel, while their carbon isotope data suggested a weaker seasonal pattern of dietary preferences. Both herbivores ate mainly C_3 plants, but the bison ate more C_4 plants than the horses. Relative to the horses, the bison exhibited more variation in their carbon isotope compositions, suggesting that they migrated away from Rancho La Brea for part of the year. In fact, the bison ate more C_4 plants in the winter and only ate from the C_3 dominated ecosystem of Rancho La Brea in the late spring.

Stevens et al. (2009) analyzed the carbon and nitrogen isotope compositions of horse bone collagen to reconstruct the past environment at two Belgian cave sites, Trou de Chaleux and Goyet, located about 30 km apart. Very limited radiocarbon dating (e.g. Hedges et al., 1993; Germonpré, 1997; see Table 1 in Stevens et al., 2009, for complete list) had suggested that these sites were occupied at about the same time during the Late-Glacial interstadial (~14.6 ka BP) (Svensson et al., 2006; Weaver et al., 2003). Stevens et al. (2009) predicted that if the sites were occupied at the same time then the nitrogen and carbon isotope compositions of the horse collagen from those two sites should be similar. However, the carbon and nitrogen isotopic results presented by Stevens et al. (2009) were different, suggesting that the two populations had different diets acquired in different environments. Additional radiocarbon dates verified that horse bones from the Trou de Chaleux cave represented the Late-Glacial interstadial, whereas those from Goyet Cave were 32-27 ¹⁴C ka BP (~36.4–31.4 ka BP) in age, mixed together with Late-glacial (between the Last-Glacial interstadial and the Holocene) horse remains.

Numerous other stable isotope studies have also been performed on Pleistocene horse fossils (e.g. Wang, 1994; Chillón et al., 1994; Higgins & MacFadden, 2004). There are also as many studies on 'non-Pleistocene' horse fossils (e.g. Hoppe et al., 2004a; Hoppe et al., 2004b; Bryant et al., 1996c; Huertas et al., 1995; Bryant et al., 1994; Chillón et al., 1994); these investigate a suite of topics ranging from the relationship between the isotopic composition of modern vs. Miocene horses and their environments to the timing of enamel mineralization in modern equid cheek teeth.

1.5 Background - Pleistocene Alberta

Alberta, Canada, was one of the areas where horses roamed in North America during the Pleistocene. Alberta is located along the eastern slopes of the Cordillera mountain range, adjacent to the northern Great Plains. Alberta was central to large-scale Late Pleistocene glacial events that affected many North American megafauna (Burns, 2010, Fig. 1.3). Burns (1996) conducted a study on the megafauna population dynamics during the Late Pleistocene. They recovered thousands of Pleistocene megafauna bone and tooth specimens, including twenty mammal species, mainly grazers, from a number of gravel sites in the Edmonton, Alberta, region. Radiocarbon dates from bone collagen samples indicated that many of the animals (e.g. horse, mammoth, etc.) were of Mid-Wisconsinan age (non-glacial, ~30–25 ¹⁴C ka BP (~34.7–29.6 ka BP)), just before the LGM. This suggests that this was a productive time period in Alberta, during which it contained vast grasslands (Burns, 1996). During this time horses had the highest abundance among the three most common megafauna in the region (bison, horse, mammoths) (Jass et al., 2011).

These megafauna were also present in the region after the LGM, indicating that 'mammoth steppe' like communities, which had existed in Alberta prior to the LGM, had returned (Burns, 1996). Therefore, the megafauna occupied Alberta throughout the Wisconsinan, but during the LGM when Alberta was glaciated, they were present only in unglaciated areas like eastern Beringia and south of the ice sheets (Burns, 1996). After the LGM, several megafauna species migrated into southern and central Alberta from the unglaciated United States as the Cordilleran and Laurentide ice sheets slowly separated (depending on when exactly they separated, Alberta may have also provided an 'ice-free' passage for humans to use to move from Beringia to south of the ice sheets, see section 1.1). These megafaunal species included bison, muskox, camel, horse, mammoth, and lion. At the same time, populations of bison, moose, and wapiti were moving south to the United States from Beringia (Shapiro et al., 2004; Burns, 2010). Bison became the most abundant megafauna in Alberta after the LGM. The 'ice-free' passage in western Alberta contained mainly herb and shrub vegetation, which supported grazing megafauna like the bison. But towards the Holocene, this vegetation was replaced by boreal forests that spread northward from the eastern plains and westward mountains. This likely impeded the northward migration of some megafauna by 10 ¹⁴C ka BP (~11.6–11.4 ka BP), for example *Bison* (MacDonald & McLeod, 1996; Burns, 2010).



Figure 1.3 Location of Alberta, Canada under glacial ice during the LGM. The red circle identifies the Edmonton region, from where the Pleistocene horse samples were excavated (modified from Munyikwa et al., 2011).

Despite its central location during past glacial episodes, and its importance as a possible migration route for animals via the 'ice-free' passage, there is a paucity of literature on the isotopic compositions of Alberta's Pleistocene-Holocene megafauna (with exception of Bocherens et al., 1994) and the changing environment during this time. Isotopic analysis of the Pleistocene horses in this study will help address some of these topics, including if and how ecosystem changes affected the horses, especially in terms of habitat quality and plant availability.

Chapter 2

Materials and Methods

2.1 Samples

Ten Late Pleistocene horse teeth and bone (from the mandible, below the incisors) samples were obtained from the Royal Alberta Museum (RAM). The samples had been excavated from gravel pits in the Edmonton region, near the North Saskatchewan River (Fig. 1.3, Table 2.1). In addition, a mandible from a modern domesticated horse was obtained from Atwood Pet Food Supplies LTD, a deadstock removal company in Atwood Ontario. The modern horse sample was used as a control for comparing teeth and bone isotopic compositions of the Pleistocene horse samples. Teeth that mineralized after weaning (see section 1.3.3) were selected for isotopic analysis and were labelled according to the Triadan system (Fig. 1.2): 308 (LP4), 408 (RP4), 306 (LP2), 406 (RP2).

| LSIS Label | RAM Label | Pit Location | Tooth |
|-------------|------------|-------------------------|------------|
| RAM 1 B, T | P97.11.2A | Riverview Pit | 306 (LP2) |
| RAM 2 B, T | P96.3.13 | Cloverbar Sand & Gravel | 306 (LP2) |
| RAM 3 B, T | P96.17.12A | Island Bluff | 306 (LP2) |
| RAM 4 B, T | P02.10.4 | Pit #48 | 108 (URP4) |
| RAM 5 B, T | P99.3.163 | Pit #48 | 406 (RP2) |
| RAM 6 B, T | P05.10.103 | Pit #48 | 406 (RP2) |
| RAM 7 B, T | P94.8.79 | Riverview Pit | 406 (RP2) |
| RAM 8 B, T | P94.1.380 | Pit #48 | 306 (LP2) |
| RAM 9 B, T | P04.3.33B | Pit #48 | 406 (RP2) |
| RAM 10 B, T | P94.2.33 | Pit # 80 | 406 (RP2) |

 Table 2.1
 Sample information

Samples: Laboratory for Stable Isotope Science (LSIS) labels; RAM labels are the Royal Alberta Museum catalogue numbers

Materials: B= bone, T= tooth

Tooth designations: L= left, R= right, U=upper (all other teeth are from the lower mandible)

2.2 Sampling techniques

The first sampling problem faced was to determine whether or not the teeth obtained from the modern horse were deciduous or adult. This question arose because, upon their removal from the mandible, it was evident that they had not fully mineralized (the enamel did not fully extend to the root, Fig. 2.1). Confirmation that the teeth were adult (and hence not influenced by weaning) is based on several observations. First, the permanent 06s and 08s erupt at 2.5 and 4 years, respectively. The 11s (M3s), which had erupted in the jaw, erupt at 3.5 years (with no deciduous precursor). Therefore, since the 11s were present in the jaw the permanent 06s and 08s should have erupted as they have coinciding eruption times. Second, radiographs of the mandible, made using a Faxitron 43855D X-ray, showed that there were not any teeth below the 06s and 08s, implying that they were permanent premolars and not deciduous (Fig.2.2). Third, the overlapping mineralization times of 06s, 08s, and 11s support the labelling of 06s and 08s as permanent teeth; the 06s and 08s mineralize from 13–31 months and 19–51 months respectively, and the 11s mineralize from 21–55 months.



Occlusal surface

Figure 2.1 Modern horse tooth 308 (LP4). Dashed line marks the outline of the enamel layer, illustrating that it does not extend all the way to the root. Sample lines are shown within the inner enamel (IE) layer.



Figure 2.2 Radiograph of left horse mandible. There are no deciduous teeth in the jaw below the permanent teeth. Scale bar=8.5 cm.

Bulk (enamel pieces representing the length of the tooth, i.e. more than one growing season) samples were collected and ground using a mortar, pestle, and a Dremel tool. The next sampling problem faced was how best to obtain serial samples from the Pleistocene and modern horse teeth in order to examine seasonal variation. There have been several studies on the use of serial sampled enamel δ^{18} O values as a tool to track seasonal variations during enamel mineralization (e.g. Sharp & Cerling, 1998; Fricke et al., 1998; Balasse, 2003a; Frémondeau et al., 2012). Most sampling techniques for studies of seasonality using stable isotope compositions of enamel involve drilling a sequence of horizontal lines on the outer enamel surface perpendicular to the height of the tooth, starting from the crown to the root (Balasse, 2003b). An alternate sampling strategy that samples the innermost enamel (IE) layer was used in this thesis. This strategy is preferred because the innermost enamel layer is heavily mineralized during matrix secretion, therefore samples obtained represent a more specific time period of growth, whereas the other method collects samples across a longer period of growth. For this reason, sampling the IE layer limits isotopic signal damping of the compositions obtained by body water and therefore ingested water (Suga, 1982; Balasse, 2003b; Zazzo et al., 2005). In addition, Hoppe et al. (2004a), using radiographic analyses, observed that the mineralization front in modern equine cheek teeth is more or less parallel to the Retizius striae (which progress outward from the enamel-dentin junction, EDJ, at an angle of 5- 10° and runs almost parallel to the height of the tooth, Fig. 1.1). Therefore, inner enamel (IE) samples with a lengthwise preference were taken along the height of the tooth.

Prior to analysis, the occlusal surface of all Pleistocene teeth was removed using a Buehler saw and preserved for future dental morphology or microwear studies. The Pleistocene and modern teeth were then cut along the lingual-buccal axis (Fig. 2.3 A), perpendicular to the long axis of the mandible. The teeth were cut in this way to expose the mesial and buccal aspects of the modern teeth as they contained the most enamel mineralization. In order to maximize the amount of enamel for serial sampling in the modern teeth and to be consistent, all teeth were cut so that the sampling surface was orientated in the mesial direction (Fig. 2.3 B), on the buccal side.



Figure 2.3 A) Location of 06 and 08 mandibular cheek teeth. Teeth were cut along buccal-lingual axis (dental orientations are shown) before isotopic analysis. B) Position of cut tooth on drill stage; mesial side of tooth is orientated towards the stage for serially sampling. Enamel layer is marked in red (modified from Hillson, 2005 and Metcalfe, 2012).

The mesial half of the tooth was mounted on a microdrill plate, IE orientated upwards. Lines were drilled on the IE surface using a Merchantek MicroMill. The inner two-thirds of the enamel was sampled (except where sample limitations required sampling most of the enamel layer) along lines with the following general dimensions: 400–600 μ m wide (lingual-buccal direction), ~2500 μ m long (occlusal-basal direction), and 720–900 μ m deep (mesial-distal direction), with sample-specific adjustments made depending on enamel variations in teeth (Fig. 2.4 A). The drill lines were separated from each other using a spacing needed to collect material representing a few months of mineralization. Based on observations made by Hoppe et al. (2004a), 06-teeth mineralize for ~18 months, and 11-teeth mineralize for ~32 months. Hence, a monthly growth rate could be estimated by dividing the length of the tooth (after occlusal surface was removed) by the mineralization rate of the tooth. Accordingly, drill lines were separated from each other by 1090 to 3000 μ m, depending on growth rate (Fig. 2.4 B).

Drilled enamel powder was transferred to glass vials by dabbing a fine-tipped paintbrush into the sample; the paintbrush was cleaned between samples with pressurized air, distilled water, and acetone. Micromilled samples were not pretreated in any fashion prior to isotopic analysis; this allowed analysis of smaller sample sizes (and therefore less averaging of the growth period represented by the sample). Enamel is much more resistant to isotopic alteration than bone (see below), making pretreatment less important for enamel than bone. There was also less difference between pretreated and untreated isotopic compositions for enamel samples compared to bone (see section 3.4.1).



Figure 2.4 A) Example of a sample line within the inner enamel (IE) layer; RAM 3T, line 4 (from occlusal surface). Black lines mark the division between outer cementum, enamel, and dentin. B) An example of sampling line spacing within the IE layer; RAM 2T lines 7 and 8. Some spacing was maintained between enamel and dentin layers to ensure dentin was not sampled.

2.3 Diagenesis

It is important to verify that bone and teeth samples of palaeontological material have retained their original isotopic compositions, both for bioapatite and collagen components. There are several *post-mortem* mechanisms that can alter the isotopic composition of bioapatite. These processes include chemical weathering, precipitation of secondary minerals, adsorption of ions at the surface of bioapatite crystals, and isotopic exchange during dissolution and re-precipitation (Zazzo et al., 2004a). This alteration (referred to as diagenesis) of bioapatite can reduce its usefulness as a tool for

reconstructing palaeodiet and palaeoenvironmental conditions from its isotopic compositions (Munro et al., 2007). Zazzo et al. (2004a), for example, showed that bone and enamel crystallites are susceptible to varying degrees of carbon and/or oxygen isotopic exchange during early diagenesis, depending on the presence or absence of microbial activity to facilitate the process. They also showed that enamel was more resistant to isotopic alteration than bone or dentin. This difference occurs because bones and dentin consist of smaller crystallites (with more surface area exposed for reaction), are more porous/permeable, and have a higher organic content than enamel. Consequently, bone and dentin can be much more susceptible to recrystallization and isotopic alteration than enamel (Kohn & Cerling, 2002; Zazzo et al., 2004a). The degree of isotopic alteration affecting different chemical constituents, specifically the structural carbonate $(CO_3^{2^-})$ and phosphate $(PO_4^{3^-})$ groups within bioapatite crystals, can also vary. In the absence of microbial effects, $PO_4^{3^-}$ oxygen is less affected by isotopic exchange than $CO_3^{2^-}$ oxygen because of the greater strength of the P-O bonds relative to C-O bonds (Kohn & Cerling, 2002; Zazzo et al., 2004a).

There are several methods that can be used to estimate the degree of alteration affecting bioapatite. One of these is the Crystallinity Index (CI), which provides a measure of the degree of order in the bioapatite structure (Surovell & Stiner, 2001). Because bioapatite crystals are small in bone and dentin, any fossil samples of these materials containing large crystals and a more ordered structure have almost certainly undergone *post-mortem* recrystallization. The CI values can be used as a semiquantitative measure of such recrystallization (Shemesh, 1990; Wright & Schwarcz, 1996; Munro et al., 2007 and references therein). Fourier Transform infra-red (FTIR) spectroscopy is typically used to measure CI.

To obtain FTIR analyses in this study, 2 mg of bone or enamel powder (sieved to a particle size of 45–63 μ m) were mixed with 200 mg of KBr, and then placed under 10 tons of pressure for 10 minutes. The resulting pellet was then scanned using a Bruker Vector 22 FTIR spectrometer and the resulting spectrum interrogated using the spectral analysis software provided with the instrument. In this method, crystallinity is determined from the degree of PO₄⁻³ band splitting and calculated as follows:

where $A_{x'}$ is the absorbance at wave number 'x', assuming a straight baseline between 750 and 450 cm⁻¹ (Weiner & Bar-Yosef, 1990; Shemesh, 1990; Munro et al., 2007 and references therein). FTIR CI values typically range from ~2.50–3.25 for modern bone, and from ~3.40–4.50 for archaeological bone (Munro et al., 2007 and references therein). Bioapatite samples with FTIR CI values less than 3.8 are considered unaltered (Shemesh, 1990).

A second test for bioapatite *post-mortem* alteration is provided by powder X-ray diffraction (pXRD), which was used in this study to confirm the presence of carbonated hydroxylapatite (bioapatite) in the bone and tooth samples, and to check for the presence of inorganic contaminants, such as secondary calcite, that can precipitate in void spaces of bone or teeth. The pXRD data were collected using a Rigaku RU-200BVH rotating anode diffractometer, operated in step-scan mode and employing Co K α radiation (λ =0.1790210 nm) at 45 kV and 160 mA. The measurements were collected from 28 to 44° 20, using a step size of 0.010° and a scan speed of 1.000°min⁻1.

The degree of correlation between the δ^{18} O values of co-existing phosphate and structural carbonate from bone or enamel can also be used to test the isotopic integrity of bioapatite samples. Iacumin et al. (1996) analyzed 31 bone and tooth samples of modern mammals for their structural carbonate and phosphate oxygen isotope compositions and found them to be positively correlated along a so-called 'equilibrium' line. They suggested that the δ^{18} O values of samples plotting on or close to this line represent unaltered material. Since then, however, Martin et al. (2008) and Pellegrini et al. (2011) have demonstrated that the primary oxygen isotope compositions of structural carbonate and phosphate do not always conform to such a strict relationship at the microscopic scale.

Information obtained from collagen analysis can also test the quality of bone. Well-preserved collagen has a yield >1 %, a carbon content of 30–43 % and a nitrogen content of 11–16 % (Ambrose, 1990; van Klinken, 1999). In addition, the atomic C/N ratio of collagen can be used to assess preservation; the C/N ratio of well-preserved collagen falls between 2.9–3.6 (DeNiro, 1985).

[Equation 2.1]

2.4 Stable isotopes

Stable isotope compositions are presented in standard delta (δ) notation in per mil (‰) relative to VPDB for carbon, VSMOW for oxygen and AIR for nitrogen (Coplen, 1996, 2011; Hoefs, 2009).

2.4.1 Phosphate preparation

In order to analyze the oxygen isotope composition of the phosphate component of bioapatite (from bulk enamel and bone samples), the bioapatite samples had to be purified and precipitated as Ag_3PO_4 . This was done following a procedure adapted from Firsching (1961), Crowson et al. (1991), and Stuart-Williams and Schwarcz (1995). Approximately 15 ml of 3 M acetic acid was added to 45<63 µm powdered bone (34 mg) or enamel (29-32 mg) for a minimum of 24 hours. To remove calcium from bioapatite, 0.55 M potassium oxalate was added to each sample and the pH was adjusted to 3.5-4.0using 8 M potassium hydroxide (KOH) to precipitate calcium in the form of calcium oxalate. Samples were then centrifuged and the supernatant solution retained and reacted with 5 mL of 0.5 M lead acetate. The pH was then adjusted to 5.3–5.5 with 8 M KOH and the samples were centrifuged to isolate the lead phosphate precipitate. This white precipitate (in each sample tube) was placed in a 95 °C water bath and 2 treatments of 1 mL of concentrated nitric acid and 1.5 mL of peroxide were added to remove any traces of K-oxalate and organics. After the peroxide had finished reacting, the samples were washed with 1 mL of distilled water and evaporated to low liquid levels. This wash was repeated 3 times to ensure that all peroxide had been removed. At this point, 8 M KOH was then added to neutralize the samples until the precipitate dissolved, and once it dissolved 3 mL of 0.5 M lead acetate was added. The pH was adjusted to 5.5–5.7 using 8 M and 4 M KOH. After reaction, the samples were centrifuged, the precipitate retained, and 2 mL of 0.25 M nitric acid added. The remaining lead-phosphate precipitate was dissolved to completion by dropwise addition of 3 M nitric acid. Once dissolved, 2 mL of 0.25 M ammonium sulphate was added to remove any remaining lead. The samples were then reacted, centrifuged, and the supernatant retained. Two or three drops of
bromothymol blue solution were added with 5 mL of distilled water, and the pH was adjusted to 5.5–6.5 with 4 M KOH.

The next day, 0.6 g of silver nitrate was dissolved in 10 mL of distilled water for each sample and ammonium hydroxide was added until the (ammoniacal) solution turned clear. Then, 1 mL of ammonium hydroxide, 1.5 mL of ammonium nitrate, and 10 mL of the ammoniacal silver solution (silver nitrate, water, NH₄OH) were added to each sample. Sample volumes were each topped up (and maintained) at a 60 mL volume with millipore water, and were left to precipitate on a hotplate for 6–6.5 hours at 55 °C. The precipitate was then cooled to room temperature and the resultant Ag₃PO₄ crystals were filtered and washed 3 times with millipore water. Samples were dried overnight at 65 °C and then were ready to be weighed for isotopic analysis.

2.4.1.1 Phosphate analysis: BrF₅ method

About 20-25 mg of the precipitated silver phosphate samples was loaded into Ni reaction vessels. Samples were loaded into the reaction vessels within a drybox (under flow of dry N_2 gas). A dry filter and fresh phosphorous pentoxide (P_2O_5) powder was put into the drybox during loading to maintain a relative humidity of < 13 %. Once loaded, samples were heated under vacuum for 2 hours at 300 °C, following which 80 mmHg of bromine pentafluoride (BrF_5) was added to each reaction vessel, which was then closed. Furnaces were also added to each reaction vessel, which were then heated at 600 °C overnight. Using an extraction line and procedures modified after Clayton and Mayeda (1963), Crowson et al. (1991), and Stuart-Williams and Schwarcz (1995), the O_2 released from the silver phosphate by high temperature reaction with BrF₅ was converted quantitatively to CO₂ by reaction with incandescent carbon. The sample gas was isolated cryogenically in order to calculate its yield (µmol/mg). The pressure of the gas (in a known volume) was measured with an electronic gauge and then the displacement of the gas was measured after it was transferred to evacuated sample tubes for oxygen isotopic analysis using a VG Prism II, dual-inlet, triple-collecting, isotope-ratio mass spectrometer (IRMS).

The theoretical yield of CO₂ produced from Ag₃PO₄ is 4.78 μ m/mg. During the course of these experiments, standard yields averaged 4.87 μ m/mg (n=7), and sample

yields averaged 4.96 μ m/mg (n=22). The oxygen isotopic results were calibrated to VSMOW following Coplen (1996), using a 2-point curve based on the following standards and accepted values: TU-1 (+21.11 ‰) and TU-2 (+5.45 ‰). Using this curve, a δ^{18} O value of +11.25 ± 0.22 ‰ (n=5, SD) was obtained for the internal laboratory phosphate standard (Aldrich 2), which compares well with its accepted value of +11.2 ‰. The one set of duplicate samples were reproducible to ± 0.13 ‰ of the average.

2.4.1.2 Phosphate analysis: TC-EA method

The δ^{18} O values of phosphate samples were also determined using a ThermoFinnigan TC-EA coupled to a ThermoFinnigan Delta Plus XL IRMS. Powdered samples in silver capsules (~0.4 mg) were introduced into the TC-EA using a Zero Blank auto sampler and reacted rapidly for a few seconds via pyrolysis at 1350 °C. The carbon monoxide gas produced was passed through a homemade GC column, packed with a 5 Å molecular sieve heated to 90 °C, to eliminate any impurities present (e.g. water vapour, etc.), and then transferred by helium flow to the IRMS for analysis. The δ^{18} O values were calibrated to VSMOW following Coplen (1996), using a 2-point curve based on the following standards and accepted values: Aldrich 2 (+11.2 ‰) and IAEA-CH-6 (+36.4 ‰). An average value of +21.8±0.4 ‰ (n=15, SD) was obtained for NBS 120c during these experiments, which compares well with its accepted value of +21.7 ‰.

2.4.2 Structural carbonate analysis

Bone and teeth samples were gently scrubbed with a wire brush to remove adhering dirt, rinsed with distilled water, and dried at room temperature. The bulk enamel and bone samples were then ground with a mortar and pestle to fine powder and sieved to less than 180 µm. In order to remove organic material, a portion of the bulk samples was treated with 0.04 ml of 2 % reagent-grade bleach per mg of sample at room temperature for 24 and 72 hours, for enamel and bone, respectively, and rinsed five times with deionized (Millipore) water. (Another sample portion was untreated to allow for comparisons with unbleached incremental samples; see below). To remove any secondary carbonates acquired during *post-mortem* processes, 0.1 M acetic acid was added in a ratio of 0.04 ml per mg of sample for 4 hours at room temperature. The samples were then rinsed five times with Millipore water and freeze-dried (Garvie-Lok et al., 2004; Koch et al., 1997). A Micromass MultiPrep auto sampler was then used to react ~0.5–1 mg of bone or enamel with H_3PO_4 at 90 °C for 25 minutes and the gas evolved analyzed using a VG Optima dual-inlet IRMS, following the method of Metcalfe et al. (2009).

The structural carbonate δ^{13} C values were calibrated to VPDB following Coplen (1994), using a 2-point curve based on the following standards and accepted values: NBS-19 (+1.95 ‰) and Suprapur (-35.28 ‰). The average δ^{13} C values obtained for NBS-18 and the internal laboratory standard WS-1 were -4.95 ± 0.19 ‰ (n= 22, SD) and +0.75 ± 0.17 ‰ (n=14, SD), in close agreement with their accepted values of -5.0 and +0.76 ‰, respectively. The δ^{18} O values were calibrated relative to VSMOW following Coplen (1996), using a 2-point calibration curve based on the following standards and accepted values: NBS-19 (+28.60 ‰) and NBS-18 (+7.20 ‰). The average δ^{18} O values of internal standards Suprapur and WS-1 were +13.22 ± 0.25 ‰ (n=20, SD) and +26.32 ± 0.25 ‰ (n=14, SD), in close agreement with their accepted values of +13.20 and +26.23 ‰, respectively.

2.4.3 Collagen analysis

A variation of the procedure described by Longin (1971) was used to extract collagen from the bone samples. The bone samples were mechanically cleaned, ground, and sieved, retaining the fraction between 0.18 and 0.85 mm. Lipids were removed from the bone sample using 8 mL of 2:1 chloroform:methanol solution. To dissolve the inorganic (mineral) phase of bone, 0.25–0.5 M HCl was added to each sample at room temperature. Humic and fulvic acids were removed using 0.1 M NaOH at room temperature and then rinsed ~8 times with distilled water. The collagen was gelatinized in 0.25 M HCl at 90 °C and then dried at 90 °C. The collagen samples were lightly ground and weighed into tin capsules (0.400 ± 0.02 mg). The δ^{13} C and δ^{15} N values and C/N ratios were analyzed using a Costech elemental combustion system (ECS 4010) coupled to a ThermoFinnigan Delta Plus XL IRMS (operated in continuous flow mode).

The δ^{13} C values of collagen were calibrated to VPDB following Coplen (1994) using a 2-point curve based on the following standards and accepted values: USGS-40 (–

26.39 ‰) and USGS-41 (+37.63 ‰). The δ^{15} N values were calibrated to AIR using a 2point curve based on the following standards and accepted values: USGS-40 (-4.52 ‰) and USGS-41 (+47.57 ‰). Nitrogen and carbon contents, which were then used to calculate atomic C/N ratios, were also calibrated using the accepted abundances in USGS-40 and USGS-41 (n= 9.5 and C= 40.7 %). An internal standard (Keratin), and international standards IAEA-CH-6 and IAEA-N2 were included in each run to monitor the accuracy and precision of measurements. The results for Keratin were: δ^{13} C = -24.04 ± 0.15 ‰ (n=11, SD; accepted = -24.04 ‰); δ^{15} N = +6.25 ± 0.23 ‰ (n=11, SD; accepted = +6.36 ‰; C/N measured= 3.78 ± 0.06 (n=12, SD; theoretical: 3.4, O'Connell & Hedges, 1999). The results for IAEA-CH-6 were: δ^{13} C = -10.47 ± 0.13 ‰ (n=5, SD; accepted = -10.45 ‰). The results for IAEA-N2 were: δ^{15} N = +20.40 ± 0.17 ‰ (n=4, SD; accepted = +20.30 ‰). Collagen C/N ratios of samples were reproducible to better than ± 0.01 (n=9, SD).

2.5 Drinking water oxygen isotope composition

The δ^{18} O values of drinking water were calculated for the Pleistocene and modern horses using the following approach. Phosphate δ^{18} O values were converted to drinking water composition using the relationship of Huertas et al. (1995), who combined their data with the δ^{18} O results of Bryant et al. (1994) and Chillón et al. (1994):

$$\delta^{18}O(PO_4^{3-})=0.71[\delta^{18}O(H_20)]+22.60,$$
 [Equation 2.2]

To obtain drinking water oxygen isotopic compositions from structural carbonate δ^{18} O values, they first had to be converted to phosphate δ^{18} O values (assuming that both components formed in isotopic equilibrium with each other; see section 1.2.2) using the fractionation factor between structural carbonate (SC) and phosphate (P) for mammalian systems (Bryant et al., 1996a, Iacumin et al., 1996):

$$\alpha_{\text{SC-P}}(1.0086) = (1000 + \delta_{\text{SC}})/(1000 + \delta_{\text{P}}),$$
 [Equation 2.3]

2.6 Radiocarbon analysis

Four Pleistocene horses were radiocarbon dated at the University of Arizona Accelerator Mass Spectrometry (AMS) Laboratory. Bone collagen was extracted using an automated flow-cell apparatus, following a modification of the technique used by Longin (1971), and combusted with copper oxide at >800 °C. Radiocarbon dates were calculated using the Libby half-life (Libbey et al., 1949). The dates were corrected for natural and ion source fractionation using measured ${}^{13}C/{}^{12}C$ ratios for bone collagen (Stuiver & Polach, 1977). Conversion to calibrated years (ka BP) from uncalibrated radiocarbon years was performed using INTCAL09 (Reimer et al., 2009).

2.7 Defleshing considerations

The modern horse sample needed to be defleshed before isotopic analysis could be completed. The mandible was cut in half (between the incisors), the left mandible was put into water at 20 °C, and the right mandible was put into water at 29 °C for a total of 17 days. After the first week, most of the tissue was scraped off, and the jaws were put back into clean water at their respective temperatures for the remainder of defleshing time. Two different water temperatures were used to determine which approach removed adhering tissues better (i.e. 'cold' or 'warm' water). Tests were conducted on the modern horse mandible to ensure that isotopic changes did not occur during defleshing. The mandibular 'hinges' were used for testing three defleshing techniques: cold water (~20 °C), warm water (~29 °C), and warm water with an enzyme (30 mL Protease at 29 °C). It was learned later, however, that the enzyme was past its expiration date and therefore unlikely to have much effect on tissue breakdown.

Bone samples from the modern mandible used for defleshing tests were analysed using the BrF₅ method for phosphate (Table 2.2 and Fig. 2.5). The piece of mandible put in cold water (CW) had a phosphate oxygen isotope composition of $+11.5 \pm 0.4 \%$ (n=2). The samples put in warm water (WW) and water with the enzyme (E) both had a δ^{18} O phosphate value of +12.1 %. (Table 2.2). No difference was expected between WW and E bones because they were both from the hinge part of the right mandible (Fig. 2.5) and the enzyme was past its expiration date and expected to behave similar to the WW experiment. Given that the Pleistocene bone samples were taken from the mandible below the incisors, a sample of the modern horse bone was also taken from this position from the right mandible (Fig. 2.5) that had been defleshed in warm water (sample MH). It had a δ^{18} O phosphate value of +11.3 ‰, as determined using the TC-EA method. The δ^{18} O phosphate values of all defleshed bone samples were within ±0.4 ‰ (SD). This range is acceptable given the inherent isotopic variability that can exist within the same bone sample. For example, collagen isotopic compositions of bone samples from the same human rib can differ on average by 0.1 and 0.3 ‰ for δ^{13} C and δ^{15} N values, respectively (K. Olsen, personal communication).

Table 2.2 Phosphate and structural carbonate (pretreated and untreated) bone isotopic compositions of the defleshing tests.

| | | Carbonate | | | | | | | | | | |
|--------|-------------------|-------------------|-------------------|-------------------|-------------------|--|--|--|--|--|--|--|
| | Phosphate | Untr | eated | Pretr | eated | | | | | | | |
| | δ ¹⁸ Ο | δ ¹⁸ O | δ ¹³ C | δ ¹⁸ O | δ ¹³ C | | | | | | | |
| Sample | (%0) | (‰) | (‰) | (‰) | (‰) | | | | | | | |
| WW | +12.1 | +19.3 | -15.0 | +19.8 | -14.9 | | | | | | | |
| Е | +12.1 | +19.4 | -14.9 | +19.8 | -15.0 | | | | | | | |
| CW | +11.5 | +19.9 | -14.8 | +20.0 | -15.3 | | | | | | | |

Values in **bold** are the average of duplicate analyses



Figure 2.5 Diagram of a horse mandible showing modern bone and teeth samples. CW, WW, and E were defleshing tests and represent cold water, warm water, and enzyme experiments, respectively (see section 2.7). The right mandible was defleshed in warm water and the left mandible was defleshed in cold water. MH is the bone sample for the modern horse (modified from Bendrey, 2007).

The carbon and oxygen isotope compositions of structural carbonate were also analyzed from the WW, E, and CW defleshed samples (Fig. 2.5). Untreated WW, E, and CW had an average δ^{13} C value of -14.9 ± 0.08 ‰ (SD) and an average δ^{18} O value of $+19.5 \pm 0.34$ ‰ (SD) (Table 2.2). Pretreated WW, E, and CW had an average δ^{13} C value of -15.1 ± 0.21 ‰ (SD) and an average δ^{18} O value of $+19.9 \pm 0.11$ ‰ (SD) (Table 2.2). Given the limited range of carbonate and phosphate isotopic compositions (Table 2.2) and 'background variability', all results from the defleshing tests were incorporated into the isotopic baseline used to evaluate tissue isotopic spacings, which are discussed later in the thesis.

Chapter 3

Results

3.1 Assessment of diagenesis of bioapatite and associated collagen

FTIR CI bone and enamel results for the modern and Pleistocene horses are listed in Table 3.1. The modern horse bone has a CI of 2.7 and the modern teeth have an average CI of 3.3. All modern and Pleistocene horse FTIR CI bone and teeth values (except for RAM 3B) fall within the range of accepted values for modern and archaeological bone samples, 2.50–3.25 and 3.40–4.50, respectively (see section 2.3). The pretreated RAM 3B had a CI of 2.6, which is higher than its untreated CI of 1.7. The increase in CI (within the acceptable range) for RAM 3B was likely because of the removal of secondary carbonate during pretreatment. The horse bone and teeth CI values are similar to those reported by Trueman et al. (2004) for other mammals of similar size (2.7–3.4; cow, elephant, zebra, rhinoceros, wildebeest, and buffalo bone samples). All of the peaks in the horse bone and teeth FTIR and pXRD spectra are characteristic of carbonated hydroxylapatite; no calcite or unusual phases (e.g. siderite, brushite, etc.) were detected. The modern horse has a carbon content of 40.1 %, a nitrogen content of 14.7 %, and a C/N ratio of 3.19. The Pleistocene bone collagen has a carbon content of 36.6-39.4 % (n=9), a nitrogen content of 12.9-14.0 % (n=9), and a C/N ratio of 3.29-3.34 (n=9) (Table 3.1). All values are typical of well-preserved collagen (see section 2.3). The collagen yield for the modern horse is 23.5 %, whereas for the Pleistocene samples it ranges from 5.6 to 15.0 % (7.8 \pm 3.0 %, SD; n=9), well above the limit (\geq 1 %) for wellpreserved collagen. RAM 3B did not contain extractable collagen, again indicating poor bone preservation. As such, results for RAM 3B are excluded from the environmental interpretations that follow.

| | | | Bone | | | Teeth |
|------------------------|------|------|------|-------|-----|-------|
| - | FTIR | Ν | С | Yield | | FTIR |
| Sample | CI | (%) | (%) | (%) | C/N | CI |
| Modern | | | | | | |
| ^a MH (bone) | 2.7 | 14.7 | 40.1 | 23.5 | 3.2 | |
| 308(P4L) | | | | | | 3.4 |
| 408(P4R) | | | | | | 3.4 |
| 306(P2L) | | | | | | 3.1 |
| 406(P2R) | | | | | | 3.2 |
| Pleistocene | | | | | | |
| RAM 1 | 2.7 | 12.9 | 36.7 | 5.6 | 3.3 | 3.4 |
| RAM 2 | 2.7 | 13.5 | 38.7 | 6.2 | 3.3 | 3.4 |
| RAM 3 | 1.7 | | | | | 3.4 |
| RAM 4 | 2.7 | 14.0 | 39.4 | 15.0 | 3.3 | 3.2 |
| RAM 5 | 2.8 | 13.3 | 37.5 | 6.8 | 3.3 | 3.3 |
| RAM 6 | 2.7 | 13.8 | 39.1 | 7.8 | 3.3 | 3.3 |
| RAM 7 | 2.7 | 13.7 | 38.9 | 8.2 | 3.3 | 3.3 |
| RAM 8 | 2.8 | 12.9 | 36.7 | 8.1 | 3.3 | 3.1 |
| RAM 9 | 2.6 | 13.2 | 37.5 | 4.2 | 3.3 | 3.3 |
| RAM 10 | 2.6 | 13.4 | 37.7 | 8.3 | 3.3 | 3.2 |

Table 3.1 Horse bone and tooth diagenetic tests.

^aMH= modern horse bone

Dental arcade location of teeth is shown in brackets beside the Triadan sample label Results indicated in **bold** are the average of duplicate analyses

3.2 Oxygen isotopic analysis of bioapatite phosphate

3.2.1 BrF₅ method versus TC-EA

The phosphate oxygen isotopic compositions ($\delta^{18}O_p$) of bone and enamel from the modern and Pleistocene horses are listed in Table 3.2, as produced using both the TC-EA and BrF₅ methods. Both methods produced comparable results for standard materials (see sections 2.4.1.1 and 2.4.1.2). For samples, replicate analyses are reproducible to \pm 0.2 ‰ (n=15; SD) using the TC-EA method, and \pm 0.1 ‰ (n=2, SD) using the BrF₅ method. For samples, there is no statistical difference between the oxygen isotopic results produced using the TC-EA and BrF₅ methods (bone phosphate; Independent-samples *t*-test, p=0.131; tooth phosphate; Independent-samples *t*-test, p=0.570). Accordingly, the results are pooled in the environmental and physiological interpretations that follow (averages from both methods are plotted in Fig. 3.1).

For future work, the TC-EA method is the preferred method for phosphate oxygen isotope analysis for several reasons: (1) A greater number of phosphate standards can be analyzed as unknowns within each analytical session, thus improving the ability to assess precision and accuracy; (2) More samples can be analysed in a shorter period of time (e.g. up to 18 samples in 1 day using the TC-EA versus no more than 5 samples in three days using the BrF₅ method); (3) greater potential for analytical error exists in the BrF₅ method, particularly from adsorption of water by samples and reaction vessels during sample loading; and (4) the BrF₅ method requires use of this highly hazardous (toxic, explosive) reagent.

Table 3.2 Horse bone and tooth $\delta^{18}O_p$ and associated drinking water oxygen isotope compositions ($\delta^{18}O_{dw}$).

| | Bone | | Те | eth | *Ave | erage | | | **W | ater |
|---------------------------|--------------------------------|---------------------------------------|--------------------------------|---------------------------------------|---------------------------|----------------------------|---|-------------------------|---------------------------|----------------------------|
| Sample | ΤC- ΕΑ δ ¹⁸ Ο | BrF ₅ δ ¹⁸ Ο | ΤC- ΕΑ δ ¹⁸ Ο | BrF ₅ δ ¹⁸ Ο | Bone δ ¹⁸ Ο | Teeth δ ¹⁸ Ο | Radiocarbon dates (¹⁴ C BP) | Radiocarbon lab # | Bone δ ¹⁸ Ο | Teeth δ ¹⁸ Ο |
| ^a MH (bone) | +113 | +11.8 | | | +11 5 | | | | -15.6 | |
| (00hc) 308(P4L) | 111.5 | 111.0 | +11.3 | +11.2 | 111.5 | +11.3 | | | 15.0 | -15.9 |
| 408(P4R) | | | +11.6 | +11.4 | | +11.5 | | | | -15.6 |
| 306(P2L) | | | +11.5 | +11.8 | | +11.7 | | | | -15.4 |
| 406(P2R) | | | +12.1 | +11.4 | | +11.8 | | | | -15.2 |
| Pleistocene | | | | | | | | | | |
| RAM 1 | +7.7 | +7.7 | +6.2 | +6.7 | +7.7 | +6.4 | 11 240 <u>+</u> 110 | AA97948 | -21.0 | -22.8 |
| RAM 2 | +9.7 | +8.4 | +6.5 | | +9.0 | +8.4 | 11 260 <u>+</u> 110 | AA97949 | -19.1 | -20.0 |
| RAM 3 | +9.6 | +9.1 | +8.6 | +8.5 | +9.3 | +8.6 | | | -18.7 | -19.8 |
| RAM 4 | +10.6 | +10.3 | +9.0 | +9.1 | +10.5 | +9.1 | | | -17.1 | -19.1 |
| RAM 5 | +10.3 | +9.7 | +8.3 | +8.4 | +10.0 | +8.4 | > 38,900 | AA97950 | -17.7 | -20.1 |
| RAM 6 | +11.2 | +11.5 | +7.5 | +7.6 | +11.3 | +7.6 | | | -15.9 | -21.2 |
| RAM 7 | +11.7 | +9.3 | +8.4 | | +10.5 | +9.3 | | | -17.0 | -18.7 |
| RAM 8 | +9.4 | +9.1 | +7.5 | | +9.3 | +9.1 | >46,700 | AA97951 | -18.8 | -19.0 |
| RAM 9 | +8.3 | +7.8 | +5.8 | | +8.0 | +7.8 | | | -20.5 | -20.8 |
| RAM 10 | +10.7 | +9.6 | +7.7 | | +10.1 | +9.6 | | | -17.6 | -18.3 |

*Calculated using the average $\delta^{18}O_p$ value obtained for all analyses, regardless of method ^aMH= modern horse bone

Dental arcade location of teeth is given in brackets beside the Triadan sample label Values in **bold** are the average of duplicate analyses

**Water δ^{18} O values are calculated using Equation 2.2; see section 2.5



Figure 3.1 Bone and tooth $\delta^{18}O_p$ values for the Pleistocene horses. Results are averaged from both methods of phosphate analysis.

3.3 Radiocarbon dating

Two Pleistocene bone samples, RAM 1B and 2B, have radiocarbon dates (11 240 \pm 110 and 11 260 \pm 110 ¹⁴C BP) that postdate the LGM (22–18 ¹⁴C ka BP; 26.2–21.4 ka BP). These samples will be referred to as post-LGM. The two other radiocarbon-dated samples, RAM 5B and 8B, have dates (>38 900 and >46 700 ¹⁴C BP) that precede the LGM (Table 3.2). These samples will be referred to as pre-LGM.

3.4 Oxygen and carbon isotopic analysis of bioapatite structural carbonate

3.4.1 Bulk structural carbonate

The bulk carbon ($\delta^{13}C_{sc}$) and oxygen ($\delta^{18}O_{sc}$) values of untreated and pretreated bone and enamel samples are listed in Table 3.3. Replicate analyses for untreated and pretreated bone $\delta^{18}O_{sc}$ values vary by $\pm 0.4 \%$ (n=4, SD) and $\pm 0.4 \%$ (n=6, SD), respectively. RAM 3B exhibits the poorest reproducibility both for treated ($\pm 0.7 \%$) and untreated ($\pm 1.5 \%$) samples, which is likely indicative of *post-mortem* alteration. These data for this sample are therefore excluded from subsequent discussions. The reproducibility of replicate $\delta^{13}C_{sc}$ analyses of bone is much better than for oxygen (pretreated, ± 0.2 ‰; n=6, SD; untreated, ± 0.2 ‰; n=4, SD). Duplicate tooth analyses for pretreated samples are within ± 0.2 ‰ for $\delta^{13}C_{sc}$ and ± 0.1 ‰ for $\delta^{18}O_{sc}$ (n=2, SD), while untreated duplicates are within ± 0.1 ‰ and ± 0.1 ‰ (n=3, SD) for $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$, respectively.

There is no statistical difference between untreated and pretreated Pleistocene enamel $\delta^{18}O_{sc}$ values (excluding RAM 3, Independent-samples *t*-test). The average difference between untreated and pretreated bone samples is $+1.2 \pm 0.6 \%$ for $\delta^{13}C_{sc}$ and $+1.4 \pm 1.5 \%$ for $\delta^{18}O_{sc}$ (excluding RAM 3B). The average differences between untreated and pretreated enamel $\delta^{13}C_{sc}$ values is $+0.9 \pm 1.0 \%$. There is a larger difference between untreated and pretreated bone $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values than for teeth. Bone is more susceptible to isotopic alteration than enamel (see section 2.3), and thus it is likely pretreatment had more effect on restoring original bone isotopic signals because their crystallites had absorbed more secondary carbonates. Accordingly, pretreated bone and enamel isotopic compositions ($\delta^{18}O_{sc}$ values are shown in Fig. 3.2) have been used to determine the diet and drinking water compositions listed in Table 3.3. Diet $\delta^{13}C$ values were calculated assuming a $\Delta^{13}C_{sc-diet}$ value of +14.1 ‰ (Table 3.3, see section 1.2.1)



Figure 3.2 Pretreated bone and tooth $\delta^{18}O_{sc}$ values of the Pleistocene horses.

| | | | Untr | eated | | Pretreated | | | | | | | | | | |
|------------------------|---|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------|--------------------------|---------------|----------------------------------|-----------------------------------|
| | | Bo | one | Те | eth | Bo | one | Те | eth | Colla | agen | Die | et δ ¹³ C, fr | om: | ^W | ater |
| Sample | Radiocarbon dates (¹⁴ C BP) | δ ¹³ C (‰) | δ ¹⁸ Ο (‰) | δ ¹³ C (‰) | δ ¹⁵ N (‰) | **Col (‰) | *Bone (‰) | *Teeth (‰) | Bone δ ¹⁸ Ο (‰) | Teeth δ ¹⁸ Ο (‰) |
| Modern | | | | | | | | | | | | | | | | |
| ^a MH (bone) | | -13.6 | +20.2 | | | -14.9 | +20.1 | | | -22.7 | +4.7 | -28.2 | -29.0 | | -15.7 | |
| 308(P4L) | | | | -14.2 | +20.3 | | | -14.6 | +19.9 | | | | | -28.7 | | -16.0 |
| 408(P4R) | | | | -14.2 | +20.8 | | | -14.4 | +20.3 | | | | | -28.5 | | -15.4 |
| 306(P2L) | | | | -14.4 | +20.4 | | | -14.6 | +20.0 | | | | | -28.7 | | -15.9 |
| 406(P2R) | | | | -14.2 | +19.3 | | | -14.3 | +20.2 | | | | | -28.4 | | -15.6 |
| Pleistocene | | | | | | | | | | | | | | | | |
| RAM 1 | 11 240 <u>+</u> 110 | -10.0 | +16.3 | -11.0 | +16.6 | -11.3 | +12.5 | -11.0 | +14.9 | -20.5 | +0.0 | -26.0 | -25.4 | -25.1 | -26.4 | -23.0 |
| RAM 2 | 11 260 <u>+</u> 110 | -8.5 | +16.4 | -11.0 | +15.7 | -9.2 | +14.6 | -11.6 | +15.1 | -20.9 | +0.2 | -26.4 | -23.3 | -25.7 | -23.5 | -22.7 |
| RAM 3 | | -14.0 | +14.5 | -12.8 | +16.2 | -13.6 | +4.2 | -13.3 | +15.7 | | | | -27.7 | -27.4 | -38.0 | -22.0 |
| RAM 4 | | -9.9 | +17.3 | -11.7 | +17.1 | -11.2 | +16.0 | -12.1 | +16.8 | -20.7 | +0.5 | -26.2 | -25.3 | -26.2 | -21.5 | -20.4 |
| RAM 5 | > 38,900 | -10.6 | +16.2 | -11.1 | +16.2 | -10.5 | +16.1 | -12.4 | +14.9 | -20.3 | +6.6 | -25.8 | -24.6 | -26.5 | -21.4 | -23.0 |
| RAM 6 | | -9.5 | +17.4 | -10.5 | +16.2 | -11.5 | +13.8 | -11.7 | +15.7 | -20.6 | +6.3 | -26.1 | -25.6 | -25.8 | -24.6 | -22.0 |
| RAM 7 | | -9.4 | +17.1 | -10.6 | +15.2 | -10.1 | +17.5 | -12.9 | +14.2 | -20.8 | +6.6 | -26.3 | -24.2 | -27.0 | -19.5 | -24.0 |
| RAM 8 | >46,700 | -10.3 | +16.3 | -12.2 | +13.7 | -11.9 | +16.1 | -13.2 | +15.7 | -21.2 | +5.4 | -26.7 | -26.0 | -27.3 | -21.4 | -21.9 |
| RAM 9 | | -11.2 | +16.2 | -11.2 | +14.8 | -12.3 | +13.5 | -11.5 | +13.7 | -20.8 | +6.2 | -26.3 | -26.4 | -25.6 | -24.9 | -24.7 |
| RAM 10 | | -9.4 | +16.4 | -10.5 | +15.6 | -11.3 | +15.9 | -11.1 | +15.3 | -20.3 | +7.0 | -25.8 | -25.4 | -25.2 | -21.6 | -22.5 |

Table 3.3 Stable isotopic compositions for structural carbonate (untreated and pretreated) and collagen, and corresponding diet and drinking water δ -values.

^aMH= modern horse bone

Dental arcade location of teeth is given in brackets beside the Triadan sample label

Values in **bold** are the average of duplicate analyses Radiocarbon ages reported as ¹⁴C age BP

* $\delta^{13}C_{sc}$ diet values are calculated assuming a $\Delta^{13}C_{sc-diet}$ value of +14.1 ‰ (Cerling & Harris, 1999) ** $\delta^{13}C_{col}$ diet values are calculated assuming a $\Delta^{13}C_{col-diet}$ value of +5.5 ‰ (Vogel et al., 1978; Sullivan & Krueger, 1981; van der Merwe, 1989; Ambrose & Norr, 1993)

 $^{\circ}\delta^{18}O_{sc}$ water values are calculated using Equations 2.2 and 2.3, see section 2.5

3.4.2 Incremental structural carbonate

The $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values for untreated samples of serially sampled modern and Pleistocene horse enamel are listed in Table 3.4. Secondary carbonate precipitation, which the pretreatment method is used to remove, is generally minimal for enamel, and the use of untreated samples makes it possible to analyze narrower increments in the direction of tooth growth. The isotopic results for the Pleistocene horses are plotted against tooth sampling position in Figure 3.3 (incremental distance starting near the occlusal surface and ending near the root). The carbon isotope variation within individual teeth ranges from +0.6 to +2.0 ‰ (excluding RAM 8T for which there was only two samples). The variation in oxygen isotopic composition within individual teeth ranges from +2.0 to +5.1 ‰ (excluding RAM 8T). Cyclicity is apparent in the oxygen isotopic compositions of RAM 2 and RAM 10. RAM 1, RAM 3, and RAM 4 display irregular fluctuations in their oxygen and carbon isotopic compositions with growth direction in the tooth. RAM 5, RAM 6, RAM 7, and RAM 9 have carbon isotope patterns that track those of the oxygen isotopic variation for most of the occlusal-root length, but at certain times during tooth growth, covariation between the $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values is not synchronized. The RAM 8 tooth is worn down substantially, indicating it was an older horse when it died (see section 1.3.3, Table 4.2); only two serial samples could be obtained.

| | Strue Carb | ctural onate | *Diet | **Water | | Strue Carb | ctural onate | *Diet | **Water |
|-----------------|-------------------|-------------------|-------------------|-------------------|-------------|-------------------|-------------------|-------------------|-------------------|
| Increment | δ ¹³ C | δ ¹⁸ Ο | δ ¹³ C | δ ¹⁸ Ω | Increment | δ ¹³ C | δ ¹⁸ Ο | δ ¹³ C | δ ¹⁸ Ω |
| (mm) | (‰) | (‰) | (‰) | (‰) | (mm) | (‰) | (‰) | (‰) | (‰) |
| Modern | | | | | Modern | | | | |
| 308(P4L) | | | | | ^306(P2L) | | | | |
| 0 | -14.4 | +19.8 | -28.5 | -16.2 | 0 | -14.5 | +17.2 | -28.6 | -19.8 |
| 4 | -14.3 | +18.4 | -28.4 | -18.1 | 0 | -15.4 | +15.9 | -29.5 | -21.7 |
| 8 | -14.6 | +16.6 | -28.7 | -20.7 | 3.5 | -14.7 | +16.5 | -28.8 | -20.8 |
| 12 | -14.4 | +18.8 | -28.5 | -17.6 | | | | | |
| 15 | -15.1 | +19.2 | -29.2 | -17.0 | | | | | |
| Mean | -14.6 | +18.6 | -28.7 | -17.9 | Mean | -14.9 | +16.5 | -29.0 | -20.8 |
| 408(P4R) | | | | | 406(P2R) | | | | |
| 0 | -14.8 | +20.0 | -28.9 | -15.9 | 0 | -14.6 | +16.9 | -28.7 | -20.3 |
| 4 | -15.2 | +17.6 | -29.3 | -19.2 | 4 | -14.8 | +17.9 | -28.9 | -18.9 |
| 8 | -15.1 | +16.3 | -29.2 | -21.1 | | | | | |
| 11 | -15.2 | +18.0 | -29.3 | -18.6 | | | | | |
| 15 | -15.4 | +17.9 | -29.5 | -18.8 | | | | | |
| Mean | -15.2 | +18.0 | -29.3 | -18.8 | Mean | -14.7 | +17.4 | -28.8 | -19.6 |
| Pleistocene | | | | | Pleistocene | | | | |
| RAM 1 | | | | | RAM 2 | | | | |
| 0 | -11.2 | +17.1 | -25.3 | -20.0 | 0 | -11.6 | +15.6 | -25.7 | -22.1 |
| 5 | -11.1 | +16.8 | -25.2 | -20.3 | 5 | -11.1 | +16.7 | -25.2 | -20.6 |
| 10 | -11.3 | +16.8 | -25.4 | -20.4 | 10 | -10.7 | +15.9 | -24.8 | -21.6 |
| 15 | -11.6 | +15.8 | -25.7 | -21.8 | 15 | -10.8 | +13.8 | -24.9 | -24.6 |
| 20 | -11.5 | +15.4 | -25.6 | -22.3 | 20 | -10.3 | +11.8 | -24.4 | -27.4 |
| 25 | -11.1 | +14.9 | -25.2 | -23.0 | 25 | -10.4 | +13.7 | -24.5 | -24.8 |
| | | | | | 30 | -10.2 | +15.7 | -24.3 | -22.0 |
| | | | | | 35 | -10.8 | +13.9 | -24.9 | -24.5 |
| Mean | -11.3 | +16.1 | -25.4 | -21.3 | Mean | -10.7 | +14.6 | -24.8 | -23.4 |
| RAM 3 | | | | | RAM 4 | | | | |
| 0 | -14.2 | +15.9 | -28.3 | -21.7 | 7 | -12.2 | +17.1 | -26.3 | -19.9 |
| 11 | -14.1 | +16.4 | -28.2 | -21.0 | 13 | -11.9 | +15.8 | -26.0 | -21.8 |
| 21 | -13.5 | +14.2 | -27.6 | -24.0 | 17 | -11.6 | +15.1 | -25.7 | -22.7 |
| 27 | -13.7 | +14.2 | -27.8 | -24.0 | | | | | |
| 32 | -13.6 | +15.9 | -27.7 | -21.7 | | | | | |
| | | | | | | | | | |
| Mean | -13.8 | +15.3 | -27.9 | -22.5 | Mean | -11.9 | +16.0 | -26.0 | -21.5 |

Table 3.4 Stable isotopic compositions of serially sampled enamel and corresponding diet and drinking water δ -values.

| RAM 5 | | | | | RAM 6 | | | | |
|-----------|-------|-------|-------|-------|---------------|-------|-----------|-------|-------|
| 0 | -11.8 | +13.8 | -25.9 | -24.5 | 1 | -11.2 | +15.6 | -25.3 | -22.0 |
| 5 | -11.6 | +15.0 | -25.7 | -22.9 | 6 | -11.7 | +13.4 | -25.8 | -25.1 |
| 10 | -11.4 | +13.7 | -25.5 | -24.7 | 11 | -11.0 | +14.5 | -25.1 | -23.6 |
| 15 | -11.9 | +13.3 | -26.0 | -25.3 | 17 | -10.8 | +14.2 | -24.9 | -24.0 |
| 20 | -11.8 | +17.7 | -25.9 | -19.2 | 22 | -11.0 | +13.8 | -25.1 | -24.6 |
| 26 | -11.2 | +15.9 | -25.3 | -21.7 | 28 | -10.9 | +14.4 | -25.0 | -23.7 |
| 31 | -12.3 | +13.3 | -26.4 | -25.2 | 33 | -10.6 | +12.4 | -24.7 | -26.5 |
| 36 | -12.0 | +14.5 | -26.1 | -23.7 | 39 | -10.7 | +13.5 | -24.8 | -24.9 |
| 38 | -13.2 | +12.6 | -27.3 | -26.3 | | | | | |
| 43 | -11.1 | +14.0 | -25.2 | -24.2 | | | | | |
| Mean | -11.8 | +14.4 | -25.9 | -23.8 | Mean | -11.0 | +14.0 | -25.1 | -24.3 |
| RAM 7 | | | | | RAM 8 | | | | |
| 0 | -11.2 | +15.7 | -25.3 | -21.9 | 0 | -14.0 | +15.0 | -28.1 | -22.9 |
| 6 | -11.2 | +14.0 | -25.3 | -24.3 | 5 | -13.8 | +17.1 | -27.9 | -20.0 |
| 10 | -12.1 | +12.5 | -26.2 | -26.9 | | | | | |
| 15 | -11.5 | +13.8 | -25.6 | -24.5 | | | | | |
| 20 | -10.7 | +12.9 | -24.8 | -25.8 | | | | | |
| 26 | -11.2 | +13.5 | -25.3 | -25.0 | | | | | |
| 29 | -10.7 | +15.0 | -24.8 | -22.8 | | | | | |
| Mean | -11.2 | +13.9 | -25.3 | -24.5 | Mean | -13.9 | +16.0 | -28.0 | -21.5 |
| RAM 9 | | | | | RAM 10 | | | | |
| 0 | 12.2 | 116 | 26.2 | 22.5 | 0 | 10.0 | +14.7 | 26.2 | 22.2 |
| 0 | -12.2 | +14.0 | -20.3 | -25.3 | 0 | -12.2 | ے 10 2 | -20.3 | -25.5 |
| 11 | -12.0 | +14.0 | -20.1 | -24.5 | 4 | -11.5 | +10.5 | -25.0 | -16.5 |
| 11 | -12.0 | +13.9 | -20.1 | -24.3 | / | -11.0 | +14.9 | -23.7 | -25.0 |
| 17 | -11.5 | +13.9 | -23.0 | -24.4 | 11 | -11.0 | +13.3 | -23.7 | -25.0 |
| 22 | -12.7 | +14.4 | -20.8 | -23.7 | 15 | -11.4 | +14.1 | -25.5 | -24.1 |
| 28 22 | -12.7 | +13.3 | -20.8 | -25.2 | 19 | -10.8 | +13./ | -24.9 | -22.0 |
| <i>33</i> | -11./ | +12.9 | -25.8 | -25.8 | | | | | |
| 39 | -11.0 | +12.6 | -25.1 | -26.2 | | 11.7 | . 15.0 | 05.6 | 22.6 |
| Mean | -12.0 | +13./ | -26.1 | -24.7 | Mean | -11.5 | +15.2 | -25.6 | -22.6 |

Values in **bold** are the average of duplicate analyses ^Sampling sequence reset to zero on cut occlusal surface because of minimal amount of buccal enamel

* $\delta^{13}C_{sc}$ diet values are calculated assuming a $\Delta^{13}C_{sc-diet}$ value of +14.1 ‰ (Cerling & Harris, 1999)

** $\delta^{18}O_{sc}$ water values are calculated using Equations 2.2 and 2.3, see section 2.5



----Carbon

Increment (mm)

----Carbon

Increment (mm)



G) RAM 7

H) RAM 8





Figure 3.3 $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values for serially sampled Pleistocene horse teeth. Bulk $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values are also plotted for each tooth. Isotopic results are plotted against tooth sampling position (incremental distance starting near the occlusal surface and ending near the root).

3.5 Collagen

The modern horse has a collagen carbon isotopic composition ($\delta^{13}C_{col}$) of -22.7 ‰ (n=1). Replicate Pleistocene $\delta^{13}C_{col}$ values have a standard deviation of \pm 0.3 ‰ (n=2; SD). Pleistocene horse $\delta^{13}C_{col}$ values average -20.7 \pm 0.28 ‰ (SD, n=9) (Table 3.3). The $\delta^{13}C_{col}$ variation is much narrower than obtained for the pretreated bulk bone $\delta^{13}C_{sc}$ (\pm 1.0 ‰; SD, n=9). Diet carbon isotopic compositions ($\delta^{13}C_{diet}$) derived from bone $\delta^{13}C_{col}$ values are also shown in Table 3.3, and determined assuming a $\Delta^{13}C_{col-diet}$ separation of +5.5 ‰. The modern horse has a $\delta^{15}N_{col}$ value of +4.7 ‰ (n=1). The $\delta^{15}N_{col}$ analyses of Pleistocene duplicates are reproducible to \pm 0.03 ‰ (n=5, SD). There are two distinct groups of bone collagen nitrogen isotopic compositions in the Pleistocene horse bone samples (Table 3.3). RAM 1, RAM 2, and RAM 4 have low $\delta^{15}N_{col}$ values, averaging +0.2 ‰, whereas the remaining samples average +6.4 ‰ (n=6). Samples with radiocarbon dates that followed the LGM belong to the grouping with low $\delta^{15}N$ values, while samples predating the LGM have high $\delta^{15}N$ values (Fig. 3.4). There is no statistical difference in $\delta^{13}C_{col}$ values between these two groupings (Independent-samples *t*-tests).



Figure 3.4 Collagen stable carbon and nitrogen isotopic compositions for Pleistocene horses. Radiocarbon ages are listed adjacent to the dated samples. Samples with high $\delta^{15}N$ values predate the LGM, while samples with low $\delta^{15}N$ values postdate the LGM.

3.6 Enamel-bone tissue spacing

3.6.1 Phosphate spacing

The phosphate $\Delta^{18}O_{\text{bone-tooth}}$ separations for all horses are listed in Table 3.5. There is no statistical difference between modern tooth and bone $\delta^{18}O_p$ values (Mann-Whitney U Test, p= >0.650). The average difference between modern bone and teeth $\delta^{18}O$ phosphate values is -0.1 ‰, which is less than the standard deviation from the mean obtained for duplicate samples. The average $\Delta^{18}O_{\text{bone-tooth}}$ separation for the Pleistocene horse samples is +1.2 ± 1.0 ‰ (SD).

3.6.2 Structural carbonate spacing

The modern and Pleistocene horse structural carbonate $\Delta^{13}C_{\text{bone-tooth}}$ and $\Delta^{18}O_{\text{bone-tooth}}$ tooth spacing are listed in Table 3.5. The average structural carbonate $\Delta^{18}O_{\text{bone-tooth}}$ and $\Delta^{13}C_{\text{bone-tooth}}$ separations for the modern horse are 0.0 ‰ and -0.4 ‰, respectively. Within analytical error, there is no difference between modern horse bone and tooth isotopic compositions. The structural carbonate $\Delta^{13}C_{\text{bone-teeth}}$ and $\Delta^{18}O_{\text{bone-teeth}}$ separations for the Pleistocene horses range from -0.1 to +2.8 ‰, and -1.0 to +3.2 ‰, respectively.

| | | Carbonate | | | | *Phos | sphate | | $\Delta_{ m bo}$ | one-tooth tissue s | pacing | |
|------------------------|---|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--------------------------|--|---|--------------------------|
| | | Во | ne | То | oth | Bone | Tooth | Carbonate Phosphate | | CO ₃ - PO ₄ Bone | CO ₃ - PO ₄ Tooth | |
| Sample | Radiocarbon dates (¹⁴ C BP) | δ ¹³ C (‰) | δ ¹⁸ Ο (‰) | δ ¹³ C (‰) | δ ¹⁸ Ο (‰) | δ ¹⁸ Ο (‰) | δ ¹⁸ Ο (‰) | Δ ¹³ C (‰) | Δ ¹⁸ Ο (‰) | Δ ¹⁸ Ο (‰) | Δ ¹⁸ Ο (‰) | Δ ¹⁸ Ο (‰) |
| Modern | | | | | | | | | | | | |
| ^a MH (bone) | | -14.9 | +20.1 | | | +11.5 | | -0.4 | +0.0 | -0.1 | +8.6 | +8.6 |
| 308(P4L) | | | | -14.6 | +19.9 | | +11.3 | | | | | |
| 408(P4R) | | | | -14.4 | +20.3 | | +11.5 | | | | | |
| 306(P2L) | | | | -14.6 | +20.0 | | +11.7 | | | | | |
| 406(P2R) | | | | -14.3 | +20.2 | | +11.8 | | | | | |
| Pleistocene | | | | | | | | | | | | |
| RAM 1 | 11 240 <u>+</u> 110 | -11.3 | +12.5 | -11.0 | +14.9 | +7.7 | +6.4 | -0.3 | -2.4 | +1.3 | +4.8 | +8.5 |
| RAM 2 | 11 260 <u>+</u> 110 | -9.2 | +14.6 | -11.6 | +15.1 | +9.0 | +8.4 | 2.4 | -0.6 | +0.6 | +5.5 | +6.7 |
| RAM 3 | | -13.6 | +4.2 | -13.3 | +15.7 | +9.3 | +8.6 | | | +0.8 | -5.2 | +7.1 |
| RAM 4 | | -11.2 | +16.0 | -12.1 | +16.8 | +10.5 | +9.1 | 0.9 | -0.8 | +1.4 | +5.5 | +7.7 |
| RAM 5 | > 38,900 | -10.5 | +16.1 | -12.4 | +14.9 | +10.0 | +8.4 | 1.9 | 1.2 | +1.7 | +6.1 | +6.6 |
| RAM 6 | | -11.5 | +13.8 | -11.7 | +15.7 | +11.3 | +7.6 | 0.2 | -1.9 | +3.8 | +2.4 | +8.1 |
| RAM 7 | | -10.1 | +17.5 | -12.9 | +14.2 | +10.5 | +9.3 | 2.8 | 3.2 | +1.2 | +6.9 | +4.9 |
| RAM 8 | >46,700 | -11.9 | +16.1 | -13.2 | +15.7 | +9.3 | +9.1 | 1.3 | 0.4 | +0.2 | +6.8 | +6.6 |
| RAM 9 | | -12.3 | +13.5 | -11.5 | +13.7 | +8.0 | +7.8 | -0.9 | -0.1 | +0.2 | +5.5 | +5.9 |
| RAM10 | | -11.3 | +15.9 | -11.1 | +15.3 | +10.1 | +9.6 | -0.3 | 0.7 | +0.5 | +5.8 | +5.7 |

Table 3.5 Structural carbonate and phosphate bone and tooth isotopic compositions (pre-treated, bulk samples) and corresponding bone-tooth isotopic tissue spacing.

^aMH= modern horse bone

*Calculated using the average $\delta^{18}O_p$ results for both methods

Chapter 4

Discussion

4.1 Possible limitations on the life history record for the modern horse

The life history of the modern horse, including its residence and migratory record, was unavailable. This can be a limitation when comparing isotopic differences between teeth (which record isotopic signals for a limited part of an animal's life) and bone (which can record adult isotopic signals because of tissue remodelling). If the modern horse had ranged widely during its life, then comparisons of isotopic compositions between bone and teeth would include isotopic signals recorded during migration in addition to physiological processes. However, there is some evidence to rule out this possible limitation, particularly for δ^{18} O values (δ^{13} C values might be less sensitive to location changes depending on the staple food of the horse, see section 4.3.1). First, the supplier of the horse sample, Atwood Pet Food Supplies Ltd., only collects animals from southwestern Ontario, and in particular from the Chatham to Kincardine area (Fig. 4.1). If the horse moved within this region, it's $\delta^{18}O_{dw}$ values would show only limited variation as the average precipitation δ^{18} O value in this area is around -9 ‰ (F.J. Longstaffe, personal communication), assuming it drank predominantly from local surface waters (Fig. 4.1 B), allowing tissue offsets to be identified. Second, it is unlikely that the horse travelled extensively given it was relatively young when it died; its 06 and 08 teeth hadn't finished mineralizing (see section 2.2), which mineralize from ~13-19 months and ~31-51 months, respectively (Hoppe et al., 2004a). Based on this evidence, the horse was probably less than 4 years old when it died. For these reasons, it is very likely that the modern horse bone and teeth were recording the same diet and drinking water isotopic signals.



Figure 4.1 A) Map of amount-weighted monthly precipitation δ^{18} O values for North America. The black box identifies the general area in which the modern horse lived. B) Monthly precipitation δ^{18} O range for southwestern Ontario, where the modern horse lived. This area has an average precipitation δ^{18} O value of -9 % (F.J. Longstaffe, personal communication). The dashed line marks the north easternmost extent of the possible range for the modern horse (modified from IAEA, 2001; The Atlas of Canada, 2012). Legend as in Figure 4.1 A.

4.2 Oxygen isotope tissue-spacing

4.2.1 Modern horse

Warinner and Tuross (2009) proposed that bone and enamel in pigs may not be isotopically equivalent tissues in terms of their bioapatite structural carbonate compositions (see section 1.2.2). This was not found to be the case for the modern horse analysed in this study; the average structural carbonate $\Delta^{18}O_{\text{bone-tooth}}$ spacing of the modern horse was ~0 ‰ (see section 3.6.2, Table 3.5). The small average structural carbonate $\Delta^{13}C_{\text{bone-tooth}}$ spacing of -0.4 % can be attributed to normally encountered variation. For example, Brookman and Ambrose (2012) sampled different lophs (ridges) of kangaroo teeth in southern Australia and found that the difference between lophs from the same tooth could vary from 0.1 to 0.7 % for $\delta^{13}C_{sc}$ and 0.1 to 0.8 % for $\delta^{18}O_{sc}$. They also found that differences between left- and right-paired molars ranged from 0.0 to 1.0 ‰ for $\delta^{13}C_{sc}$ and 0.0 to 1.7 ‰ for $\delta^{18}O_{sc}.$ Brookman and Ambrose (2012) suggested that the differences correspond to the 'inherent isotopic variability' caused by different timing of enamel mineralization or because of slight variations in sampling techniques (e.g. depth of sample interval, etc.). This example illustrates the isotopic variability that is present in a tissue like enamel. Similar isotopic variability may be present in the teeth and bone of the modern horse in this study.

There is a chance that because the horse teeth sampled here were not fully mineralized their isotopic compositions may be seasonally biased, reflecting maximum or minimum isotopic compositions (corresponding to summer highs or winter lows, respectively). However, if this was the case more of an offset would be predicted between teeth and bone isotopic compositions, the latter of which represents more of a seasonal average. Horses grow very quickly in the first years after birth. At birth, a foal has only ~17 % of their mature bone mineral content (Lawrence, 2005). They reach 90 % of their adult weight (~500 kg) by the time they are 22 months old (Thompson, 1995; Lawrence, 2005). However, maximum bone mineral content is not reached until ~6 years of age (Lawrence, 2005). Therefore, if the horse sampled here died after the 08 teeth began mineralizing (which starts at ~19 months of age), then its bones would still have been

forming. Hence, it is highly likely that both tissues were incorporating the same body water at the same time.

In these modern samples, there is no statistical difference in $\delta^{18}O_p$ values between tooth enamel and bone that formed at the same time (see section 3.6.1), implying that the phosphate component of bioapatite fractionates in the same way in both tissues in horses. Similarly, the lack of difference in structural carbonate isotopic compositions between horse bone and tooth enamel formed at the same time suggests that oxygen and carbon isotopes, respectively, fractionate very similarly in these two tissues, at least in horses. Hence any significant measured differences are not of metabolic origin, and therefore can be used to study external factors such as migration, dietary, and climatic change. The difference between the findings of this study and those of Warinner and Tuross (2009) may indicate that isotopic offsets arising from metabolic processes are species-specific (e.g. pigs vs. horses). That said, it should be noted that only one horse has been studied here, whereas Warinner and Tuross (2009) had a larger sample set (9 pigs). Certainly, more horse samples fed a controlled diet need to be analyzed to confirm the results of the one set of observations reported here.

4.2.2 Phosphate-structural carbonate

The use of bone-enamel isotopic offsets for both oxygen and carbon to detect migration and/or dietary changes is only possible if both tissues have retained their original isotopic compositions. The relationship between $\delta^{18}O_{sc}$ and $\delta^{18}O_{p}$ values in mammalian bioapatite has been used as a tool for identifying *post-mortem* alteration (see section 2.3). The use of this tool is based on the assumption that during enamel and bone mineralization, structural carbonate and phosphate components form at the same time from the same body water (Iacumin et al., 1996; Pellegrini et al., 2011). However, it has been shown previously that the $\Delta^{18}O_{sc-p}$ relationship can vary depending on the mammalian species analyzed, and that $\delta^{18}O_{sc}$ and $\delta^{18}O_{p}$ do not adhere to a strictly defined relationship with each other (e.g. Bryant et al., 1996a; Zazzo et al., 2004b; Martin et al., 2008; Pellegrini et al., 2011). Possible causes of such variation are that the phosphate and carbonate components of bioapatite do not precipitate at exactly the same time and/or from the same body water (Pellegrini et al., 2011).

Bryant et al. (1996a) and Pellegrini et al. (2011) calculated fractionation factors for equids between $\delta^{18}O_{sc}$ and $\delta^{18}O_{p}$ (α_{sc-p}) of 1.0086 ± 0.0007 (for bone and enamel) and 1.0086 ± 0.0005 (for enamel), respectively. The average α_{sc-p} for the modern horse analyzed in this study (using both bone and enamel compositions) is 1.0085 ± 0.0001, which is identical within error of the values reported by Bryant et al. (1996a) and Pellegrini et al. (2011). The average α_{sc-p} for the bone and enamel of Pleistocene horses sampled here is 1.0061 ± 0.0012, which is substantially lower than expected. The average $\Delta^{18}O_{sc-p}$ for Pleistocene horses is +6.2 ± 1.4 ‰ (SD, for both bone and teeth, n=19), which is lower than the average value for the modern horse of +8.6 ± 0.2 ‰ (SD, for bone and teeth, n=5, Table 3.5) and the +8.7 ‰ offset determined by Bryant et al. (1996a).

As mentioned earlier, $\Delta^{18}O_{sc-p}$ can vary depending on the species analysed; the extent of deviation from the established norm for a species before *post-mortem* alteration is suspected has not been established conclusively in most cases. Pellegrini et al. (2011) compiled the available α_{sc-p} values for modern animals, noting the average value to be $+1.00867 \pm 0.00077$. In other words, the average $\Delta^{18}O_{sc-p}$ offset was $+8.8 \pm 0.8 \%$ with a range of +6.6 to +10.6 %. While a better understanding of the relationship between $\delta^{18}O_{sc}$ and $\delta^{18}O_{p}$ is needed to use the relationship as a test for *post-mortem* alteration with high confidence, Pellegrini et al. (2011) suggested that $\Delta^{18}O_{sc-p}$ offsets <+7 % or >+11 % should be approached with caution as they likely indicate diagenetic alteration of one tissue or the other, or perhaps both.

Coexisting $\delta^{18}O_p$ and $\delta^{18}O_{sc}$ values for Pleistocene bone and enamel are plotted in Figure 4.2. If the recommended 'envelope' of accepted values is applied to the bulk isotopic analyses, then only RAM 7B and RAM 1T, 2T, 3T, 4T, and 6T samples lie within the 'acceptable' range for unaltered material. If the other Pleistocene bone and teeth samples have undergone *post-mortem* alteration that affected their bulk $\delta^{18}O$ (and $\delta^{13}C$) values, it is important to try to identify which component (i.e. carbonate and/or phosphate) has been affected and by what mechanism. Under inorganic conditions and at low temperatures the phosphate moiety is very resistant to isotopic exchange (e.g. Blake et al., 1997; Lecuyer et al., 1999), while the structural carbonate moiety is much more susceptible to alteration because of faster rates of exchange (Zazzo et al., 2004a).

Significant oxygen isotope exchange, however, can occur in the phosphate moiety under certain conditions because microbial enzymes are able to break inorganic P-O bonds (Blake et al., 1997; Zazzo et al., 2004a). Zazzo et al. (2004b) plotted intratooth sequential δ^{18} O values in a $\delta^{18}O_p$ - $\delta^{18}O_{sc}$ space and proposed a model to determine whether microbial mediated or inorganic diagenesis affected tooth δ^{18} O values. This model is based on the different kinetics of *post-mortem* alteration reactions involving phosphate and carbonate moieties under inorganic and microbial conditions. They suggested that a slope of phosphate versus structural carbonate δ^{18} O values <1 indicates bioapatite-water alteration dominated by microbial conditions, which mainly affects the phosphate moiety, whereas a slope >1 indicates alteration dominated by inorganic processes, which mainly affects the structural carbonate moiety. The oxygen isotope phosphate/structural carbonate slopes obtained for both the Pleistocene teeth and bone samples are <1 (Fig. 4.2), which suggests a role for microbial alteration of phosphate oxygen isotopic compositions according to the Zazzo et al. (2004b) model. In this scenario, it appears that bone was affected more than teeth as the $\Delta^{18}O_{sc-p}$ spacings of the former deviate most from the 'envelope' of accepted values described by Pellegrini et al. (2011).

All bone $\delta^{18}O_p$ values obtained for the Pleistocene samples are higher than their corresponding enamel $\delta^{18}O_p$ values (Fig. 3.1). This offset is in contrast to the lack of a significant difference between modern horse bone and teeth $\delta^{18}O_p$ (or $\delta^{18}O_{sc}$) values (see section 3.6.1). In addition, this consistent bone>tooth offset was not observed in Pleistocene $\delta^{18}O_p$ values (Fig. 3.2). The same pattern of higher bone $\delta^{18}O_p$ values relative to teeth $\delta^{18}O_p$ has been reported elsewhere for large animals (e.g. elephants, bison, horses, etc.; Ayliffe et al., 1992; Tütken et al., 2008), and was attributed to *post-mortem* isotopic exchange of $\delta^{18}O_p$ with surrounding fluids. While Ayliffe et al. (1992) did not rule out microbial activity as its cause, Tütken et al. (2008) found no evidence for a *post-mortem* microbial influence on their samples. Ayliffe et al. (1992) suggested that the observed $\delta^{18}O_p$ alteration was caused by re-equilibration with soil waters through replacement of the original bioapatite or because of isotopic exchange. Tütken et al. (2008) speculated that changes in the $\delta^{18}O_p$ values of bone could result from *post-mortem* inorganic processes on the microscopic scale, during which isotopic exchange with surrounding fluids occurs despite the presence of collagen.



Figure 4.2 $\delta^{18}O_p$ versus $\delta^{18}O_{sc}$ for individual samples of Pleistocene teeth A) and bone B). The line of best fit for modern equids is shown by the solid black line (Bryant et al., 1996a). The dashed line represents the lower limit (+7 ‰) of the acceptable 'envelope' for unaltered samples, according to Pellegrini et al. (2011). The regression line (and its equation) for the isotopic data for the Pleistocene horses are shown in blue.

For the Pleistocene horses studied here, both the bone and teeth $\delta^{18}O_n$ values suggest some degree of microbially mediated *post-mortem* alteration (if the Zazzo et al. 2004b model is accepted). The $\delta^{18}O_p$ values of bone are more severely affected than those of teeth, which might explain the observed offset in $\delta^{18}O_p$ values between these two tissues (Table 3.2 and Fig. 3.1). The Pleistocene horse samples were obtained from gravel deposits that were frequently saturated by meteoric water. Modern Edmonton has an average meteoric water δ^{18} O value of -17.1 % (Fig. 4.3) (IAEA/WMO, 2010). Some of the Pleistocene horse bones could have undergone microbially mediated isotopic exchange with such meteoric water, causing a shift in δ^{18} O values towards such compositions. If this was the case, it would explain why a bone-enamel oxygen isotope offset wasn't observed in structural carbonate compositions for these samples; structural carbonate has been shown to be more resistant to microbial alteration (Zazzo et al., 2004a). This does not rule out the possibility that the structural carbonate component also experienced *post-mortem* alteration under inorganic conditions. For example, based on the calcite-water oxygen isotopic geothermometer of Friedman and O'Neil (1977), modern secondary calcite formed at 5 ° and 10 °C from Edmonton meteoric water (avg. -17.1 ‰) has δ^{18} O values (+16.0 and +14.7 ‰, respectively). Such compositions lie within the range of $\delta^{18}O_{sc}$ values measured for the Pleistocene teeth and bone (Fig. 3.2). Therefore, it is not impossible that some of the Pleistocene horse bones and teeth underwent isotopic exchange with meteoric water (under inorganic conditions), causing a shift in $\delta^{18}O_{sc}$ values towards meteoric compositions. However, it is very likely that the teeth samples would have been less affected by such alteration because enamel is more resistant to alteration because of its smaller crystallite surface area and lower porosity (Zazzo et al., 2004a). Also, it seems that not all Pleistocene samples were affected equally by this putative *post-mortem* alteration; for example, the $\delta^{18}O_p$ value of RAM 7B, and the $\delta^{18}O_{sc}$ and $\delta^{18}O_{p}$ values of RAM 1T, 2T, 3T, 4T, and 6T likely reflect their original composition, given their $\Delta^{18}O_{sc-p}$ spacings (Table 3.5). Preservation of original isotopic compositions is particularly probable for RAM 2T, given its sinusoidal $\delta^{18}O_{sc}$ curve that follows expected seasonal patterns of growth (see section 4.4.3).



Figure 4.3 Calculated $\delta^{18}O_{dw}$ for Pleistocene and modern bone and teeth based on $\delta^{18}O_p$ values (Equation 2.2). The average annual meteoric water $\delta^{18}O$ value for Edmonton is represented by the black line (IAEA/WMO, 2010).

The possibility also exists that the pattern of phosphate $\Delta^{18}O_{bone-tooth}$ spacings observed for the Pleistocene samples reflects the age of individual horses. Some teeth had been worn down substantially; hence the isotopic compositions of the remaining bulk material could be seasonally biased. If the teeth reflect minimum $\delta^{18}O_{dw}$ values, representing the winter season, this would be reflected in lower teeth $\delta^{18}O_p$ values, and comparatively higher $\delta^{18}O_p$ values in the associated bone (and *vice versa*). Table 4.1 lists the Pleistocene horses in descending order from oldest to youngest and their corresponding phosphate $\Delta^{18}O_{bone-tooth}$ spacing. The ordering of horses from oldest to youngest is based on the height of the teeth, and the fact that young horses have relatively unworn teeth. For example, before wear, the 06s (the shortest cheek teeth) can be ~5 cm long, while the rest of the cheek teeth (e.g. 08s) can be up to ~7–8 cm long in young Thoroughbreds. Cheek teeth wear at a rate of ~2–3 mm/yr, and therefore – the shorter the tooth, the older the horse (Dixon, 2002). All of the Pleistocene teeth analysed in this study were 06s except RAM 4T, which was an 08 (Table 2.1). Therefore RAM 4T is considered to be the oldest given that it was likely a taller tooth at the outset and because its tooth is most worn. The results of this examination, however, found no statistical relationships between the age of the horse ('youngest' five and 'oldest' five) and phosphate $\Delta^{18}O_{\text{bone-tooth}}$ spacing (Independent-samples *t*-test, p= 0.294) or structural carbonate $\Delta^{18}O_{\text{bone-tooth}}$ spacing (Independent-samples *t*-test, p= 0.339).

| | Tooth | PO ₄ | CO ₃ | |
|--------|---------|-----------------|-----------------|----------|
| | height, | bone- | bone- | |
| | distal | enamel | enamel | |
| | end | $\Delta^{10}O$ | $\Delta^{10}O$ | |
| Sample | (mm) | (‰) | (‰) | |
| RAM 4 | 35.5 | +1.4 | -0.8 | Oldest |
| RAM 8 | 34.5 | +0.2 | +0.4 | |
| RAM 1 | 51.5 | +1.3 | -2.4 | |
| RAM 10 | 53.5 | +0.5 | +0.7 | |
| RAM 2 | 58.0 | +0.6 | -0.6 | |
| RAM 3 | 60.5 | +0.8 | | |
| RAM 7 | 61.5 | +1.2 | +3.2 | |
| RAM 5 | 67.5 | +1.7 | +1.2 | |
| RAM 9 | 73.0 | +0.2 | -0.1 | |
| RAM 6 | 75.5 | +3.8 | -1.9 | Youngest |

Table 4.1 The height of Pleistocene horse teeth (distal end, mm) and corresponding phosphate and structural carbonate $\Delta^{18}O_{\text{bone-tooth}}$ spacing, listed from oldest to youngest horse.

If the Pleistocene horses (particularly the bone) underwent *post-mortem* isotopic alteration, why was this not detected by the other diagenetic tests, for example C/N ratio and CI index? While all but one of the Pleistocene bone samples have C/N ratios within the acceptable range, there was still significant *post-mortem* loss of collagen. Collagen can retain its amino acid composition, its C/N ratio, and $\delta^{13}C_{col}$ and $\delta^{15}N_{col}$ values until 99 % of the collagen by weight has been lost (e.g. Dobberstein et al., 2009; Tütken & Vennemann, 2011). Collagen loss increases bone porosity and the exposure of bone bioapatite, which facilitates its dissolution and/or recrystallization – typically into larger crystallites (Tütken & Vennemann, 2011 and references therein). That said, a significant increase in bioapatite crystallite size was not detected in the Pleistocene samples; all bone and enamel CI values remained within the accepted range (see section 3.1). Tütken et al. (2008) also report alteration of bone $\delta^{18}O_p$ values without an associated increase in crystallinity, which they could not explain. Tütken and Vennemann (2011) noted that the best-preserved bone typically has a collagen yield of ~14 % or higher, similar to that of modern bone. All but one of the Pleistocene bone samples have collagen yields lower than 14 % (Table 3.1), with an average yield of 6.9 % (excluding RAM 4B which has a yield of 15 %). Breakdown of collagen by microbial attack and/or hydrolysis are common and important processes in bone degradation (Child, 1995; Collins et al., 2002; Tütken & Vennemann, 2011). Therefore, while original Pleistocene bone collagen isotopic signals may have been preserved, the low amount of collagen remaining in these samples may have increased microbial access to the bioapatite crystallites, thus facilitating *post-mortem* alteration via this pathway.

4.3 Bulk and serially sampled structural carbonate

4.3.1 Modern horse structural carbonate drinking water and diet isotopic signatures

The $\delta^{18}O_{dw}$ value calculated for the modern horse from its bulk $\delta^{18}O_{sc}$ value is -15.7 ‰ for both bone and teeth (similar to the average $\delta^{18}O_{dw}$ value of -15.5 ‰ calculated using $\delta^{18}O_p$). This is lower than the average annual precipitation $\delta^{18}O$ value known for the region where the horse most likely lived: -9 ‰ (F.J. Longstaffe, personal communication, Fig. 4.1). This area has a seasonal range in meteoric water δ^{18} O values of -14.2 to -7.0 % (F.J. Longstaffe, personal communication). The serially sampled $\delta^{18}O_{dw}$ values, by comparison, display a smaller range: 308 (P4L), -20.7 to -16.4 ‰, avg. -17.9 ‰; 408 (P4R), −21.1 to − 15.9 ‰, avg. −18.8 ‰ (06 results are not included because of the limited amount of enamel present, Table 3.4). The smaller range and lower bulk average $\delta^{18}O_{dw}$ values could arise in several ways. First, it is very likely that the horse did not drink directly from precipitation, but rather drank from well water. Shallow groundwater in southwestern Ontario typically has δ^{18} O values that are ~2 % lower than average precipitation, because of preferential recharge during winter months (F.J. Longstaffe, personal communication). Average groundwater δ^{18} O values range from around -11.5 % near the southern shores of Lake Huron to around -12.5 % in the vicinity of Barrie, Ontario, on the shores of Lake Simcoe (North et al., 2013). Second,

some groundwater in the region has still lower δ^{18} O values because it still contains some components of Late Pleistocene glacial meltwater (Husain et al., 2004). Third, there are errors associated with the two-step calculation of $\delta^{18}O_{dw}$ values from $\delta^{18}O_{sc}$ values, as discussed in section 4.4.2. Without further details on the exact drinking water composition of the modern horse analyzed here, it is not possible to speculate further.

The modern horse diet compositions, as calculated from their pretreated bulk $\delta^{13}C_{sc}$ values, are -29.0 and - 28.6 ‰ for bone and teeth, respectively (serially sampled diet $\delta^{13}C_{sc}$ values are similar; Table 3.3, 3.4). These compositions are indicative of an entirely C₃ diet (see section 1.2.1). Modern domesticated horses in Ontario mainly eat hay, pasture, grains, or concentrates (OMFRA, 2012). Pieces of hay were found imbedded between the teeth and jawbone of the modern horse. Sponheimer et al. (2003) reported $\delta^{13}C$ values of modern alfalfa hay feed at -27.0 ± 0.4 ‰, similar to what is suggested here. It is likely that the modern horse's diet largely consisted of hay, with little to no supplement from C4 plants, such as corn. The diet composition determined from the $\delta^{13}C_{col}$ values (-28.2 ‰) for this horse lead to the same conclusion (Table 3.3).

4.3.2 Differences between bulk and serially sampled structural carbonate

The majority of the average $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values obtained for the serially sampled teeth are consistently lower than their corresponding bulk isotopic compositions (Table 3.3, 3.4). Serially sampled average $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values are on average 0.6 ‰ and 1.4 ‰ (n=14) lower than bulk values, respectively (for both modern and Pleistocene horses). If this difference was caused by a seasonal bias in the material sampled, the compositions obtained for serially sampled materials wouldn't be expected to be consistently lower than bulk compositions. This is also unlikely to be a species-specific effect. The same pattern of lower average incremental $\delta^{18}O_{sc}$ values has been observed for mammoths and mastodons (Metcalfe, 2011). Metcalfe (2011) attributed the difference to faster mineralization of the innermost enamel (used for serially sampling) compared to the rest of the enamel (used for bulk samples) and associated compositional differences in the fluid from which the enamel precipitates.

4.3.3 Pleistocene bulk structural carbonate drinking water and diet isotopic signatures

The average bone and tooth $\delta^{18}O_{dw}$ values (± SD) for the Pleistocene horse samples were $-22.7 \pm 2.2 \ \text{$\%$}$ and $-22.6 \pm 1.2 \ \text{$\%$}$, respectively (Table 3.3). The similarity between bone and tooth $\delta^{18}O_{dw}$ values suggests that the horses drank from comparable drinking water sources during bone and tooth mineralization.

To interpret the carbon isotopic data for the Pleistocene horses, we must first consider the Suess effect. The δ^{13} C value of atmospheric CO₂ has been decreasing over the past 200 years as a result of the increase in CO₂ added to the atmosphere by fossil fuel burning and deforestation (Verburg, 2007). This decrease in atmospheric CO₂ δ^{13} C values from –6.5 ‰ (around 1744 AD) to –8.0 ‰ (modern day) is known as the Suess effect (Friedli et al., 1986; Dotsika et al., 2011 and references therein). Modern C₃ plant isotopic compositions must therefore be adjusted upwards by 1.5 ‰ for comparison with Pleistocene bone or tooth carbon isotopic compositions (Friedli et al., 1986; Marino & McElroy, 1991).

The average bone and tooth diet δ^{13} C values (± SD) obtained for the Pleistocene horse samples were $-25.2 \pm 1.0 \%$ and $-26.2 \pm 0.9 \%$, respectively (excluding RAM 3B; Table 3.3). This suggests that the Pleistocene horses had a diet consisting almost entirely of C₃ plants (C₃ plants have an average value of -25.0 %; after correcting for the Suess effect). This was expected as Pleistocene Alberta was overwhelmingly dominated by C₃ grasses and Pleistocene horses consumed predominantly grasses (see section 1.4). The narrow range among bulk bone and teeth $\delta^{13}C_{sc}$ and $\delta^{18}O_{sc}$ values, respectively, suggests that the Pleistocene horses sampled here consumed similar food and water during enamel (which mineralizes early on in an animal's life) and bone (which remodels itself throughout an animal's life) formation.

4.3.4 Growth rate estimates for serially sampled Pleistocene horse teeth.

The collective amount of time (in months) represented by the serial samples of the Pleistocene horse teeth (Table 4.2) has been estimated as:

Total growth time (in months) = $(DL*n_{DL} + S*n_S)/(GR)$, [Equation 4.1] where DL = drill line length (in μ m), usually ~2500 μ m parallel to tooth growth direction; n_{DL} = number of growth lines; S = space (in μ m) between drill lines parallel to tooth growth direction; n_S = number of spaces, and GR = growth rate in μ m/month (see section 2.2).

This provides an estimate (in months) of the time represented by serially sampled enamel collected along the length of the tooth. Important caveats are that this calculation assumes a constant growth rate. Enamel growth rates are not completely constant and are known to vary among modern equid species (and likely ancient equids, e.g. Hoppe et al., 2004a).

Table 4.2Growth time estimates (in months) for serially sampled Pleistocene horseteeth.

| | RAM | RAM | RAM | RAM | RAM | RAM | RAM | RAM | RAM | RAM |
|------------------|------|-----|------|-----|------|------|------|-----|------|------|
| Sample | 1 | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 | 10 |
| Time represented | | | | | | | | | | |
| (months) | 13.2 | 15 | 14.1 | 21 | 16.6 | 13.5 | 12.8 | 3.8 | 13.6 | 16.7 |

4.4 Palaeoecological implications

4.4.1 Late Pleistocene climate

As mentioned in sections 3.3 and 3.5, there is an association between $\delta^{15}N_{col}$ values and pre- versus post-LGM radiocarbon dates. Horses with pre-LGM dates have significantly higher $\delta^{15}N$ values (averaging +6.0 ‰, n=2) than post-LGM horses (averaging +0.1 ‰, n=2) (Fig. 4.4). Low post-LGM $\delta^{15}N_{col}$ values have also been reported elsewhere. For example, reindeer from Siberia have a range of $\delta^{15}N_{col}$ values from +1.3 to +1.5 ‰ (Iacumin et al., 2000), horses and deer from northwest Europe have $\delta^{15}N_{col}$ values ranging between +0 to ~+1.2 ‰ (Richards & Hedges, 2003), and horses from northwest Europe have $\delta^{15}N_{col}$ values as low as -0.9 to +2.2 ‰ (Stevens & Hedges, 2004). This relationship of low $\delta^{15}N_{col}$ values in post-LGM time can be used to tentatively classify the other Pleistocene horse samples as either pre- (RAM 6, 7, 9, 10) or post-LGM (RAM 4) (RAM 3 did not contain extractable collagen).
In light of the different horse $\delta^{15}N_{col}$ values between pre- and post-LGM time periods, it is important to review the North American (and Eurasian) environments in which the horses (and other megafauna) lived during these times. During pre-LGM conditions in the Mid-Wisconsinan (concurrent with Marine Isotope Stage 3, MIS 3, ~70–30 ka BP), the ice sheets were smaller compared to ice volumes before MIS 3 and during the LGM (Anderson & Lozhkin, 2001 and references therein; Miller et al., 2010). Northern areas (e.g. Alaska, Beringia, and Yukon Territory) supported open herb or steppe tundra as well as shrubland, and conifers were present in interior regions (Anderson & Lozhkin, 2001; MacDonald et al., 2012). Grasses and willows were abundant during the Mid-Wisconsinan, but birch was less common than during the Pleistocene-Holocene transition (Anderson & Lozhkin, 2001). In Alaska, which was cool and dry during the Mid-Wisconsinan, megafauna were dominated by horse, bison, and mammoths (Guthrie, 2001). These three megafauna were also dominant in Alberta prior to the LGM (Burns, 1996; Jass et al., 2011). This observation suggests that this period in Alberta was also cool and dry (but warmer than during the LGM) with open grasslands; for example, 40 % of mammals found in an Early-Wisconsinan Alberta floodplain deposit are known to have occupied open grassland habitats (Anderson & Lozhkin, 2001; Harington, 1990 and references therein; Dredge & Thorleifson, 1987).



Figure 4.4 Bone collagen carbon and nitrogen isotopic compositions for Pleistocene horses. Radiocarbon ages are listed adjacent to the dated samples. LGM = Last Glacial Maximum.

The LGM was particularly cold globally and arid in the north (Miller et al., 2010; Guthrie, 2001). Most of southern Alberta did not experience a full glaciation during the Pleistocene until this time (Young et al., 1994; Burns, 1996; Burns, 2010). An estimate for the appearance of ice cover in the Edmonton area is ~21.3 14 C ka BP (~25.5 ka BP), based on radiocarbon dated fossils (Young et al., 1994). At the LGM, forests were absent north of 55° N (including Edmonton), there was a mixture of steppe and tundra in the northern Eurasian interior, and a mixture of graminoid and forb in Beringia (Bigelow et al., 2003).

After the LGM, the ice sheets retreated as temperatures rose in North America. Deglaciation occurred slowly between 18-13 ¹⁴C ka BP (~21.4–15.5 ka BP) (Dyke & Prest, 1987; Dyke et al., 2003). There is some debate as to when central Alberta became ice-free (see section 1.1). Dyke and Prest (1987) and Dyke et al. (2003) (based on radiocarbon chronologies) portrayed it as ice-free between 12-11 ¹⁴C ka BP (~13.8–12.8, conversion from INCAL, 2009) (Fig. 4.5), while Munyikwa et al. (2011) suggested it was ice-free one thousand years earlier (using optically stimulated luminescence dating).

After the ice retreat, northern populations of boreal taxa such as *Picea*, *Pinus*, and *Betula* expanded into areas where ice had been, while southern taxa declined (based on fossil pollen records, Williams et al., 2004). From ~15–10 ka BP, there was development of large peatlands, dense *Betula* cover, and conifer forest (e.g. *Pinus*, *Picea*, etc.) in Beringia (including Alaska and Yukon territory) (MacDonald et al., 2012 and references therein). Pollen records from 12–10¹⁴C ka BP (~13.8–11.4 ka BP) indicate that central Alberta was dominated by herbs, grasses, and deciduous shrubs, with few trees (MacDonald & McLeod, 1996). Picea and Pinus also increased in abundance in central Canada, connecting with western species that were expanding north into the Canadian Rockies and Picea populations from Alaska (Williams et al., 2004 and references therein). After the onset of the Holocene at $\sim 10^{14}$ C ka BP ($\sim 11.6-11.4$ ka BP), dense spruce forests were established across Alberta (MacDonald & McLeod, 1996; MacDonald et al., 2012). There was also an increase in the abundance of temperate trees moving north from the southeastern United States (Williams et al., 2004). The increase in temperature and moisture availability was quite notable in Alaska with the formation of lakes, bogs, shrub tundra, forests, and soil paludification (Guthrie, 2003; Mann et al., 2002 and references therein). In summary, the environment in Alberta changed from a relatively cool and dry climate through the Mid-Wisconsinan, to a drastically colder and more arid climate during the LGM. It then became warmer and moister after the LGM than it had been during the Mid-Wisconsinan.



Figure 4.5 Map and timing of the ice-sheet retreat near Alberta after the LGM, following Dyke et al. (2003) and as modified from Munyikwa et al. (2011).

4.4.2 Bulk patterns in $\delta^{l8}O_{sc}$ and $\delta^{l3}C_{sc}$ values

Enamel is assumed to be more resistant to *post-mortem* alteration than bone, and therefore of most value for palaeoenvironmental reconstruction. Accordingly, only the isotopic results for enamel are considered in this section. For the pre- and post-LGM horses the $\delta^{18}O_{dw}$ values (calculated from enamel $\delta^{18}O_p$ values) are similar, avg. –19.7 ‰ (n=6) and –20.6 ‰ (n=3), respectively, as are pre- and post-LGM $\delta^{18}O_{dw}$ values (avg. –23.0 ‰, n=6, and –22.1 ‰, n=3, respectively) calculated using bulk (pretreated) $\delta^{18}O_{sc}$ values. The relatively small discrepancy between these two sets of results could arise in a number of ways. First, and most likely, either or both of the $\delta^{18}O_{sc}$ and $\delta^{18}O_p$ values were affected to some extent by *post-mortem* alteration. This is likely the case as there was no significant difference between the $\delta^{18}O_{dw}$ values calculated using structural carbonate versus phosphate oxygen for the modern horse (see section 4.3.1). Another factor is that two equations were required (Equations 2.2 and 2.3) to calculate $\delta^{18}O_{dw}$ from $\delta^{18}O_{sc}$ values, whereas only one equation was needed to convert $\delta^{18}O_p$ values (Equation 2.2).

Third, both of these equations were determined for different equid species from different time periods (see Bryant et al., 1996a and Huertas et al., 1995).

The similarity between $\delta^{18}O_{dw}$ values for pre- and post-LGM times suggest that the horses drank from comparable water sources and/or seasonal climatic patterns were similar during both time periods. Average pre-LGM $\delta^{18}O_{dw}$ values calculated using $\delta^{18}O_{sc}$ for mammoths from Alberta are higher, –19.1 ‰ (n=4) (Metcalfe, 2011). However, the pre-LGM mammoth and horse $\delta^{18}O_{dw}$ values are not directly comparable for several reasons: (1) the mammoths were sampled from different localities in Alberta, not only the Edmonton region; (2) there may be $\delta^{18}O$ variations arising from species-specific metabolic effects, given that mammoth physiology, and diet and drinking behaviour was likely different from horses, and (3) the mammoths and horses are likely not temporally comparable; some of the mammoths had ages beyond the limits of radiocarbon dating, and hence likely represent much earlier pre-LGM time than some of the horses sampled here. That said, some of the horses labelled as pre-LGM were not radiocarbon dated and therefore could also be well beyond the radiocarbon dating limit. Therefore, a potentially large span of time could be represented by the pre-LGM horses, and hence only general environmental reconstructions can be made for such a potentially long interval.

The pre-LGM horses have slightly lower $\delta^{13}C_{diet}$ values (calculated from bulk pretreated $\delta^{13}C_{sc}$ values) compared to post-LGM horses: -26.2 ‰ (n=6) and -25.7 ‰ (n=3), respectively. As mentioned in section 1.2.1, $\delta^{13}C_{sc}$ values represent bulk carbon from the whole diet. The pre-and post-LGM structural carbonate $\delta^{13}C_{diet}$ values suggest that the horses were consuming relatively similar plants, with little change in horse diet between these time periods. The average $\delta^{13}C_{diet}$ values (calculated from bulk $\delta^{13}C_{col}$ values) for the pre- and post-LGM horses are identical: -26.2 ‰ (n=6) and -26.2 ‰ (n=3), respectively. The $\delta^{13}C_{col}$ values are mainly influenced by dietary protein, and because horses are herbivores, their protein input would have been from plant material. The similar pre- and post-LGM collagen $\delta^{13}C_{diet}$ values indicate that the horses were eating similar plant types with similar protein components. It is important to mention the climatic factors that can affect the carbon isotope composition of plants (and therefore structural carbonate and collagen), which include light, nutrient and water availability, temperature, CO₂ atmospheric concentrations, etc. (as reviewed in Heaton, 1999). These environmental parameters can influence stomatal conductance and carboxylation rates of plants, therefore affecting the amount of isotopic fractionation during photosynthesis (Stevens & Hedges, 2004; as reviewed in Heaton, 1999). The similar structural carbonate (bulk diet) and identical pre- and post-LGM collagen (protein source) $\delta^{13}C_{diet}$ values suggests that the climatic differences between these two time periods were too small to cause large changes in plant and corresponding bulk diet and protein $\delta^{13}C$ values.

The pre-LGM horse structural carbonate $\delta^{13}C_{diet}$ values are slightly lower than dated pre-LGM (~40 ¹⁴C ka BP, 44.1 ka BP) mammoth structural carbonate $\delta^{13}C_{diet}$ values from Alberta (-24.6 ‰, n=3) (Metcalfe, 2011 and references therein). Overall similarities and only minor differences in diet selection (both animals are grazers) likely explain this small variation (Metcalfe, 2011 and references therein). The horse collagen $\delta^{13}C_{diet}$ values are similar to pre- (~40 ¹⁴C ka BP, 44.1 ka BP) and post- (~11 ¹⁴C ka BP, 12.9–12.8 ka BP) LGM collagen $\delta^{13}C_{diet}$ values of proboscideans from Alberta (-25.5 ‰, n=4; -25.4 ‰, n=1, respectively; Metcalfe, 2011 and references therein). Iacumin et al. (2000) also reported little variation in $\delta^{13}C_{col}$ variation between Late Pleistocene mammoths and reindeer from Eurasia (i.e. Russia and Siberia), and suggest that pre- and post-LGM environmental conditions were similar, but more arid before the LGM. Based on the collagen isotopic compositions of mammoths, Szpak et al. (2010) likewise suggested that Siberia was cold and arid before the LGM, on average more so than central Alaska and Yukon (eastern Beringia) at this time, with eastern Beringia becoming colder and more arid after the onset of the LGM.

4.4.3 Serial sampling patterns in $\delta^{I8}O_{sc}$ and $\delta^{I3}C_{sc}$

Home ranges are undefended areas in which a horse restricts its activities and seeks shelter, food, and mates (McCort, 1984; Linklater et al., 2000). The extent of a home range can vary seasonally, depending on water and food availability (McCort, 1984; Smuts, 1975). Larger home ranges are typically associated with poorer habitat quality, and the size of a home range usually increases during the winter (Smuts, 1975; Linklater et al., 2000). The size of a horse's home range can influence the seasonal isotopic signal recorded in enamel. The average maximum variation in pre-LGM serially sampled $\delta^{18}O_{sc}$ values is 3.7 ‰ (n=5) (3.4 ‰ including RAM 8T), with the maximum

variation ranging from 2.0 to 5.1 ‰. The average maximum variation in post-LGM $\delta^{18}O_{sc}$ values is 3.0 ‰ (range, 2.2 to 4.9 ‰; n=3). Variability in the $\delta^{13}C_{sc}$ values of serially sampled pre-LGM horses ranges from 0.2 to 2.0 ‰ (n=5) and in post-LGM horses, from 0.6 to 1.5 ‰ (n=3). This intra tooth variation in $\delta^{13}C_{sc}$ lies within the range of annual variation typical for a C₃ environment (Heaton, 1999; Sharp & Cerling, 1998 and references therein). The general similarity of these oxygen and carbon isotopic ranges between time periods suggest that seasonal environmental conditions varied in more or less the same fashion during both time periods.

The $\delta^{18}O_{sc}$ pattern of the pre-LGM RAM 5T contains an abrupt increase, suggestive of a change in water source to one enriched in ¹⁸O, perhaps because of evaporative enrichment or domination by summer precipitation. This sample's $\delta^{18}O_{sc}$ pattern of variation is tracked moderately well by its variation in $\delta^{13}C_{sc}$ of 2.0 ‰. The somewhat larger variability of RAM 5T $\delta^{13}C_{sc}$ values (compared to other pre-LGM horses, Table 3.4) may indicate that it occupied a larger home range or consumed a more varied diet. However, such a range in enamel $\delta^{13}C_{sc}$ values may also reflect annual variations in C₃ plant δ^{13} C values. For example, variations of ±1.5 ‰ can be produced over the growing season because of changes in weather conditions and/or changes in plant physiology (Heaton, 1999; Sharp & Cerling, 1998 and references therein).

Pre-LGM serial samples from RAM 6T, 7T, and 9T have $\delta^{18}O_{sc}$ patterns similar to RAM 5T, in that they are irregular and exhibit abrupt changes. This may reflect shortterm seasonal climate changes (e.g. Brookman & Ambrose, 2012), consumption of different drinking water sources at different times because of a larger home range (and poorer habitat quality), and/or consumption of drinking water that did not mirror the oxygen isotopic compositions of local precipitation. RAM 6T, 7T, and 9T exhibit ranges in carbon isotopic composition that were unaccompanied by recognizable patterns in $\delta^{18}O_{sc}$. The $\delta^{13}C_{sc}$ values of these horses fall within the range of annual variations in C₃ plant $\delta^{13}C$ values. Only two datapoints are available for pre-LGM RAM 8T, with a range of 2.1 ‰ for $\delta^{18}O_{sc}$ and 0.2 ‰ for $\delta^{13}C_{sc}$.

Post-LGM sample RAM 2T is the only horse to display a well-developed sinusoidal variation in its $\delta^{18}O_{sc}$ and hence its inferred $\delta^{18}O_{dw}$ values (Table 3.4, Fig. 3.3). Such patterns are typical of mid- to high-latitude precipitation. High $\delta^{18}O_{dw}$ values

correspond to summer precipitation and low values correspond to winter precipitation (Rozanski et al., 1993). The seasonal differences at mid- to high-latitudes are caused by several factors, including seasonally changing temperature, humidity, and air masses (Rozanski et al., 1993). If this horse's drinking water was mainly controlled by precipitation input, then this should be reflected in the enamel's oxygen isotopic composition, which appears to be the case. RAM 2T's $\delta^{18}O_{sc}$ curve spans a period of ~15 months (Table 4.2), with two summer highs and one winter low. In addition, RAM 2T has a $\delta^{13}C_{sc}$ range of 1.5 ‰, which is within the range of annual variations in C₃ plant $\delta^{13}C$ values. For these reasons, RAM 2 likely had a small home range with a water source dominated by precipitation input. Had RAM 2 occupied a large home range, it would have imbibed a variety of water sources with isotopic compositions not necessarily dominated by seasonal signals.

Pre-LGM RAM 10T has a somewhat sinusoidal pattern of $\delta^{18}O_{sc}$ values, with one possible summer high and a winter low. These data are interpreted to indicate that this horse drank from a more 'constant' water source dominated by precipitation input during enamel mineralization, as was the case for RAM 2T. This pattern of $\delta^{18}O_{sc}$ values, combined with the relatively low range of measured $\delta^{13}C_{sc}$ values (1.4 ‰; similar to RAM 2T), suggests that RAM 10 had a relatively small home range.

The pattern of $\delta^{18}O_{sc}$ variation in post-LGM sample RAM 1T shows a weak inverse relationship with its $\delta^{13}C_{sc}$ values. This horse likely drank from a water source (i.e. river, lake, etc.), which was not highly responsive to seasonal changes in isotopic composition (Metcalfe, 2011 and references therein). Post-LGM RAM 4T has an irregular $\delta^{18}O_{sc}$ pattern similar to RAM 5T. RAM 1T and 4T have a low range of $\delta^{13}C_{sc}$ values (0.6 ‰ for both), suggesting that these individuals occupied a relatively small home range or consumed a restricted diet during enamel formation. Low $\delta^{13}C_{sc}$ enamel variability was also observed in Late Pleistocene horses from California by Feranec et al. (2009), who concluded that these horses had a more restricted range in habitat or plant choice compared to bison from the same general area. The extent of variation in $\delta^{13}C_{sc}$ values, however, doesn't necessarily correlate to home range size. As discussed earlier, RAM 2T (which had a sinusoidal $\delta^{18}O_{sc}$ pattern, interpreted to indicate a 'constant' water source) displays a larger range in $\delta^{13}C_{sc}$ values compared to RAM 1T and 4T but likely also occupied a small home range (see above). Furthermore, the serially sampled $\delta^{13}C_{sc}$ values for all horses fall within the range of annual variations expected for a C₃ environment, and the similarity in $\delta^{13}C_{sc}$ and $\delta^{13}C_{col}$ values before and after the LGM suggest that the horses were eating similar plants that grew under very similar conditions. RAM 3T (which does not have an associated date) has an irregular $\delta^{18}O_{sc}$ pattern, like RAM 5T, and exhibits a low range in $\delta^{13}C_{sc}$ values (0.7 ‰) similar to RAM 1T and RAM 4T.

Modern Edmonton has a seasonal range in meteoric water δ^{18} O values of -28.0 to -12.6 % with an average value of -17.1 ± 0.9 % (IAEA/WMO, 2010). The calculated serially sampled $\delta^{18}O_{dw}$ values, by comparison, display a much smaller range (see above). The smaller ranges and/or poor sinusoidal $\delta^{18}O_{dw}$ patterns determined for these Pleistocene horses (and for serially sampled teeth more generally) could have several origins. One possibility is consumption of water from lakes or groundwater springs, which average precipitation inputs and outputs over the residence time of these reservoirs. Another is that the horses drank from multiple water sources with different oxygen isotopic compositions. Perhaps of most importance is the averaging of isotopic compositions that occurs during enamel mineralization. The sampling approach is critical in this regard (Metcalfe et al., 2011; Metcalfe & Longstaffe, 2012). While innermost enamel was sampled here in an attempt to limit damping of the isotopic signal, this approach required long drill lines along the length of the enamel in order to obtain enough material for analysis, resulting in fewer serial samples overall. Finally, the absence of higher $\delta^{18}O_{dw}$ seasonal values closer to those present in the modern precipitation record could indicate overall cooler climatic conditions (compared to modern conditions) for the Pleistocene horses, with less pronounced seasonality. This would yield lower average precipitation δ^{18} O values and less variation among them over the seasonal cycle compared to the modern Edmonton region.

4.4.4 Structural carbonate-collagen carbon isotope relationship

The nature of the $\Delta^{13}C_{sc-col}$ relationship in bone is complex (Loftus & Sealy, 2012). Lee-Thorp et al. (1989) analyzed the carbon isotopic composition of bone collagen and structural carbonate from southern African fauna. They interpreted the patterns in

bone $\Delta^{13}C_{sc-col}$ spacings to be related to trophic level differences, with herbivores having a value of +6.8 ± 1.5 ‰. According to Ambrose et al. (1997), $\Delta^{13}C_{sc-col}$ spacings >+4.4 ‰ indicate that the δ^{13} C value of dietary protein is more negative than whole diet, whereas spacings <+4.4 % indicate that dietary protein is less negative than whole diet. It has also been shown (e.g. Tieszen & Fagre, 1993; Clementz et al., 2009) that $\Delta^{13}C_{sc-col}$ spacings are useful for understanding diet quality (i.e. lipid and protein content). In addition, Clementz et al. (2009) found that herbivore $\Delta^{13}C_{sc-col}$ spacings in bone were consistent spatially and temporally, having a value of $+7.6 \pm 0.5$ %. The $\Delta^{13}C_{sc-col}$ spacing of the modern horse analyzed here (+7.8 %) is consistent with this observation (Table 3.3). The average bone $\Delta^{13}C_{sc-col}$ spacing for the Pleistocene horses is +9.6 ±1.0 % (SD) (Table 3.3), similar to that obtained for mammoths from Alberta ($+9.6 \pm 0.2 \%$, n=7) (Metcalfe, 2011). Pre- and post-LGM horses have average $\Delta^{13}C_{sc-col}$ values of +9.4 and +10.1 ‰, respectively. The post-LGM horses have a higher $\delta^{13}C_{sc}$ value (the $\delta^{13}C_{col}$ values from both time periods were identical, see above), perhaps suggesting that their bulk diet contained less lipids and protein and was of somewhat lower quality (Ambrose & Norr, 1993; Tieszen & Fagre, 1993; Iacumin et al., 2000; Clementz et al., 2009). That said, both pre- and post-LGM $\Delta^{13}C_{sc-col}$ spacings are typical of large herbivores in the wild (~+7 to +10 ‰) (Sullivan & Krueger, 1981; Lee-Thorp et al., 1989; Bocherens et al., 1994; Iacumin et al., 2000).

4.4.5 Collagen $\delta^{15}N$ values

As previously described, higher $\delta^{15}N_{col}$ values are associated with the pre-LGM horses and lower $\delta^{15}N_{col}$ values are associated with the post-LGM horses (Fig. 4.4), and low post-LGM $\delta^{15}N_{col}$ values have also been reported from elsewhere (see section 4.4.1). Decreases in soil and plant $\delta^{15}N$ values are known to accompany increased precipitation and soil moisture availability (e.g. Austin & Vitousek, 1998; Amundson et al., 2003; Szpak et al., 2013). Accordingly, Iacumin et al. (2000) suggested that low $\delta^{15}N_{col}$ values of reindeer in Siberia reflected increased precipitation after the LGM and that melting ice sheets and permafrost increased the amount of soil moisture in areas adjacent to the glaciers. Richard and Hedges (2003) proposed that low $\delta^{15}N_{col}$ values of horses and deer from northwest Europe arose from a combination of changing soil organic $\delta^{15}N$ values

during cold periods (because of increased retention of ¹⁵N-depleted volatile compounds), and low δ^{15} N values in soils that were undergoing regeneration after deglaciation. The δ^{15} N_{col} values of a consumer are mainly determined by their diet, which in the case of horses refers to the type of plants eaten (additional factors such as stress can also influence δ^{15} N values, e.g. Fuller et al., 2005; see section 1.2.1). Stevens and Hedges (2004) reviewed the many factors that can lead to low plant δ^{15} N values (and therefore δ^{15} N_{col} values), and concluded that many might be at play, including a delay in the response of plants and bone collagen to deglaciation, post-LGM soil development, and increased water availability.

Because of the possible influence of plant δ^{15} N values on δ^{15} N_{col} values, it is helpful to review processes that cause changes in plant δ^{15} N values. Most plants (with the exception of legumes, which live in symbiosis with bacteria attached to their roots) do not utilize atmospheric nitrogen directly; rather, they rely on soil inorganic nitrogen (Stevens & Hedges, 2004 and references therein; Ambrose, 1993). Most of this nitrogen comes from soil nitrates, ammonia, ammonium, and animal urea (Ambrose, 1993). Some of these products are produced by soil microbial processes (as part of the nitrogen cycle), which fractionate ¹⁵N. The main soil transformations produced by microbes include mineralization, nitrification, and denitrification (Dawson, 2002 and references therein). These processes lead to soil products that have low δ^{15} N values relative to the soil in which they were produced (Dawson, 2002 and references therein).

Is it possible that an increase in nitrification and denitrification processes (which produce N₂O) following the LGM caused lower soil and consequently lower plant δ^{15} N values? Atmospheric concentrations of N₂O tend to be higher in warm intervals (e.g. interglacials) than in colder periods (e.g. LGM) (Wolff & Spahni, 2007 and references therein; Flückiger et al., 2004). The increase in soil disturbance from LGM ice sheet advance and retreat (freeze/thaw) could have led to increased N₂O production (which would lower soil and plant δ^{15} N values). For example, Christensen and Christensen (1991) observed that freeze/thaw cycles increased denitrification in soils. However, identifying these processes as the cause for low δ^{15} N plant (and therefore collagen) values might oversimplify the different nitrogen transformations involved in the nitrogen cycle, which vary geographically and with different environmental conditions (Dawson et al.,

2002 and references therein). For example, modern terrestrial sources of N_2O in high latitudes are relatively small, and can be assumed to have been even less significant during glacial times; temperature and precipitation changes can also strongly influence the source strength in these regions (Flückiger et al., 2004 and references therein).

In the Arctic, δ^{15} N values can vary with plant type. For example, spruce trees, evergreen shrubs, and deciduous grass in Alaska can have a δ^{15} N value of -7.7, -4.3. and +0.9 ‰, respectively (Schulze et al., 1994; Bocherens, 2003; Ben-David, 2001). This pattern of δ^{15} N values among vegetation types suggests the possibility that the horses incorporated more browse into their diets as a result of changing vegetation in the region or as a means to compensate for poorer food availability/quality proximal to the ice sheets. However, such an explanation is not well supported by the very similar structural carbonate and collagen $\delta^{13}C_{diet}$ values measured for the pre- and post-LGM horses (see section 4.4.2). Also, as Stevens and Hedges (2004) pointed out, deer, reindeer, and horses in Eurasia exhibited similar ¹⁵N-depletion; if diet was the cause they all would have had to choose similar diets and foods with low δ^{15} N values, which seems unlikely. In summary, the low post-LGM horse $\delta^{15}N_{col}$ values may reflect low plant $\delta^{15}N$ values, which can be attributed to a combination of the factors listed above, as well as other factors that affect the nitrogen cycle, including permafrost activity, nutrient availability in the soil, and microbial activity. A comprehensive study of Pleistocene δ^{15} N plant values is needed to evaluate their role in determining the $\delta^{15}N_{col}$ values reported here for the preand post-LGM horses from Alberta.

Chapter 5

Conclusions and Future Work

5.1 Major Conclusions

- 1. Coeval bone and tooth enamel in the modern horse sample show no significant differences between these tissues for each of $\delta^{13}C_{sc}$, $\delta^{18}O_{sc}$, and $\delta^{18}O_p$ values. In other words, bioapatite carbon and oxygen isotopic signatures are recorded in almost the same way in bone and tooth enamel from this animal. This observation implies that tissue-dependent physiological and/or metabolic processes do not cause differences between bone and tooth oxygen or carbon isotopic compositions in horses. Since bone and tooth enamel record the diet and drinking water isotopic signals in virtually identical fashion, differences in $\delta^{13}C_{sc}$, $\delta^{18}O_{sc}$, and/or $\delta^{18}O_p$ values between these tissues can be used to infer migration behaviour and diet changes in horses.
- 2. Previously determined relationships between $\delta^{18}O_{sc}$ and $\delta^{18}O_{p}$ values in mammalian bioapatite have been used to identify *post-mortem* alteration in the Pleistocene horse samples from Alberta. Together, $\Delta^{18}O_{sc-p}$ spacings of <+7 ‰ (following Pellegrini et al., 2011) and an array of $\delta^{18}O_{p}$ versus $\delta^{18}O_{sc}$ values with a slope of <1 (following Zazzo et al., 2004b) suggest that microbially mediated *post-mortem* alteration may have affected the bioapatite phosphate component of most bone and some tooth samples. Most tooth samples have likely retained primary $\delta^{18}O_{sc}$ values, as indicated by their CI values, and for a few samples, by variations in the $\delta^{18}O_{sc}$ values of serially sampled enamel that match patterns expected for seasonal variation.
- 3. The δ¹⁸O_{dw} water values calculated for Alberta using bulk enamel phosphate and structural carbonate oxygen isotopic compositions are similar for the Pleistocene horses from both before and immediately after the Last Glacial Maximum (LGM). This observation suggests that the horses drank from sources with very similar oxygen isotopic compositions. The broader implication is that climatic conditions controlling the oxygen isotopic composition of meteoric water were similar during the lifetimes of the pre- and post-LGM horses.

- 4. The bulk $\delta^{13}C_{diet}$ values calculated from enamel structural carbonate compositions of all Pleistocene horse samples examined here are very similar between bone and tooth tissues, as is also the case for the $\delta^{13}C_{diet}$ values calculated from bulk collagen compositions. All of these measures indicate consumption exclusively of C_3 plants and also suggest that horse diets did not vary significantly between infancy and adulthood. There is also very little difference in these carbon isotopic compositions between pre- and post-LGM horse samples. This indicates ecological conditions yielding very similar carbon isotopic compositions for grazed vegetation during both time periods.
- 5. The $\delta^{18}O_{sc}$ values of most serial samplings of the Pleistocene tooth enamel exhibited irregular variation, interpreted to indicate larger home ranges and/or consumption of varied drinking water sources by these horses. The $\delta^{18}O_{sc}$ values for a few serially sampled teeth are consistent with seasonal variations in drinking water expected within a smaller home range. Serially sampled $\delta^{13}C_{sc}$ values display the range of variation typical for a C₃ environment.
- 6. The post-LGM horses have much lower $\delta^{15}N_{col}$ values than the pre-LGM horses. This difference may indicate lower plant $\delta^{15}N$ values following the LGM, perhaps arising from a significant change in the parameters controlling the nitrogen cycle following the LGM. Regardless of its origin, this shift in $\delta^{15}N_{col}$ values may provide a coarse screening test for distinguishing between horses (and perhaps other megafauna) that lived before and after the LGM.

5.2 Future Work

To use equid enamel and bone tissue $\delta^{18}O_{sc}$, $\delta^{18}O_{p}$, and $\delta^{13}C_{sc}$ values in reconstruction of Pleistocene (and other) environments, it is important to verify first whether or not these isotopic signals are recorded free of differences arising from metabolic or physiological processes during formation of each tissue. This has been shown to be the case for the one modern horse specimen here. However, to extend this finding more broadly with real confidence requires a much larger sample population. Ideally, the isotopic composition of drinking water and diet for domestic horses would be monitored in a controlled setting from birth to death. Such controlled studies are also needed for other large animals that can serve as models for isotopic systematics in other Pleistocene megafauna (>44 kg).

Preservation of primary isotopic compositions in bone and enamel is necessary for these tissue signatures to be useful in palaeoenvironmental reconstruction. Given the possibility of *post-mortem* alteration of bioapatite phosphate and/or structural carbonate components in some of the horse bone and tooth samples analyzed here, it would be helpful to identify more directly the extent and mechanism(s) (i.e. inorganic or microbial) of alteration. A good starting point is the use of scanning electron microscopy to identify physical traces of microbial activity such as tunneling or tissue reworking (Tütken et al., 2008, and references therein). Such observations could then be followed by *in situ* isotopic analysis of affected versus unaffected areas at the µm level, such as described by Brady et al. (2008).

Given the very large shift in nitrogen isotopic compositions observed between the pre- and post-LGM horses, it would be of great interest to study the influence of environmental parameters on plant (and faunal) δ^{15} N values. In particular, detailed investigations are recommended to determine: (i) how changing temperature and moisture conditions directly affect plant and herbivore δ^{15} N values, and (ii) how nitrogen isotopic fractionations in the water-soil-plant nitrogen cycle may have changed from pre-to post-LGM times.

To further understand the Late Pleistocene environment of Alberta, more studies are warranted of the stable isotopic composition of temporally and geographically related megafauna. The stable isotopic compositions of a suite of well-dated megafauna – spanning time from before the LGM into the Holocene – would provide more detailed description of terrestrial environmental change. Stable carbon isotopic analysis of a variety of megafauna, in particular, would capture the range of dietary items available to contemporary animals and hence the ecological variability of the landscape. Oxygen and carbon isotopic analysis of serially sampled enamel from a large population of herbivores during this time span would also provide a more robust characterization of seasonal climatic patterns and vegetation availability.

References

- Alley, R.B. (2007). Wally was right: predictive ability of the North Atlantic "Conveyor Belt" hypothesis for abrupt climate change. *Annual Review of Earth and Planetary Science*, *35*, 241–272.
- Ambrose, S.H. (1990). Preparation and characterization of bone and tooth collagen for isotopic analysis. *Journal of Archaeological Science*, *17*(4), 431–451.
- Ambrose, S.H. (1993). Isotopic analysis of paleodiets: Methodological and interpretive considerations. In M.K. Sandford (Ed.), *Investigation of Ancient Human Tissue: Chemical Analyses in Anthropology* (pp. 59–130). Langhorne PA, USA: Gordon and Breach Science Publishers.
- Ambrose, S.H, & Norr L. (1993). Experimental evidence for the carbon isotope ratios of whole diet and dietary protein to those of bone collagen and carbonate. In J.B Lambert, G. Grupe (Eds.), *Prehistoric Human Bone: Archaeology at the Molecular Level* (pp. 1–37). New York, USA: Springer-Verlag.
- Ambrose, S.H., Butler, B.M., Hanson, D.B., Hunter-Anderson, R.L., & Krueger, H.W. (1997). Stable isotopic analysis of human diet in the Marianas Archipelago, Western Pacific. *American Journal of Physical Anthropology*, 104, 343–361.
- Amundson, R., Austin, A.T., Schuur, E.A.G., Yoo, K., Matzek, V., Kendall, C., Uebersax, A., Brenner, D., & Baisden, W.T. (2003). Global patterns of the isotopic composition of soil and plant nitrogen. *Global Biogeochemical Cycles*, 17(1), 1031–1040.
- Anderson, P.M., & Lozhkin, A.V. (2001). The Stage 3 interstadial complex (Karginskii/middle Wisconsinan interval) of Beringia: variations in paleoenvironments and implications for paleoclimatic interpretations. *Quaternary Science Review*, 20, 93–125.
- Austin, A., & Vitousek, P.M. (1998). Nutrient dynamics on a precipitation gradient in Hawai'i. Oecologia, 113, 519–529.
- Ayliffe, L.K., Lister, A.M., & Chivas, A.R. (1992). The preservation of glacialinterglacial climatic signatures in the oxygen isotopes of elephant skeletal phosphate. *Palaeogeography, Palaeoclimatology, Palaeoecology, 99*, 179– 191.
- Ayliffe, L.K., Cerling, T.E., Robinson, T., West, A.G., Sponheimer, M., Passey, B.H., Hammer, J., Roeder, B., Dearing, M.D., & Ehleringer, J.R. (2004). Turnover of carbon isotopes in tail hair and breath CO₂ of horses fed on isotopically varied diet. *Oecologia*, 139, 11–22.
- Balasse, M. (2003a). Determining Sheep Birth Seasonality by Analysis of Tooth Enamel Oxygen Isotope Ratios: The Late Stone Age site of Kasteelberg (South Africa). *Journal of Archaeological Science*, 30, 205–215.
- Balasse, M. (2003b). Potential biases in sampling design and interpretation of intra-tooth isotope analysis. *International Journal of Osteoarchaeology*, *13*, 3–10.
- Ben-David, M., Shochat, E., & Adams, L.G. (2001). Utility of stable isotope analysis in studying foraging ecology of herbivores: examples from moose and caribou. *Alces*, 37(2), 421–434.

- Bendrey, R. (2007). New methods for the identification of evidence for bitting on horse remains from archaeological sites. *Journal of Archaeological Science*, *34*(7), 1036–1050.doi:10.1016/j.jas.2006.09.010.
- Bigelow, N.H., Brubaker, L.B., Edwards, M.E., Harrison, S.P., Prentice, I.C., Anderson, P.M., Andreev, A.A., Bartlein, P.J., Christensen, T.R., Cramer, W., Kaplan, J.O., Lozhkin, A.V., Matveyeva, N.V., Murray, D.F., McGuire, A.D., Razzhivin, V.Y., Ritchie, J.C., Smith, B., Walker, D.A., Gajewski, K., Wolf, V., Holmqvist,
- B.H., Igarashi, Y., Kremenetskii, K., Paus, A., Pisaric, M.F.J., & Volkova, V.S. (2003). Climate change and Arctic ecosystems: 1. Vegetation changes north of 55°N between the last glacial maximum, mid-Holocene, and present. *Journal of Geophysical Research*, 108(D19), 25.
- Blake, R.E., O'Neil, J.R., & Garcia, G.A. (1997). Oxygen isotope systematics of biologically mediated reactions of phosphate. I. Microbial degradation of organophosphoruscompounds. *Geochimica et Cosmochimica Acta*, 61, 4411– 4422.
- Bocherens, H., Fizet, M., Mariotti, A., Gangloff, R.A., & Burns, J.A. (1994).
 Contribution of isotopic biogeochemistry (¹³C, ¹⁵N, ¹⁸O) to the paleoecology of mammoths (*Mammuthus Primigenius*). *Historical Biology*, 7, 187–202.
- Bocherens, H., & Drucker, D. (2003). Trophic level isotopic enrichment of carbon and nitrogen in bone collagen: Case studies from recent and ancient terrestrial ecosystems. *International Journal of Osteoarchaeology*, *13*(1–2), 46–53.
- Brady, A.L., White, C.D., Longstaffe, F.J., & Southam, G. (2008). Investigating intrabone isotopic variations in bioapatite using IR-laser ablation and micromilling: Implications for identifying diagenesis? *Palaeogeography, Palaeoclimatology, Palaeoecology*, 266, 190–199.
- Brookman, T.H., & Ambrose, S.H. (2012). Seasonal variation in kangaroo tooth enamel oxygen and carbon isotopes in southern Australia. *Quaternary Research*, 78, 256–265.
- Bryant, J.D., Luz, B., & Froeclich, P.N. (1994). Oxygen isotopic composition of fossil horse tooth phosphate as a record of continental paleoclimate. *Palaeogeography, Palaeoclimatology, Palaeoecology, 107*, 303–316.
- Bryant, J.D., & Froelich, P.N. (1995). A model of oxygen isotopic fractionation in body water of large mammals. *Geochimica et Cosmochimica Acta*, 59(21), 4523–4537.
- Bryant, J.D., Koch, P.L., Froelich, P.N., Showers, W.J., & Genna, B.J. (1996a). Oxygen isotope partitioning between phosphate and carbonate in mammalian apatite. *Geochimica et Cosmochimica Acta*, 60, 5145–5148.
- Bryant, J.D., Froelich, P.N., Showers, W.J., & Genna, B.J. (1996b). Biologica and climatic signals in the oxygen isotopic composition of Eocene-Oligocene equid enamel phosphate. *Palaeogeography, Palaeoclimatology, Palaeoecology, 126*, 75–89.
- Bryant, J.D., Froelich, P.N., Showers, W.J., & Genna, B.J. (1996c). A tale of two quarries: biologic and taphonomic signatures in the oxygen isotope composition of tooth enamel phosphate from modern and Miocene equids. *Palaios*, 11, 397– 408.

- Burns, J.A. (1996). Vertebrate paleontology and the alleged Ice-Free Corridor: The meat of the matter. *Quaternary International*, *32*, 107–112.
- Burns, J.A. (2010). Mammalian faunal dynamics in Late Pleistocene Alberta, Canada. *Quaternary International*, 217, 37–42.
- Cerling, T.E., & Harris, J.M. (1999). Carbon isotope fractionation between diet and bioapatite in ungulate mammals and implications for ecological and paleoecological studies. *Oecologia*, *120*(3), 347–363.
- Child, A.M. (1995). Towards an understanding of the microbial decomposition of archaeological bone in the burial environment. *Journal of Archaeological Science*, *22*, 165–174.
- Chillón, B.S., Alberdi, M.T., Leone, G., Bonadonna, F.P., Stenni, B., & Longinelli, A. (1994). Oxygen isotopic composition of fossil equid tooth and bone phosphate: an archive of difficult interpretation. *Palaeogeography, Palaeoclimatology, Palaeoecology, 107*, 317–328.
- Christensen, S., & Christensen, B.T. (1991). Organic matter available for denitrification in different soil fractions: effect of freeze/thaw cycles and straw disposal. *Journal* of Soil Science, 42, 637-647.
- Clayton, R.N., & Mayeda, T.K. (1963). The use of bromine pentafluoride in the extraction of oxygen from oxides and silicates for isotopic analysis. *Geochimica et Cosmochimica Acta*, 27, 43–52.
- Clementz, M.T., Fox-Dobbs, K., Wheatley, P.V., Koch, P.L., & Doak, D.F. (2009). Revisiting old bones: coupled carbon isotope analysis of bioapatite and collagen as an ecological and palaeoecological tool. *Geological Journal*, 44, 605–620.
- Collins, M.J., Nielsen-Marsh, C.M., Hiller, J., Smith, C.I., Roberts, J.P., Prigodich, R.V., Wess, T.J., Csapó, J., Millard, A.R., & Turner-Walker, G. (2002). The survival of organic matter in bone: a review. *Archaeometry*, 44, 383–394.
- Coplen, T.B. (1994). Reporting stable hydrogen, carbon, and oxygen isotopic abundances. *Pure and Applied Chemistry*, 66, 271–276.
- Coplen, T.B. (1996). New guidelines for reporting stable hydrogen, carbon, and oxygen isotope-ratio data. *Geochimica et Cosmochimica Acta, 60*(17), 3359–3360.
- Coplen, T.B. (2011). Guidelines and recommended terms for expression of stableisotope-ratio and gas-ratio measurement results. *Rapid Communications in Mass Spectrometry*, 25, 2538–2560.
- Crowley, B.E., Carter, M.L., Karpanty, S.M., Zihlman, A.L., Koch, P.L., & Dominy, N.J. (2010). Stable carbon and nitrogen isotope enrichment in primate tissues. *Oecologia*, 164, 611–626.
- Crowson, R.A., Showers, W.J., Wright, E.K., & Hoering, T.C. (1991). Preparation of phosphate samples for isotopic analysis. *Analytical Chemistry*, *63*, 2397–2400.
- Dansgaard, W. (1964). Stable isotopes in precipitation. Tellus, 16, 436-468.
- Daux, V., Lécuyer, C., Héran, M-A., Amiot, R., Simon, L., Fourel, F., Martineau, F., Lynnerup, N., Reychler, H., & Escarguel, G. (2008). Oxygen isotope fractionation between human phosphate and water revised. *Journal of Human Evolution*, 55, 1138–1147.
- Dawson, T.E., Mambelli, S., Plamboeck, A.H., Templer, P.H., & Tu, K.P. (2002). Stable isotopes in plant ecology. *Annual Review of Ecology, Evolution, and Systematics,* 33, 507–59.

- DeNiro, M.J., & Epstein, S. (1981). Influence of diet on the distribution of nitrogen isotopes in animals. *Geochimica et Cosmochimica Acta*, 45, 341–351.
- DeNiro, M.J. (1985). Post-mortem preservation and alteration of "in vivo" bone collagen ratios: implications for paleodietary analysis. *Nature*, *317*, 806–809.
- Dixon, P.M. (2002). The gross, histological, and ultrastructural anatomy of equine teeth and their relationship to disease. *American Association of Equine Practitioners Annual Convention, 48*, 421–437.
- Dobberstein, R., Collins, M., Craig, O., Taylor, G., Penkman, K., & Ritz-Timme, S. (2009). Archaeological collagen: why worry about collagen diagenesis? *Archaeological andAnthropological Sciences*, 1, 31–42.
- Dotsika, E., Zisi, N., Tsoukala, E., Poutoutkis, D., Lykoudis, S., & Giannakopolous, A. (2011). Palaeoclimatic information from isotopic signatures of Late Pleistocene Ursus ingressus bone and teeth apatite (Loutra Arideas Cave, Macedonia, Greece). Quaternary International, 245, 291–301.
- Dredge, L.A., & Thorleifson, L.H. (1987). The Middle Wisconsinan history of the Laurentide Ice Sheet. Géographie physique et Quaternaire, 41(2), 215–235. Driessens, F.C.M., & Verbeeck, R.M.H. (1990). Biominerals. Boca Raton, Florida: CRC Press.
- Dyke, A.S., & Prest, V.K. (1987). Late Wisconsinan and Holocene history of the Laurentide ice sheet. *Géographie physique et Quaternaire*, 41, 237–263.
- Dyke, A.S., Moore, A.J., & Robertson, L. (2003). *Deglaciation of North America. Open File 1574*. Ottawa: Geological Survey of Canada.
- Feranec, R.S., Hadley, E.A., & Payton, A. (2009). Stable isotopes reveal seasonal competition for resources between late Pleistocene bison (*Bison*) and horse (*Equus*) from Rancho La Brea, southern California. *Palaeogeography*, *Palaeoclimatology*, *Palaeoecology*, 271, 153–160.
- Firsching, F.H. (1961). Precipitation of silver phosphate from homogeneous solution. *Analytical Chemistry*, 33, 873–887.
- Flückiger, J., Blunier, T., Stauffer, B., Chappellaz, J., Spahni, R., Kawamura, K., Schwander, J., Stocker, T.F., & Dahl-Jensen, D. (2004). N₂O and CH₄ variations during the last glacial epoch: Insight into global processes. *Global Biogeochemical Cycles, 18*, 1–14.
- Fogel, M.L., Tuross, N., & Owsley, D. (1989). Nitrogen isotope tracers of human lactation in modern and archaeological populations. In *Annual report of the Director, Geophysical Laboratory, 1988–1989* (pp.111–116). Washington, DC: Carnegie Institution of Washington.
- Fox-Dobbs, K., Leonard, J.A., & Koch, P.L. (2008). Pleistocene megafauna from eastern Beringia: Paleoecological and paleoenvironmental interpretations of stable carbon and nitrogen Isotope and radiocarbon records. *Palaeogeography*, *Palaeoclimatology*, *Palaeoecology*, 261, 30–46.
- Frémondeau, D., Cucchi, T., Casabianca, F., Ughetto-Monfrim, J., Horard-Herbin, M.-P., & Balasse, M. (2012). Seasonality of birth and diet of pigs from stable isotope analyses of tooth enamel (δ^{18} O, δ^{13} C): a modern reference data set from Corsica, France. *Journal of Archaeological Science*, *39*, 2023–2035.

- Fricke, H.C., Clyde, W.C., & O'Neil, J.R. (1998). Intra-tooth variations in δ^{18} O (PO₄) of mammalian tooth enamel as a record of seasonal variations in continental climate variables. *Geochimica et Cosmochimica Acta*, 62(11), 1839–1850.
- Friedli, H., Lotscher, H., Oeschger, H., Siegenthaler, U., & Stauffer, B. (1986). Ice core record of the ¹³C/¹²C ratio of atmospheric CO₂ in the past two centuries. *Nature* 324, 237–238.
- Friedman, I., & O'Neil, J.R. (1977). Compilation of stable isotope fractionation factors of geochemical interest. In M. Fleischer (Ed.), *Data of Geochemistry 6th Edition*. Washington, D.C.: Government Printing Office.
- Fuller, B.T., Fuller, J.L., Sage, N.E., Harris, D.A., O'Connell, T.C., & Hedges, R.E.M. (2005). Nitrogen balance and δ^{15} N: why you're not what you eat during nutritional stress. *Rapid Communications in Mass Spectrometry*, *19*, 2497–2506.
- Garvie-Lok, S.J., Varney, T.L. & Katzenberg, M.A. (2004). Preparation of bone carbonate for stable isotope analysis: the effects of treatment time and acid concentration. *Journal of Archaeological Science*, *31*(6), 763–776.
- Germonpré, M. (1997). AMS dating of the Upper Magdalenian horizon 1 at Goyet Cave. Bulletin de L'Institut Royal des Sciences Naturelles de Belgique (Sciences de la Terre), 67, 167–182.
- Gill, J.L., Williams, J.W., Jackson, S.T., Lininger, K.B., & Robinson, G.S. (2009). Pleistocene megafaunal collapse, novel plant communities, and enhanced fire regimes in NorthAmerica. *Science*, 326, 1100–1103.
- Guthrie, R.D. (1968). Paleoecology of the large-mammal community in interior Alaska during the Late Pleistocene. *American Midland Naturalist*, 79, 346–36.
- Guthrie, R.D. (1982). Mammals of the mammoth steppe as paleoenvironmental indicators. In D.M. Hopkins, J.V. Matthews Jr., C.E. Schweger, S.B. Young, (Eds.), *Paleoecology of Beringia* (pp. 307–326). New York: Academic Press.
- Guthrie, R.D. (1984). Mosaics, allelochemics and nutrients: an ecological theory of late Pleistocene megafaunal extinctions. In P.S. Martin, R.G. Klein (Eds.), *Quaternary Extinctions: a Prehistoric Revolution* (pp. 259–298). Tuscon: University of Arizona Press.
- Guthrie, R.D. (2001). Origin and causes of the mammoth steppe: a story of cloud cover, woolly mammal tooth pits, buckles, and inside-out Beringia. *Quaternary Science Reviews*, 20, 549–574.
- Guthrie, R.D. (2003). Rapid body size decline in Alaskan Pleistocene horses before extinction. *Nature*, 426, 169–171.
- Harington, C.R. (1990). Vertebrates of the Last Interglaciation in Canada: A review, with new data. *Géographie physique et Quaternaire*, 44(3), 375–387.
- Harrington, C.R. (2005). The eastern limit of Beringia: Mammoth remains from Banks and Melville Islands, Northwest Territories. *Arctic*, *58*(4), 361–369.
- Head, M.J., Gibbard, P., & Salvador, A. (2008). The Quaternary: its character and definition. *Episodes*, *31*(2), 234–238.
- Heaton, T.H.E. (1999). Spatial, species and temporal variations in the ${}^{13}C/{}^{12}C$ ratios of C₃ plants: implications for palaeodiet studies. *Journal of Archaeological Science*, 26, 637–649.

- Hedges, R.E.M., Housley, R.A., Bronk Ramsey, C., & Van Klinken, G.J. (1993). Radiocarbon dates from the Oxford AMS system: archaeometry datelist 17. *Archaeometry*, 35(2), 305–326.
- Hedges, R.E.M., Clement, J.G., Thomas, C.D.L., & O'Connell, T.C. (2007). Collagen turnover in the adult femoral mid-shaft: Modeled From anthropogenic radiocarbon tracer measurements. *American Journal of Physical Anthropology*, 133, 808–816.
- Hillson, S. (2005). Teeth. New York: Cambridge University Press.
- Higgins, P., & MacFadden, B.J. (2004). "Amount effect" recorded in oxygen isotopes of Late Glacial horse (*Equus*) and bison (*Bison*) teeth from the Sonoran and Chihuahuan deserts, southwestern United States. *Palaeogeography*, *Palaeoclimatology*, *Palaeoecology*, 206, 337–353.
- Hoefs, J. (2009). Stable Isotope Geochemistry. Berlin: Springer.
- Hoppe, K.A., Stover, S.M., Pascoe, J.R., & Amundson, R. (2004a). Tooth enamel biomineralization in extant horses: implications for isotopic microsampling. *Palaeogeography, Palaeoclimatology, Palaeoecology, 206*, 355–365. doi: 10.1016/j.palaeo.2004.01.012.
- Hoppe, K.A., Amundson, R., Vavra, M., McClaran, M.P., & Anderson, D.L. (2004b). Isotopic analysis of tooth enamel carbonate from modern North American feral horses: implications for paleoenvironmental reconstructions. *Palaeogeography*, *Palaeoclimatology*, *Palaeoecology*, 203, 299–311.
- Hoppe, K.A., Stuska, S., & Amundson, R. (2005). The implications for paleodietary and paleoclimatic reconstructions of intrapopulation variability in the oxygen and carbon isotopes of teeth from modern feral horses. *Quaternary Research*, 64, 138–146.
- Huertas, A.D., Iacumin, P., Stenni, B., Chillón, S., & Longinelli, A. (1995). Oxygen isotope variations of phosphate in mammalian bone and tooth enamel. *Geochimica et Cosmochimica Acta*, *59*(20), 4299–4305.
- Husain, M.M., Cherry, J.A., & Shaun, K.F. (2004). The persistence of a large stagnation zone in a developed regional aquifer, southwestern Ontario. *Canadian Geotechnical Journal*, *41*, 943–958.
- Iacumin, P., Bocherens, H., Mariotti, A., & Longinelli, A. (1996). Oxygen isotope analyses of co-existing carbonate and phosphate in biogenic apatite; a way to monitor diagenetic alteration of bone phosphate? *Earth and Planetary Science Letters*, 142, 1–6.
- Iacumin, P., Nikolaev, V., & Ramigni, M. (2000). C and N stable isotope measurements on Eurasian fossil mammals, 40 000 to 10 000 years BP: Herbivore physiologies and palaeoenvironmental reconstruction. *Palaeogeography, Palaeoclimatology, Palaeoecology, 163*, 33–47.
- IAEA. (2001). GNIP Maps and Animations, International Atomic Energy Agency, Vienna. Retrieved February, 2013, from http://isohis.iaea.org.
- IAEA/WMO. (2010). *Global Network of Isotopes in Precipitation, The GNIP Database*. Retrieved from http://www.iaea.org/water.
- Jass, C.N., Burns, J.A., & Millot, P.J. (2011). Description of fossil muskoxen and relative abundance of Pleistocene megafauna in central Alberta. *Canadian Journal of Earth Science*, 48, 793–800.

- Jim, S., Ambrose, S.H., & Evershed, R.P. (2004). Stable carbon isotopic evidence for differences in the dietary origin of bone cholesterol, collagen and apatite: Implications for their use in palaeodietary reconstruction. *Geochimica et Cosmochimica Acta*, 68(1), 61–72.
- Koch, P.L., Tuross, N., & Fogel, M.L. (1997). The effects of sample treatment and diagenesis on the isotopic integrity of carbonate in biogenic hydroxylapatite. *Journal of Archaeological Science*, 24, 417–429.
- Kohn, M.J., & Cerling, T.E. (2002). Stable isotope compositions of biological apatite. In M.J. Kohn, J. Rakovan, J.M. Hughes (Eds.), *Phosphates: Geochemical, Geobiological, and Materials Importance. Reviews in Mineralogy and Geochemistry vol. 48* (pp. 455–488). Washington, DC: Mineralogical Society of America.
- Kim S-T., & O'Neil, J.R. (1997). Equilibrium and nonequilibrium oxygen isotope effects in synthetic carbonates. *Geochimica et Cosmochimica Acta*, 61, 3461–3475.
- Koch, P.L., Fogel, M.L., & Tuross, N. (1994). Tracing the diets of fossil animals using stable isotopes. In K. Lajtha, R.H. Michener (Eds.), *Methods in Ecology, Stable Isotopes in Ecology and Environmental Science* (pp.63–92). Oxford, UK: Blackwell Scientific Publications.
- Koch, P.L. (1998). Isotopic reconstruction of past continental environments. *Annual Review of Earth and Planetary Sciences*, 26, 573–613.
- Koch, P.L., & Barnosky, A.D. (2006). Late Quaternary extinctions: State of the debate. *Annual Review of Ecology, Evolution, and Systematics*, *37*, 215–50.
- Kolodny Y., Luz B., & Navon O. (1983). Oxygen isotope variations in phosphate of biogenic apatites: I. Fish bone apatite-rechecking the rules of the game. *Earth and Planetary Science Letters*, 64, 398–404.
- Kohn, M.J., & Cerling, T.E. (2002). Stable isotope compositions of biological apatite. In M.J. Kohn, J. Rakovan, J.M. Hughes (Eds.), *Reviews in Mineralogy and Geochemistry vol. 48. Phosphates: Geochemical, Geobiological, and Materials Importance* (pp.455–488). Washington, DC: Mineralogical Society of America.
- Lawrence, L.A. (2005). Principles of bone development. In J.D. Pagan (Ed.), Advances in Equine Nutrition III (pp. 289–294). Nottingham, United Kingdom: Nottingham University Press.
- Lecuyer, C., Grandjean, p., & Sheppard, S.M.F. (1999). Oxygen isotope exchange between dissolved phosphate and water at temperatures ≤135 °C: Inorganic versus biological fractionations. *Geochimica et Cosmochimica Acta*, 63(6), 855– 862.
- Lecuyer, C., Balter, V., Martineau, F., Fourel, F., Bernard, A., Amiot, R., Gardien, V., Otero, O., Legendre, S., Panczer, G., Simon, L., & Martini, R. (2010). Oxygen isotope fractionation between apatite-bound carbonate and water determined from controlled experiments with synthetic apatites precipitated at 10–37 °C. *Geochimica et Cosmochimica Acta*, 74, 2072–2081.
- Lee-Thorp, J., & van der Merwe, N.J. (1987). Carbon isotope analysis of fossil bone apatite. *South African Journal of Science*, 83, 712–715.
- Lee-Thorp, J.A., Sealy, J.C., van der Merwe, N.J. (1989). Stable carbon isotope ratio differences between bone collagen and bone apatite, and their relationship to diet. *Journal of Archaeological Science*, *16*, 585–599.

Lee-Thorp, J.A. (2008). On isotopes and old bones. Archaeometry, 50(6), 925–950.

- LeGeros, R.Z. (1991). Calcium Phosphates in Oral Biology and Medicine (Monographs in Oral Science). New York: Karger. Libby, W.F., Anderson, E.C., & Arnold, J.R. (1949). Age determination by radiocarbon content: world-wide assay of natural radiocarbon. Science, 109(2827), 227–228.
- Linklater, W.L., Cameron, E.Z., Stafford, K.J., & Veltman, C.J. (2000). Social and spatial structure and range use by Kaimanawa wild horses (*Equus caballus*: Equidae). *New Zealand Journal of Ecology*, 24(2), 139–152.
- Loftus, E., & Sealy, J. (2012). Technical note: Interpreting stable carbon isotopes in human tooth enamel: An examination of tissue spacings from South Africa. *American Journal of Physical Anthropology*, *147*, 499–507.
- Longin, R. (1971). New method of collagen extraction for radiocarbon dating. *Nature*, 230, 241–242.
- Longinelli, A. (1984). Oxygen isotopes in mammal bone phosphate: a new tool for paleohydrological and paleoclimatological research? *Geochimica et Cosmochimica Acta*, 48, 385–390.
- Luz, B., Kolodny, Y., & Horowitz, M. (1984). Fractionation of oxygen isotopes between mammalian bone-phosphate and environmental water. *Geochimica et Cosmochimica Acta*, 48, 1689–1693.
- MacFadden, B.J. (1986). Fossil horses from "Eohippis" (*Hyracotherium*) to *Equus*: Scaling, Cope's Law, and the evolution of body size. *Paleobiology*, *12*(4), 355–369.
- MacFadden, B. (1988). Fossil horses from "Eohippus"(*Hyracotherium*) to *Equus*, 2: Rates of dental evolution revisited. *Biological Journal of the Linnean Society*, 35, 37–48.
- MacFadden, B.J., Cerling, T.E., Harris, J.M., & Prado, J. (1999). Ancient latitudinal gradients of C_3/C_4 grasses interpreted from stable isotopes of New World Pleistocene horse (*Equus*) teeth. *Global Ecology and Biogeography*, 8, 137–149.
- MacDonald, G.M., & McLeod, T.K. (1996). The Holocene closing of the "ice-free" corridor: a biogeographical perspective. *Quaternary International*, *32*, 87–95.
- MacDonald, G.M., Beilman, D.W., Kuzmin, Y.V., Orlova, L.A., Kremenetski, K.V., Shapiro, B., Wayne, R.K., & Valkenburgh, B.V. (2012). Pattern of extinction of the woolly mammoth in Beringia. *Nature Communications*, 3(893), 1–8.
- Mann, D.H., Peteet, D.M., Reanier, R.E., & Kunz, M.L. (2002). Responses of an arctic landscape to Lateglacial and early Holocene climatic changes: the importance of moisture. *Quaternary Science Reviews*, 21, 997–1021.
- Marino, B.D., & McElroy, M.B. (1991). Isotopic composition of atmospheric CO₂ inferred from carbon in C₄ plant cellulose. *Nature*, 249, 127–131.
- Martin, C., Bentaleb, I., Kaandorp, R., Iacumin, P., & Chatri, K. (2008). Intra-tooth study of modern rhinoceros enamel δ^{18} O: Is the difference between phosphate and carbonate δ^{18} O a sound diagenetic test? *Palaeogeography, Palaeoclimatology, Palaeoecology, 266,* 183–189.
- McCort, W.D. (1984). Behaviour of feral horses and ponies. *Journal of Animal Science*, 58, 493–499.

- Metcalfe, J.Z., White, C.D., Longstaffe, F.J., Wrobel, G., Cook, D.C., & Pyburn, K.A. (2009). Isotopic evidence for diet at Chau Hiix, Belize: testing regional models of hierarchy and heterarchy. *Latin American Antiquity*, 20(1), 15–36.
- Metcalfe, J.Z., Longstaffe, F.J., Ballenger, J.A.M., & Haynes Jr., C.V. (2011). Isotopic paleoecology of Clovis mammoths from Arizona. *Proceedings of the National Academy of Sciences*, 108(44), 17916–17920.
- Metcalfe, J.Z. (2011). Late Pleistocene climate and Proboscidean paleoecology in North America: insights from stable isotope compositions of skeletal remains (Ph.D). London, Ont.: Faculty of Graduate Studies, University of Western Ontario.
- Metcalfe, J.Z., & Longstaffe, F.J. (2012). Mammoth tooth enamel growth rates inferred from stable isotope analysis and histology. *Quaternary Research*, 77, 424–432. doi:10.1016/j.yqres.2012.02.002.
- Miller, G.H., Brigham-Grette, J., Alley, R.B., Anderson, L., Bauch, H.A., Douglas, M.S.V., Edwards, M.E., Elias, S.A., Finney, B.P., Fitzpatrick, J.J., Funder, S.V., Herbert, T.D., Hinzman, L.D., Kaufman, D.S., MacDonald, G.M., Polyak, L., Robock, A., Serreze, M.C., Smol, J.P., Spielhagen, R., & White, J.W.C. (2010). Temperature and precipitation history of the Arctic. *Quaternary Science Reviews*, 29, 1679–1715.
- Munro, L.E., Longstaffe, F.J., & White, C.D. (2007). Burning and boiling of modern deer bone: Effects on crystallinity and oxygen isotope composition of bioapatite phosphate. *Palaeogeography, Palaeoclimatology, Palaeoecology, 249*, 90–102.
- Munyikwa, J.K., Feathers, J.K., Rittenour, T.M., & Shrimpton, H.K. (2011). Constraining the Late Wisconsinan retreat of the Laurentide ice sheet from western Canada using luminescence ages from postglacial aeolian dunes. *Quaternary Geochronology*, 6, 407–422. doi: 10.1016/j.quageo.2011.03.010.
- Nardoto, G.B., Godoy, P.B., Barros Ferraz, E.S., Ometto, J.P.H.B., & Martinelli, L.A. (2006). Stable carbon and nitrogen isotopic fractionation between diet and swine tissues. *Scientia Agricola*, *63*, 579–582.
- North, R.L., Barton, D., Crowe, A.S., Dillon, P.J, Dolson, R.M.L., Evans, D.O., Ginn, B.K., Håkanson, L., Hawryshyn, J., Jarjanazi, H., King, J.W., La Rose, J.K.L., León, L., Lewis, C.F.M., Liddle, G.E., Lin, Z.H., Longstaffe, F.J., Macdonald, R.A., Molot, L., Ozersky, T., Palmer, M.E., Quinlan, R., Rennie, M.D., Robillard, M.M., Rodé, D., Rühland, K.M., Schwalb, A., Smol, J.P., Stainsby, E., Trumpickas, J.J., Winter, J.G., & Young, J.D. (2013). The state of Lake Simcoe (Ontario, Canada): the effects of multiple stressors on phosphorus and oxygen dynamics. *Inland Waters*, *3*, 51–74.
- O'Connell, T.C., & Hedges, R.E.M. (1999). Investigations into the effect of diet on modern human hair isotopic values. *American Journal of Physical Anthropology*, *108*, 409–425.
- O'Leary, M.H. (1988). Carbon isotopes in photosynthesis. BioScience, 38(5), 328-336.
- OMAFRA. (2013). Ontario Ministry of Agriculture, Horses- Feeding and Pastures. Retrieved January, 2013, from http://www.omafra.gov.on.ca.
- O'Neil J. R., Clayton R. N., & Mayeda T.K. (1969). Oxygen isotope fractionation in divalent metal carbonates. *Journal of Chemical Physics*, 51, 5547–5558.
- Pasteris, J.D., Wopenka, B., & Valsami-Jonwa, E. (2008). Bone and tooth mineralization: Why apatite? *Elements*, *4*, 97–104.

- Pellegrini, M., Lee-Thorp, J.A., & Donahue, R.E. (2011). Exploring the variation of the $\delta^{18}O_p$ and $\delta^{18}O_c$ relationship in enamel increments. *Palaeogeography*, *Palaeoclimatology*, *Palaeoecology*, *310*, 71–83.
- Polley, H.W., Johnson, H.B., Marino, B.D., & Mayeux, H.S. (1993). Increase in C3 plant water- use efficiency and biomass over Glacial to present CO₂ concentrations. *Nature*, 361, 61–64.
- Reimer, P.J., Baillie, M.G.L., Bard, E., Bayliss, A., Beck, J.W., Blackwell, P.G., Bronk Ramsey, C., Buck, C.E., Burr, G.S., Edwards, R.L., Friedrich, M., Grootes, P.M., Guilderson, T.P., Hajdas, I., Heaton, T.J., Hogg, A.G., Hughen, K.A., Kaiser, K.F., Kromer, B., McCormac, F.G., Manning, S.W., Reimer, R.W., Richards, D.A., Southon, J.R., Talamo, S., Turney, C.S.M., van der Plicht, J., & Weyhenmeyer, C.E. (2009). IntCal09 and Marine09 radiocarbon age calibration curves, 0–50,000 years cal BP. *Radiocarbon*, 51(4), 1111–1150.
- Richards, M.P., Mays, S., & Fuller, B.T. (2002). Stable carbon and nitrogen isotope values of bone teeth reflect weaning age at the Medieval Wharram Percy site, Yorkshire, UK. *American Journal of Physical Anthropology*, *119*, 205–210.
- Richards, M.P., & Hedges, R.E.M. (2003). Variations in bone collagen δ^{13} C and δ^{15} N values of fauna from Northwest Europe over the last 40 000 years. *Palaeogeography, Palaeoclimatology, Palaeoecology, 193*, 261–267.
- Rozanski, K., Araguas-Araguas, L. & Gonfiantini, R. (1993). Isotopic patterns in modern global precipitation. In P.K. Swart, J. Mckenzie, K.C. Lohmann, S. Savin (Eds.), *Climate Change in Continental Isotope Records: AGU Geophysical Monograph* (pp. 1–36). Washington, DC: American Geophysical Union.
- Schoeninger, M.J., DeNiro, M.J., & Tauber, H. (1984). Stable nitrogen isotope ratios of bone collagen reflect marine and terrestrial components of prehistoric human diet. *Science*, 220, 1381–1383.
- Schulze, E.-D., Chapin III, F.S., & Gebauer, G. (1994). Nitrogen nutrition and isotope differences among life forms at the northern treeline of Alaska. *Oecologia*, 100, 406–412.
- Shapiro, B., Drummond, A.L., Rambaut, A., Wilson, M.C., Matheus, P.E., Sher, A.V., Pybus, O.G., Gilbert, M.T.P., Barnes, I., Binladen, J., Willerslev, E., Hansen, A.J., Baryshnikov, G.F., Burns, J.A., Davydov, S., Driver, J.C., Froese, D.G., Harington, C.R., Keddie, G., Kosintsev, P., Kunz, M.L., Martin, L.D., Stephenson, R.O., Storer, J., Tedford, R., Zimov, S., & Cooper, A. (2004). Rise and fall of the Beringian steppe bison. *Science*, *306*, 1561–1565.
- Sharp, Z.D., & Cerling, T.E. (1998). Fossil isotope records of seasonal climate and ecology: Straight from the horse's mouth. *Geology*, *26*(3), 219–222.
- Shemesh A., Kolodny Y., & Luz, B. (1988) Isotope geochemistry of oxygen and carbon in phosphate and carbonate of phosphorite francolite. *Geochimica et Cosmochimica Acta*, 52, 2565–2572.
- Shemesh, A. (1990). Crystallinity and diagenesis of sedimentary apatites. *Geochimica et Cosmochimica Acta*, 54, 2433–2438.
- Simpson, G.G. (1951). Horses: The Story of the Horse Family in the Modern World and Through Sixty Million Years History. New York: Oxford University Press.
- Skinner, H.C.W. (2005). Biominerals. *Mineralogical Magazine*, 69, 621–641.

- Smith, B.N. (1972). Natural abundance of the stable isotopes of carbon in biological systems. *BioScience*, 22, 226–231.
- Smuts, G.L. (1975). Home range sizes for Burchell's zebra *Equus burchelli antiquorum* from the Kruger National Park. *Koedoe*, 18, 139–146.
- Sponheimer, M., Robinson, T., Ayliffe, L., Roeder, B., Hammer, J., Passey, B., West, A., Cerling, T., Dearing, D., & Ehleringer, J. (2003a). Nitrogen isotopes in mammalian herbivores: Hair δ^{15} N values from a controlled feeding study. *International Journal of Osteoarchaeology*, *13*(1–2), 80–87.
- Sponheimer, M., Robinson, T., Ayliffe, L., Passey, B., Roeder, B., Shipley, L., Lopez, E., Cerling, T., Dearing, D., & Ehleringer, J. (2003b). An experimental study of carbon-isotopefractionation between diet, hair, and feces of mammalian herbivores. *Canadian Journal of Zoology*, *81*, 871–876.
- Stevens, R.E., & Hedges, R.E.M. (2004). Carbon and nitrogen stable isotope analysis of northwest European horse bone and tooth collagen, 40, 000 BP–present: Palaeoclimatic interpretations. *Quaternary Science Reviews*, 23, 977–991.
- Stevens, R.E., Germonpré, M., Petrie, C.A., & O'Connell, T.C. (2009).
 Palaeoenvironmental and chronological investigations of the Magdalenian sites of Goyet Cave and Trou de Chaleux (Belgium), via stable isotope and radiocarbon analyses of horse skeletal remains. *Journal of Archaeological Science*, 36, 653–662.
- Stuart-Williams, H., & Schwarcz, H.P. (1995). Oxygen isotope analysis of silver orthophosphate using a reaction with bromine. *Geochimica et Cosmochimica Acta*, 58, 3837–3841.
- Stuiver, M., Polach, & H.A. (1977). Discussion: Reporting of ¹⁴C data. *Radiocarbon, 19*, 355–363.
- Suga, S. (1982). Progressive mineralization pattern of developing enamel during the maturation stage. *Journal of Dental Research*, *61*, 1532–1542.
- Sullivan, C.H., & Krueger, H.W. (1981). Carbon isotope analysis of separate chemical phases in modern and fossil bone. *Nature*, 292(5821), 333–335.
- Surovell, T.A., & Stiner, M.C. (2001). Standardizing infra-red measures of bone mineral crystallinity: an experimental approach. *Journal of Archaeological Science*, 28, 633–642.
- Svensson, A., Andersen, K.K., Bigler, M., Clausen, H.B., Dahl-Jensen, D., Davies, S.M., Johnsen, S.J., Muscheler, R., Rasmussen, S.O., Röthlisberger, R., Steffensen, J.P., & Vinther, B.M. (2006). The Greenland ice core chronology 2005, 15–42 ka. Part 2:comparison to other records. *Quaternary Science Reviews*, 25, 3258–3267.
- Szpak, P., Gröcke, D.R., Debruyne, R., MacPhee, R.D.E., Guthrie, R.D., Froese, D., Zazula, G.D., Patterson, W.P., & Poinar, H.N. (2010). Regional differences in bone collagen δ^{13} C and δ^{15} N of Pleistocene mammoths: Implications for paleoecology of the mammoth steppe. *Palaeogeography*, *Palaeoclimatology*, *Palaeoecology*, 286, 88–96.
- The Atlas of Canada. (2012). *Provincial and Territorial outline, Ontario*. Copy of an official work that is published by the Government of Canada, the reproduction has not been produced in affiliation with, or with the endorsement of the Government of Canada. Retrieved January, 2013, from http://atlas.gc.ca.

- Thompson, K.N. (1995). Skeletal growth rates of weanling and yearling thoroughbred horses. *Journal of Animal Science*, *73*, 2513–2517.
- Tieszen, L.L., & Fagre, T. (1993). Effect of diet quality and composition on the isotopic composition of respiratory CO₂, bone collagen, bioapatite, and soft tissues. In J.B. Lambert, G., Grupe (Eds.), *Prehistoric Human Bone-Archaeology at the Molecular Level* (pp. 121–155). Berlin: Springer-Verlag.
- Trueman, C.N.G., Behrensmeyer, A.K., Tuross, N., & Weiner, S. (2004). Mineralogical and compositional changes in bones exposed on soil surfaces in Amboseli National Park, Kenya: diagenetic mechanisms and the role of sediment pore fluids. *Journal of Archaeological Science*, 31, 721–739.
- Tütken, T., Vennemann, T.W., & Pfretzschner, H.-U. (2008). Early diagenesis of bone and tooth apatite in fluvial and marine settings: Constraints from combined oxygen isotope, nitrogen and REE analysis. *Palaeogeography, Palaeoclimatology, Palaeoecology, 266*, 254–268.
- Tütken, T., & Vennemann, T.W. (2011). Fossil bones and teeth: Preservation or alteration of biogenic compositions? *Palaeogeography, Palaeoclimatology, Palaeoecology, 310*, 1–8.
- Urey, H.C. (1947). The thermodynamic properties of isotopic substances. *Journal of the Chemical Society*, 562–81.
- van Der Merwe, N.J. (1989). Natural variations in ¹³C concentration and its effect on environmental reconstruction using ¹³C/ ¹²C ratios in animal bones. In T.D. Price (Ed.), *The Chemistry of Prehistoric Human Bone* (pp.105–125). Cambridge: Cambridge University Press.
- van Der Merwe, N.J., & Medina, E. (1989). Photosynthesis and ¹³C/¹²C ratios in Amazonian rain forests. *Geochimica et Cosmochimica Acta*, 53, 1091–1094.
- van Klinken, G.J. (1999). Bone collagen quality indicators for paleodietary and radiocarbon measurements. *Journal of Archaeological Science*, *26*, 687–695.
- Verburg, P. (2007). The need to correct for the Suess effect in the application of δ^{13} C in sediment of autotrophic Lake Tanganyika, as a productivity proxy in the Anthropocene. *Journal of Paleolimnology*, *37*, 591–602.
- Vogel, J.C. (1978). Isotopic assessment of the dietary habits of ungulates. *South African Journal of Science*, 74, 298–301.
- Wang, Y., Cerling, T.E., & MacFadden, B.J. (1994). Fossil horses and carbon isotopes: new evidence for Cenozoic dietary, habitat, and ecosystem changes in North America. *Palaeogeography, Palaeoclimatology, Palaeoecology, 107*, 269–279.
- Warinner, C., & Tuross, N. (2009). Alkaline cooking and stable isotope tissue-diet spacing in swine: archaeological implications. *Journal of Archaeological Science*, 36, 1690–1697.
- Weaver, A.J., Saenko, O.A., Clark, P.U., & Mitrovica, J.X. (2003). Meltwater pulse 1A from Antarctica as a trigger of the Bølling-Allerød warm interval. *Science*, 299(5613), 1709–1713.
- Weiner, S., & Bar-Yosef, O. (1990). States of preservation of bones from prehistoric sites in the Near East: a survey. *Journal of Archaeological Science*, *17*, 187–196.
- Williams, J.W., Shuman, B.N., Webb III, T., Bartlein, P.J., & Leduc, P.L. (2004). Late-Quaternary vegetation dynamics in North America: scaling from taxa to biomes. *Ecological Monographs*, 74(2), 309–334.

- Wolff, E., & Spahni, R. (2007). Methane and nitrous oxide in the ice core record. Philosophical Transactions: *Mathematical, Physical and Engineering Sciences*, 365(1856), 1775–1792.
- Wright, L.E., & Schwarcz, H.P. (1996). Infrared and isotopic evidence for diagenesis of bone apatite at Dos Pilas, Guatemala: palaeodietary implications. *Journal of Archaeological Science*, 23, 933–944.
- Young, R.R., Burns, J.A., Smith, D.G., Arnold, L.D., & Reins, R.B. (1994). A single, late Wisconsin, Laurentide glaciation, Edmonton area and southwestern Alberta. *Geology*, 22, 683–686.
- Zazula, G.D., Froese, D.G., Schweger, C.E., Mathewes, R.W., Beaudoin, A.B., Telka, A.M., Harington, C.R., & Westgate, J.A. (2003). Palaeobotany: ice-age steppe vegetation ineast Beringia. *Nature*, 423, 603–603.
- Zazzo, A., Lecuyer, C., & Mariotti, A. (2004a). Experimentally-controlled carbon and oxygen isotope exchange between bioapatites and water under inorganic and microbially-mediated conditions. *Geochimica et Cosmochimica Acta*, 68(1), 1–12.
- Zazzo, A., Lecuyer, C., Sheppard, S., M., F., Grandjean, P., & Mariotti, A. (2004b). Diagenesis and the reconstruction of paleoenvironments: a method to restore original δ^{18} O values of carbonate and phosphate from fossil tooth enamel. *Geochimica et Cosmochimica Acta, 68*, 2245–2258.
- Zazzo, A., Balasse, M., Patterson, W.P., & Patterson, P. (2005). High-resolution δ^{13} C intratooth profiles in bovine enamel: Implications for mineralization pattern and isotopic attenuation. *Geochimica et Cosmochimica Acta*, 69(14), 3631–3642.
- Zheng, Y.-F. (1996). Oxygen isotope fractionations involving apatites: application to paleotemperature determination. *Chemical Geology*, *127*, 177–187.

Curriculum Vitae

| Name: | Nicolle S Bellissimo |
|---|---|
| Post-secondary Education and Degrees: | The University of Western Ontario London, Ontario, Canada 2006–2010 B.Sc. (Honours Specialization in Geology) |
| | The University of Western Ontario London, Ontario, Canada 2010–2013 M.Sc. (Geology) |
| Honours and Awards: | Graduate Thesis Research Award, 2012 Queen Elizabeth II Scholarship, 2011 Ontario Graduate Scholarship in Science and Technology (OGSST), 2010 Caradocian Prize in Geochemistry, 2010 Dean's Honor List, 2006–2010 |
| Related Work Experience: | Teaching Assistant, Earth Science Department The University of Western Ontario 2010–2012 |
| | Research Assistant The University of Western Ontario 2009, 2010 |